

# The Demographic and Clinical Characteristics of Primary Liver Cancer Patients in Tasmania

by

Thi Thu Hoa Nguyen

Menzies Institute for Medical Research

College of Health and Medicine

Submitted in fulfilment of the requirements for the degree of Master of Medical Science

University of Tasmania, Australia

October 2021

# **Declaration of Originality**

This thesis contains no material which has been accepted for a degree or diploma by the University or any other institution, except by way of background information and duly acknowledged in the thesis, and to the best of my knowledge and belief it contains no material previously published and written by another person except where due acknowledgement is made in the text of the thesis, nor does the thesis contain any material that infringes copyright.

Singed:

Date: 31st March 2021

# **Authority of Access Statement**

This thesis may be made available for loan. Copying and communication of any part of this thesis are prohibited for two years from the date this statement was signed; after that time limited copying and communication is permitted in accordance with the Copyright Act 1968.

Signed:

Date: 31st March 2021

# **Statement of Co-Authorship**

The following people and institutions contributed to the studies of undertaken as part of this thesis:

**Candidate:** Hoa Thi Thu Nguyen; Menzies Institute for Medical Research, University of Tasmania, Hobart, Tasmania, Australia

Author 1: Dr. Barbara de Graaff; Menzies Institute for Medical Research, University of Tasmania, Hobart, Tasmania, Australia

Author 2: Prof. Andrew J. Palmer; Menzies Institute for Medical Research, University of Tasmania, Hobart, Tasmania, Australia; Melbourne School of Population and Global Health, University of Melbourne, Victoria, Australia

Author 3: Dr. Fiona Cocker; School of Medicine, University of Tasmania, Australia

Author 4: Dr. Kwang Chien Yee; Royal Hobart Hospital, Hobart, Tasmania, Australia; School of Medicine, University of Tasmania, Australia

Author 5: Mr. Brian Stokes; Tasmanian Cancer Registry, University of Tasmania, Hobart, Tasmania, Australia

Author 6: Dr. Karen Wills; Menzies Institute for Medical Research, University of Tasmania, Hobart, Tasmania, Australia

Author 7: Dr. Michael Braude; Department of Gastroenterology and Hepatology, Monash University, Melbourne, Victoria, Australia

Author 8: Dr. Cristina Moldovan; Consultant Medical Oncologist, Icon Cancer Centre Hobart, Tasmania, Australia

Contribution of work by co-authors for each paper:

Paper 1: Located in Chapter 3

Hoa, NTT, Chien Yee K, Wills K, Stokes B, Cocker F, Palmer AJ, de Graaff B. Characteristics and survival of primary liver cancer patients in Tasmania by subtypes: A data linkage study. 2021.

Author contributions:

Designed study: Candidate, Author 1, Author 2, Author 3, Author 4, Author 5

Ethics approval process: Author 1, Author 2, Author 3

Analysed the data: Candidate

Interpreted the data: Candidate, Author 1, Author 6

Wrote the manuscript: Candidate

Revised the manuscript: Candidate, Author 1, Author 2, Author 3, Author 4, Author 5, Author 6

Paper 2: Located in Chapter 4

Hoa NTT, Chien Yee K, Moldovan C, Braude M, Stokes B, Cocker F, Palmer AJ, de Graaff B. Accuracy of coded cause of death data: A case study based on primary liver cancer. 2021.

Author contributions:

Designed study: Candidate, Author 1, Author 2, Author 3, Author 4, Author 5

Ethics approval processes: Author 1, Author 2, Author 3

Reviewed coded cause of death differences: Author 4, Author 7, Author 8

Analysed the data: Candidate

Interpreted the data: Candidate, Author 1

Wrote the manuscript: Candidate

Revised the manuscript: Candidate, Author 1, Author 2, Author 3

We, the undersigned, endorse the above stated contribution of work undertaken for each of the manuscripts contributing to this thesis:

#### Signed:

Hoa Nguyen	Dr. Barbara de Graff	Prof Alison Venn
Candidate	Primary Supervisor	Director of
Menzies Institute for Medical Research	Menzies Institute for Medical Research	Menzies Institute for Medical Research
University of Tasmania	University of Tasmania	University of Tasmania
Date: March 2021	March 2021	March 2021

# **Statement of Ethical Conduct**

The research investigations conducted for this thesis abide by the ethical requirements of the University of Tasmania Human Research Ethics Committee. Ethics approval was gained through applications H0016958.

Signed:

Date: 31st March 2021

# Acknowledgement

I would like to express my deepest appreciation to many people that without their assistance this thesis would not have been possible. First, I would like to acknowledge my primary supervisor Dr. Barbara de Graff and two co-supervisors, Prof. Andrew Palmer and Dr. Fiona Cocker, for their tremendous support, patience, intellectual contributions to my research, especially, their tireless encouragement during the Covid-19 outbreak in Australia and Vietnam.

I wish to thank my collaborating co-authors Dr. Kwang Yee, Dr. Karen Wills, Dr. Brian Stokes, Dr. Michael Braude, Dr. Cristina Moldovan for the assistance they provided to enable access to the health data system, as well as clinical, statistical advice along the learning journey.

Thank you to colleagues and my lunch group for their wonderful support, encouragement, and useful advice they have given to me.

Finally, I am very grateful to my parents and siblings for their love and support regardless of distance and time. Special thanks to my husband and my daughter for their on-going support in every situation and decision I have made in my life.

# **Table of Contents**

Declaration of	of Originality ii
Authority of	Access Statement iii
Statement of	Co-Authorshipiv
Statement of	Ethical Conductvi
Acknowledge	ement vii
Table of Con	tents viii
List of Figure	es xii
List of Table	s xiii
List of Abbro	eviationsxiv
Abstract	xvi
Chapter 1. Ir	ntroduction1
1.1. Prin	nary liver cancer1
1.2. Тур	es of liver cancer1
1.2.1.	Hepatocellular carcinoma1
1.2.2.	Cholangiocarcinoma1
1.2.3.	Angiosarcoma2
1.2.4.	Hepatoblastoma2
1.2.5.	Fibrolamellar carcinoma2
1.3. Inci	dence2
1.4. Mo	-tality4
1.5. Dat	a sources4
1.6. Der	nographic profile of liver cancer patients
1.6.1.	Sex disparities
1.6.2.	Age

1.6.3.	Ethnicity	6
1.7. Ris	sk factors of liver cancer	7
1.7.1.	Cirrhosis	7
1.7.2.	Chronic hepatitis B infection	7
1.7.3.	Chronic hepatitis C infection	8
1.7.4.	Alcohol related liver disease	8
1.7.5.	Non-alcoholic fatty liver disease	9
1.7.6.	Aflatoxin	9
1.7.7.	Obesity and Type 2 Diabetes	10
1.8. Dia	agnosis of liver cancer	10
1.8.1.	Tumour makers	10
1.8.2.	Liver ultrasound	10
1.8.3.	CT scan	11
1.8.4.	MRI scan	11
1.8.5.	Biopsy	11
1.8.6.	Staging of liver cancer	11
1.9. Su	rveillance of liver cancer	12
1.10.	The context of primary liver cancer in Australia	13
1.10.1.	Incidence and mortality	13
1.10.2.	Risk factors for liver cancer in Australia	15
1.11.	What is the need for primary liver cancer research in Tasmania?	16
1.12.	Aims of this research	16
1.13.	The structure of the thesis	17
Chapter 2. N	Viethods	19
2.1. Stu	ıdy design	19
2.2. Par	rticipants	19

2.3.	Data Linkage process	19
2.4.	Diagnostic coding of liver cancer	21
2.5.	Data collection from each dataset	22
2.6.	Data analysis	24
2.7.	Ethics	25
Chapter	r 3. Characteristics and survival of primary liver cancer patients in Tasmania by	7
subtype	es: A data linkage study	27
3.1.	Abstract	27
3.2.	Introduction	27
3.3.	Methods	28
3.4.	Results	30
3.5.	Discussion	35
3.6.	Conclusions	
Chapter	r 4. Accuracy of coded cause of death data: A case study based on primary liver	cancer
Chapter	r 4. Accuracy of coded cause of death data: A case study based on primary liver	cancer 39
<b>Chapter</b> 	r 4. Accuracy of coded cause of death data: A case study based on primary liver	cancer 39 39
<b>Chapter</b> 4.1. 4.2.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	cancer 39 39 40
Chapter 4.1. 4.2. 4.3.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42
Chapter 4.1. 4.2. 4.3. 4.4.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42 44
Chapter 4.1. 4.2. 4.3. 4.4. 4.5.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42 42 44
Chapter 4.1. 4.2. 4.3. 4.4. 4.5. 4.6.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42 44 44 49 51
Chapter 4.1. 4.2. 4.3. 4.4. 4.5. 4.6. Chapter	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42 42 44 44 51 <b>53</b>
Chapter 4.1. 4.2. 4.3. 4.4. 4.5. 4.6. Chapter 5.1.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	cancer 
Chapter 4.1. 4.2. 4.3. 4.4. 4.5. 4.6. Chapter 5.1. 5.2.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	<b>cancer</b> <b>39</b> 40 42 42 44 49 51 <b>53</b> 53
Chapter 4.1. 4.2. 4.3. 4.4. 4.5. 4.6. Chapter 5.1. 5.2. 5.3.	r 4. Accuracy of coded cause of death data: A case study based on primary liver of Abstract	cancer 39 40 42 42 44 49 51 53 53 53

References	
Appendix A: A publication co-authored by the c	andidate69

# List of Figures

Figure 1: The region-specific ASR incidence rates by sex for liver cancer in 2018
Figure 2: The region-specific ASR mortality rates by sex for liver cancer in 2018
Figure 3: Region specific incidence age-standardized rates by sex for cancers of the liver in 2018
Figure 4: Incidence trends of liver cancer in Australia from 1982-201514
Figure 5: Mortality trends of liver cancer in Australia from 1982-201514
Figure 6: Liver cancer stages15
Figure 7: Data linkage process
Figure 8: Kaplan-Meier survival curve for all liver cancer cases from the time of diagnosis between 2007 and 2015
Figure 9: Kaplan-Meier survival curves for those diagnosed with liver cancer by subtypes
Figure 10: The process of recording the cause of death in ABS, TCR
Figure 11: Cumulative incidence function curves illustrate deaths from liver cancer patients and other
reasons by gender (2-A), place of residence (2-B), country of birth (2-C), and type of liver cancer (2-
D) from ABS, TCR, Medical Practitioners' data48

# List of Tables

Table 1: Data collection from each dataset	22
Table 2: Demographic characteristics of liver cancer patients in Tasmania between 2007 and 2015	32
Table 3: Prevalence of risk factors of liver cancer patients by inpatient admission ICD-10 codes	33
Table 4: 1-, 3- and 5-year relative survival for patients diagnosed with liver cancer from 2007-2015	35
Table 5: Characteristics of deceased cases among liver cancer patients in Tasmania (2007-2015)	45
Table 6: The agreement between the ABS, TCR and the medical practitioners regarding underlying ca   of death of liver cancer patients in Tasmania	iuse 45
Table 7: Hazard Ratios by sex, place of residence, country of birth, and type of liver cancer from the   ABS, TCR, Medical Practitioners' data	47

# List of Abbreviations

AASLD	American Association for the Study of Liver Diseases
ABS	Australian Bureau of Statistics
AFP	Alpha fetoprotein
AIHW	Australian Institute of Health and Welfare
ASR	Age standardize rate
BCLC	Barcelona Clinic Liver Cancer
BMI	Body mass index
CHB	Chronic hepatitis B
CHC	Chronic hepatitis C
CI	Confidence Intervals
COD-URF	Cause of Death Unit Record File
СТ	Computerised tomography
DAA	Direct-acting antiviral agents
DMR	Digital Medical Record
DOH	Department of Health
EASL	European Association for the Study of the Liver
ECOG	Eastern Cooperative Oncology Group
НСС	Hepatocellular carcinoma
IARC	International Agency for Research on Cancer
ICD-10	International Classification of Diseases and Related Health Problems 10th Revision
ICD-O3	International Classification of Diseases for Oncology, 3rd Revision
MRI	Magnetic resonance imaging
NAFLD	Non-alcoholic fatty liver disease
NOS	Not otherwise specified

PLC	Primary liver cancer
PPID	Projected Person Identifier
SA2	Area of Usual Residence
SACC	Standard Australian Classification of Countries
SHR	Subdistribution hazard ratios
SVR	Sustained virological response
TCR	Tasmanian Cancer Registry
US	United States
WHO	World Health Organization

## Abstract

Primary liver cancer (PLC) is the sixth most common cancer globally, and is the fourth leading cause of cancer mortality, accounting for 8.2% of all cancer deaths in 2018. Despite advances of new diagnostics and treatments, survival time of PLC patients remains poor, particularly in the absence of targeted screening programs to increase rates of early diagnosis. In Australia, both incidence and mortality rates of PLC have increased substantially, in contrast to the improved outcomes that have been observed for almost all other cancers.

The risk factors associated with PLC include chronic hepatitis B (CHB), chronic hepatitis C (CHC), diabetes, alcohol related liver disease, and non-alcoholic fatty liver disease (NAFLD). Demographic and epidemiological changes are impacting incidence of PLC in Australia, for example, the migration of CHB patients from high hepatitis B prevalence countries, and the increasing rates of obesity and diabetes. Understanding these changing risk factors and the impact on survival is important because it will help to inform healthcare policy for appropriate allocation of resources for prevention, targeted screening/surveillance and treatment. The aims of this Masters' research program are to describe the demographic characteristics, risk factors and survival time for all PLC cases in Tasmania (Study 1: Chapter 3) and to investigate the level of agreement for cause of death data between the Australian Bureau of Statistics (ABS) and Tasmanian Cancer Registry (TCR), along with medical specialist opinions and its impact on cause-specific survival (Study 2: Chapter 4).

Over a nine year period between 2007 and 2015, there were 293 PLC cases identified by using linked administrative data between the datasets. Hepatocellular carcinoma (HCC) (51.9%) and cholangiocarcinoma (20.5%) were the main types reported. Three-quarters of all PLC cases were male, and the average age was 70 years. For cases with a public hospital admission, 43% did not have a risk factor for PLC identified. Of those who did, the most common were cirrhosis (37%), chronic viral hepatitis (35%), diabetes (27%), and alcohol-related liver disease (23%). The mean age at diagnosis for all cases was 69.6 years, with a median survival time of 6.2 months. The 1-,3- and 5- year relative survival rates were 38.3%, 12.8%, and 6.7% respectively.

The linked PLC cases provided the opportunity to compare specific causes of death between the TCR and ABS. Conflicting records of cause of death were recorded for almost half of all PLC, with 20 cases had non-PLC underlying causes of death from the TCR dataset and 42 cases from the ABS dataset. These cases were independently reviewed by medical specialists with expertise in PLC. Concordance of cause of death data was estimated using Kappa statistics, and the impact on cause-specific survival time was explored using a competing risk framework. The overall concordance regarding causes of death data was minimal between the ABS and TCR (Kappa=0.35), moderate between the ABS and medical practitioners (Kappa=0.61), and strong agreement (Kappa=0.87)

xvi

between the medical practitioners were observed. These results reflect a greater level of agreement between medical practitioners than between coding agencies. Overall, cause-specific survival time was similar across the TCR, ABS and medical practitioners by sex, place of residence and country of birth, however, a small difference was observed for the type of liver cancer between the ABS, TCR and medical practitioners. As liver cancer is a low survival cancer, such results may be different to cancers with better survival such as breast cancer.

This thesis has made many distinct contributions including reporting the epidemiology of PLC in Tasmania and characterising the risk factors specific to this setting. These results support the need for surveillance to increase the rate of early detection of HCC in high risk patients. In turn, this will improve the very poor survival of PLC cases in Australia. This is the first study that has evaluated the impact of different coded causes of death on estimates of cause-specific survival for liver cancer cases.

# **Chapter 1. Introduction**

This chapter provides the outline of this thesis, beginning with the definition of PLC and its subtypes (Sections 1.1 and 1.2). The incidence, mortality, demographic profile and risk factors of PLC worldwide is presented in Sections 1.3-1.7. Following this, the stage, methods and surveillance of PLC are presented in Sections 1.8 and 1.9. The context of PLC in Australia is discussed in Section 1.10. The need for PLC research in Tasmania is in Section 1.11. Finally, the aims and structure of the thesis are presented in Sections 1.12 and 1.13.

### **1.1. Primary liver cancer**

Cancers of the liver can be classified as primary and secondary types. PLC arise from liver cells: hepatocellular carcinoma, cholangiocarcinoma, angiosarcoma or hepatoblastoma<sup>1, 2</sup>.

Secondary liver cancer is common and is metastases from tumours elsewhere in the body. The focus of this thesis is PLC, and as such all references to "liver cancer" refer to PLC.

# **1.2.** Types of liver cancer

There are five main types of PLC, each named for the type of cell from which the cancer develops. Each of these types of PLC will now be described.

#### 1.2.1. Hepatocellular carcinoma

HCC also known as hepatoma, is the most common type of PLC. HCC development starts in the hepatocytes, the predominant cell type in the liver, and accounts for 85-90% of all PLC worldwide<sup>1, 3</sup>. HCC mostly occurs in the setting of cirrhosis (i.e., scarring of the liver), a condition often related to chronic liver diseases including CHB and CHC infections, alcohol related liver disease, NAFLD, and ingestion of the fungal aflatoxin B1<sup>1, 2, 4</sup>. These risk factors are discussed in Section 1.7 below.

The incidence and mortality of HCC are discussed in Sections 1.3 and 1.4.

#### **1.2.2.** Cholangiocarcinoma

Cholangiocarcinoma, also known as bile duct cancer, develops from the biliary epithelium, and it can be divided to extrahepatic (75%-80%) and intrahepatic cholangiocarcinoma (20%-25%)<sup>1</sup>. These two types have different risk factors, presentation, prognosis, and treatment. However, only intrahepatic cholangiocarcinoma is considered as a PLC based on the anatomical location in the liver, and as such, this thesis only includes this type of cholangiocarcinoma.

Cholangiocarcinoma is the second most common PLC subtype globally, accounting for 10-15% of cases<sup>5-8</sup>. It is more common in Asian countries, with the highest incidence rates observed in Thailand (22.9 per 100,000 persons)<sup>9</sup>, followed by China (7.5 per 100,000 persons) and South Korea (5.6 per 100,000 persons)<sup>10</sup>. Cholangiocarcinoma is rare in European countries and the US, but there is evidence of an increasing trend in incidence from 0.1/100,000 to 2.0/ 100,000 in some Western countries<sup>7, 8</sup>.

#### 1.2.3. Angiosarcoma

Angiosarcoma is a very rare type of PLC that forms in the lining of the hepatic blood vessels or lymphatic vessels<sup>11, 12</sup>. About 0.1-2.0% of PLC cases in Australia are hepatic angiosarcomas. This form is predominantly observed in patients aged between 60 and 80 years<sup>11, 12</sup>. This rare cancer tends to progress very quickly and has a poor prognosis as it is readily carried by the blood flow, easily metastasizes to distant areas, and has a high rate of recurrence rate<sup>12</sup>.

#### 1.2.4. Hepatoblastoma

Hepatoblastoma is a very rare type of PLC that develops from the primitive cell lines including foetal or embryonic stage hepatocytes<sup>13</sup>. It affects children during the first three years of life, and it appears as an abdominal mass in the liver<sup>13</sup>. Whilst the aetiology of hepatoblastoma is unknown, several genetic abnormalities have been identified as risk factors including Beckwith-Wiedemann syndrome, familial adenomatous polyposis, hemihypertrophy<sup>14</sup>.

#### 1.2.5. Fibrolamellar carcinoma

Fibrolamellar carcinoma is a distinct histological variant of HCC that typically affects young adults (10-35 years old) without sex predilection. The aetiology remains unknown but unlike HCC, this form is not associated with the risk factors of HCC such as cirrhosis, CHB, CHC<sup>15</sup>. Fibrolamellar carcinoma is characterized by a large, solitary, firm tumour that is well differentiated, with a dense fibrotic background in the liver. It accounts for 1-9% of all HCC<sup>16</sup>.

#### **1.3. Incidence**

According to 2018 Global Cancer Statistics, PLC was the sixth most diagnosed cancer and the second leading cause of cancer mortality<sup>17</sup>. Annually, an estimated 840,000 new cases are diagnosed, and 780,000 deaths occur<sup>17</sup>. The incidence and mortality rates of PLC vary greatly, reflecting the uneven distribution of its risk factors, which are discussed in Section 1.7.

Low-and middle-income countries experience the greatest burden of PLC, where nearly 85% of PLC are diagnosed<sup>17, 18</sup> and the age standardize rate (ASR) incidence in is typically greater than 20 per

100,000 persons, e.g. countries in Eastern Asia, South-Eastern Asia, and Sub-Saharan Africa with incidence rates greater than 20/100,000 include: Mongolia: 93.7/100,000; China: 18.3/100,000; Vietnam: 23.2/100,000; Laos: 22.4/100,000; Thailand: 21.0/100,000, Gambia: 23.9/100,000; and Guinea: 21.8/100,000<sup>17</sup>. Although China bears more than half of all PLC cases globally due to its large population, Mongolia has the highest PLC rate in the world, with 117.0/100,000 for males and 74.1/100,000 for females<sup>17</sup>.

In contrast, lower ASR incidence rates are reported in high income countries, such as Australia (5.7/100,000), the US (6.8/100,000), and in many European countries, the rate is less than 10 per 100,000 persons<sup>17</sup>. That said, these low rates have been increasing in recent years. For example, over the last 30 years in the US, ASR of PLC incidence has tripled from 1.6 to 4.9 per 100,000 persons between 1975 and 2005<sup>19</sup>. Similarly, in Australia, ASR incidence increased by more than 300% between 1982 and 2015, from 1.38/100,000 to 5.7/100,000<sup>4, 20</sup>. These increasing trends are related to changes in risk factors including CHB, CHC, alcoholic liver disease, and non-alcoholic fatty liver disease which is associated with type 2 diabetes and obesity<sup>18, 21</sup>.

In contrast, incidence rates have declined in most Asian countries. The most notable reason for this is the reduction of CHB, the predominant risk factor for PLC in this region, due to introduction of an effective CHB vaccination of newborns in 2000 and a decline in the rates of HCC in younger age groups<sup>18</sup>. Additionally, the rates have declined in Asian countries by reducing exposure to aflatoxin B1 levels in the population, especially in China and Thailand<sup>21, 22</sup>.



Estimated age-standardized incidence rates (World) in 2018, liver, both sexes, all ages

Figure 1: The region-specific ASR incidence rates by sex for liver cancer in 2018<sup>17</sup>

## **1.4.** Mortality

PLC is the sixth leading cause of cancer mortality worldwide, with an estimated 782,000 deaths in 2018<sup>17</sup>. Many countries reported PLC as the fastest growing cause of any cancer death<sup>17, 18</sup>.

The highest ASR mortality rates were reported for Asian countries (Mongolia: 75.4/100,000; Vietnam: 23.2/100,000; Thailand: 20.9/100,000)<sup>17</sup>, and Egypt (31.8/100,000)<sup>17</sup>, followed by some Sub-Saharan countries (Guinea: 19.5/100,000; Liberia 15.4/100,000; and Ghana: 15.4/100,000)<sup>17</sup>. Reflecting incidence rates, ASR mortality rates for high income countries were substantially lower: US: 4.9/100,000; Italy: 5.7/100,000; France: 6.3/100,000)<sup>17</sup>.

The prognosis of PLC is poor, with low survival rates despite advances in treatment<sup>2, 4, 23</sup>. Many studies have found survival time is significantly longer in patients who are diagnosed at early stages and receive immediate treatment<sup>23</sup>. Treatments such as transplant or resection in these patients can achieve a 5-year survival rate of 70%<sup>23</sup>. For patients diagnosed at an advanced stage of PLC, palliative care is frequently the only treatment option, with median survival time less than one year<sup>2, 23</sup>.

The five-year net survival for PLC has increased from 11.9% between 1995-2000 to 14.5% between 2004-2009 globally based on a study using 187 population-based registries and 36 countries<sup>24</sup>.



Estimated age-standardized mortality rates (World) in 2018, liver, both sexes, all ages

# Figure 2: The region-specific ASR mortality rates by sex for liver cancer in 2018<sup>17</sup>

## **1.5. Data sources**

Cancer registries are the core source of data on cancer globally, even though the International Association of Cancer Registries (IACR) estimated that cancer registries cover only 21% of the world's population<sup>25</sup>. The population coverage and quality of cancer registries vary by locations,

states and countries, are more commonly established in high income countries and are often not an integral part of middle and low income countries' cancer data collection and reporting infrastructure. In addition, the availability and quality of death certificate information vary widely, especially in low and middle income countries where either poor quality data or a complete absence of death registrations occur. The cause of death data sources are described in more detail in section 4.2, Chapter 4.

## 1.6. Demographic profile of liver cancer patients

#### **1.6.1. Sex disparities**

Globally, almost all countries report PLC incidence and mortality rates in males that are two to three times higher than rates in females<sup>17</sup> (Figure 3). The greatest sex disparities in incidence are observed in Samoa with a male to female ratio of 6.9 (17.3/100,000 in male and 2.5/100,000 in female), and French Guiana (6.7; 15.5/100,000 in men and 2.3/100,000 in women)<sup>17</sup>. Intermediate disparities in incidence are observed in China: 3.0 with 27.6/100,000 for males and 9.0/100,000 for females; Australia: 3.0 with 8.6/100,000 for males and 2.9/100,000 for females<sup>17</sup>. Exceptions to this, in which only marginal disparities are observed include Bolivia: 0.9 with 5.6/100,000 for males and 2.7/100,000 for females<sup>17</sup>.

Although the reason for this difference is not completely understood, it is partly explained by higher levels of androgen in males, which promotes development of HCC<sup>26</sup>. In contrast, higher levels of estrogen in females may inhibit HCC development through its anti-inflammatory effects. Further, CHB, CHC, alcohol consumption is more prevalent in males, contribute to higher rates of HCC<sup>26</sup>.



Figure 3: Region specific incidence age-standardized rates by sex for cancers of the liver in 2018<sup>17</sup>

#### 1.6.2. Age

PLC incidence rates increase with age in all populations, with the highest incidence among those aged 70 years and older. In countries with lower incidence rates such as the US and many European countries, the incidence rates are generally very low before age of 40 and then gradually increase with age. In high incidence countries of East Asia and Africa, PLC incidence rates increase until age of 55 years and then plateau until age of 70 years. This is likely due to the high rates of CHB in these countries, many of which occur through vertical transmission at birth<sup>27, 28</sup>.

#### 1.6.3. Ethnicity

HCC incidence and mortality also vary between different ethnic groups. For example, in the US the incidence and mortality rates were highest in the American Indian/Inuit groups (incidence 15.2 per 100,000 persons; mortality 11.9 per 100,000 persons), followed by Asian and Pacific Islanders (incidence 13.5 per 100,000 persons; mortality 9.8 per 100,000 persons). The lowest rates were reported for non-Hispanic whites (incidence 6.3 per 100,000 persons; mortality 5.5 per 100,000 persons)<sup>29</sup>. Regarding survival time between different ethnicities, 5-year survival is highest in Asian and Pacific Islanders (27.1%) and lowest in American Indian/Inuit (16.2%), and African American

groups (16.3%)<sup>29</sup>. Differences in the prevalence of risk factors (CHB, CHC, alcoholic liver disease, non-alcoholic fatty liver disease) account for much of the observed variation in liver cancer rates, and to some degree, the disparities in access to high quality care<sup>21, 22, 29, 30</sup>.

#### **1.7. Risk factors of liver cancer**

There are several factors associated with the risk of developing PLC including cirrhosis of the liver, CHB infection, CHC infection, NAFLD, obesity, type 2 mellitus, and smoking<sup>4, 31</sup>. As HCC is the main type of PLC (making up to 70 - 90% of total numbers), the following will focus on its risk factors.

#### 1.7.1. Cirrhosis

Cirrhosis is severe scarring of the liver due to risk factors including viral hepatitis, excessive consumption of alcohol and NAFLD. In response to chronic liver injury caused by these risk factors, nodules within the liver develop, which are surrounded by fibrous bands. In turn, this frequently leads to portal hypertension and liver cancer<sup>32</sup>. More than 80% of PLC occurs in the setting of cirrhosis<sup>33-35</sup>. This condition is characterised by scarring of the liver and usually occurs when the liver is damaged over an extended period. It is referred to as the end-sate of chronic liver disease<sup>34-36</sup> and is strongly associated with HCC with one US-based cohort study in the reporting 95% of HCC patients had a diagnosis of cirrhosis<sup>33</sup>.

#### 1.7.2. Chronic hepatitis B infection

CHB is considered is one of the most common risk factors for PLC. Overall, almost 50% of cases of PLC are attributable to CHB infections and CHC infection<sup>37</sup>. The virus acts as a mutagenic agent which causes chromosomal rearrangement and increases genomic instability<sup>38</sup>. Through integration of DNA into the host genome in hepatocytes, the virus has been clearly linked to the development of HCC with the risk of developing HCC is higher among CHB patients compared to people who are not infected<sup>39</sup>.

CHB affects almost 3.5% of the world population and results in 887,000 deaths<sup>37</sup>. Asian and sub-Saharan African countries have the highest prevalence of CHB, accounting for 68% of cases. The annual incidence from CHB to HCC is 2-8%<sup>38,40</sup>. Most viral transmissions in high prevalence CHB countries occur via vertical transmission, either perinatally or soon after birth. In contrast, in lower prevalence countries the main transmission routes are through blood and bodily fluid contacts such as injecting drug use, sexual contact, and tattooing<sup>37</sup>.

Vaccination against HBV for all newborns and high-risk groups is recommended by the World Health Organization (WHO)<sup>37</sup>. By the end of 2016, 186 countries had introduced universal hepatitis B

vaccination programs to their national immunization campaigns<sup>37</sup>. This has effectively reduced the risk of HCC, with many countries reducing the prevalence of HBV across younger age groups, especially in Eastern Asian countries<sup>37, 41</sup>.

WHO recommends the use of oral treatments including tenofovir or entecavir. These treatments do not cure CHB infection, but it suppresses the replication of the virus<sup>37</sup>. However, even with the known benefits of CHB vaccinations and treatments, the risk of developing HCC cannot be eliminated. Thus, lifelong surveillance programs for patients with CHB must be continued.

#### 1.7.3. Chronic hepatitis C infection

About 1% of the world population is estimated to have CHC, with the crude incidence of 23.7 per 100,000 persons<sup>37</sup>. CHC accounts for nearly 400,000 deaths every year, mostly due to CHC-related cirrhosis and HCC<sup>37</sup>. The regions most affected are Eastern Mediterranean, with the incidence rate of 62.5/100,000, and the European Region (61.8/100,000), followed by African counties (31.0/100,000), and South-East Asia (14.8/100,000)<sup>37</sup>. The Western Pacific region recorded the lowest rate of 6.0/100,000<sup>37</sup>. Among CHC patients, the risk of developing cirrhosis after 20 years varies between studies and areas but is estimated to be 10-15% for males and 1-5% for females<sup>42, 43</sup>.

The HCV transmissions route is associated with direct percutaneous exposure to blood, via blood transfusions, needle sharing by intravenous drug users, the primary route in high income countries. In contrast, in low- and middle-income countries, the main transmission mechanisms are contaminated blood transfusions and unsafe medical procedures<sup>37</sup>.

Although no vaccines for the CHC have been developed, the discovery of direct-acting antiviral agents (DDAs) such as sofosbuvir have revolutionised treatment. Over the last five years, DAAs have largely replaced previous treatments using pegylated interferon: which was characterised by low uptake and serious adverse events<sup>44</sup>. DAAs are highly effective, with approximately 95% of patients achieving a sustained virological response (SVR), i.e., a cure. However, DAAs are very expensive and access to treatment is lacking in many countries. SVR has shown to reduce the risk of HCC in CHC patients<sup>44</sup>.

#### 1.7.4. Alcohol related liver disease

Alcohol-related liver disease occurs in the context of alcohol overconsumption. Clinic presentations range from fatty liver/hepatic steatosis to alcoholic hepatitis, fibrosis, cirrhosis, and leading to HCC<sup>45</sup>. The International Agency for Research on Cancer (IARC) has categorised alcoholic drinks as "carcinogenic to humans" and likely to increase the risk of PLC<sup>46</sup>. Some guidelines consider consumption of more than 40g of alcohol per day at risk of developing liver disease<sup>47</sup>.

8

Alcoholic liver disease is a more prominent PLC risk factor in high income countries such as the US and across Europe. The highest per capita alcohol consumption (21.3g/day) has been reported for Europe and the lowest (1.2g/day) in the Middle East and Northern Africa, with a global average of 15.1g/day<sup>48</sup>. High levels of alcohol consumption increase the relative risk of developing HCC by 3-10 times<sup>49</sup>. In addition, alcohol acts synergistically with other risk factors such as CHB, CHV, and tobacco use to increase the risk of PLC <sup>46, 49</sup>.

#### 1.7.5. Non-alcoholic fatty liver disease

NAFLD can result in liver damage particularly if fat accumulation in the liver progresses with inflammation<sup>50, 51</sup>. NAFLD is part of Metabolic Syndrome characterized by insulin resistance and being overweight or obese<sup>50, 52</sup>. In the general population, the global prevalence of NAFLD is estimated at 21% and expected to increase to 100 million in 2030<sup>52</sup>. Prevalence is increasing, particularly in low- and middle-income regions such as Asia and Africa due to rapid lifestyle and dietary changes in these areas<sup>53</sup>.

Prevalence is highest in the Middle East and South America (more than 30%), followed by Asian countries (estimate of 27%) and European countries with an average of 23.71%)<sup>53</sup>. Whilst HCC due to NAFLD is less commonly observed than other aetiologies, the increasing global prevalence of obesity is expected to contribute to increased prevalence of NAFLD associated HCC in the coming decades. In contrast to other liver diseases, NAFLD can lead to HCC without cirrhosis, similar to CHB infection

The mean annual rate of progression in non-alcoholic steatohepatitis (NASH) (the extreme form of NAFLD) to fibrosis was 41% and HCC incidence among NAFLD cases is 0.44 per 1,000 personyears. Although PLC cases related to NAFLD are less common compared to other factors, the impact may be substantial in the context of the increasing prevalence trends of obesity, diabetes and metabolic syndrome observed globally<sup>51</sup>.

#### 1.7.6. Aflatoxin

Aflatoxin is a mycotoxin produced by the Aspergillus flavus and Aspergillus parasiticus moulds that contaminates peanuts, wheat, soybeans, ground nuts, corn, and rice<sup>54</sup>. Cultivation and storage in moist, warm environments can lead to the growth of this fungus<sup>54</sup>. The most potent aflatoxin B1 has been classified as a group 1 human carcinogen by IARC<sup>54-57</sup>. Between 4.6-28.2% of all HCC cases are attributable to aflatoxins globally<sup>57</sup>. It is more common in warmer and tropical countries such as sub-Saharan Africa and East Asia. Further, many countries, such as Australia have strict rules in place governing testing of imported at-risk food products for aflatoxins<sup>58</sup>.

#### 1.7.7. Obesity and Type 2 Diabetes

Obesity is a condition in which excess body fat has accumulated to an extent that may have negative health effects. Obesity is defined as a body mass index (BMI) more than 30kg/m<sup>2</sup>. The WHO estimated that there were 650 million obese adults (13%) in 2016<sup>59</sup>. Obesity has been shown in many studies to independently increase the risk of developing HCC, largely due to metabolic syndrome and its hepatic manifestations (i.e., NAFLD)<sup>59, 60</sup>. For example, a report in European Prospective Investigation into Cancer and Nutrition found obesity was attributable to 16% of HCC cases in Europe<sup>60, 61</sup>. Further, a systematic review of BMI and HCC risks in Western populations found obesity was independently associated with a two-fold of HCC-related mortality compared to those with normal BMI<sup>62</sup>.

Type 2 diabetes is related to an increased risk of developing HCC, usually in those who have other risk factors such as CHB, CHC patients. Globally, there were an estimated 392 million people (6% in 2015) with type 2 diabetes, making up 90% of all cases of diabetes<sup>63</sup>. Type 2 diabetes occurs mainly because of obesity, overweight and lack of physical activities<sup>64</sup>. Some studies have found patients with type 2 diabetes have about a 2-3 times higher risk of developing HCC compared to those without diabetes<sup>21, 65</sup>.

#### 1.8. Diagnosis of liver cancer

#### 1.8.1. Tumour makers

Alpha fetoprotein (AFP) is a biomarker protein that can be measured in the blood and is widely used in surveillance programs for HCC<sup>27</sup>. As the sensitivity of this measure (approximately 60%) and specificity (80% at the cut-off level of 10-20ng/mL<sup>66</sup>) is limited, this test is carried out in conjunction with abdominal imaging such as ultrasound, CT and MRI. AFP levels greater than the healthy reference range can occur in the context of pregnancy, hepatitis, and jaundice<sup>67</sup>. Due to the limitations of this test, AFP is used as a sign of HCC, often in combination with liver ultrasound, for surveillance of HCC for high-risk patients.

#### 1.8.2. Liver ultrasound

Liver ultrasound is a useful, real-time, non-invasive test that can be used for screening patients at high-risk and in the diagnosis of PLC<sup>3, 68</sup>. It is used to assess the structural integrity of the liver and features of portal hypertension<sup>3, 68, 69</sup>. Ultrasound may be indicated in people with abnormal liver function tests; symptomatic presentations with abdominal pain, jaundice<sup>3, 68, 69</sup>.

The sensitivity of liver ultrasound to detect PLC ranges between 51-71% and specificity ranges between 80-100%<sup>70-72</sup>. Importantly, ultrasound is dependent on operator expertise and is of limited

effectiveness in obese patients. It is usually performed at 6 months intervals for screening high-risk patients<sup>73-76</sup>.

#### 1.8.3. CT scan

Computerised tomography (CT) is a medical imaging procedure that uses x-ray measurements to produce detailed, cross-sectional pictures of the body. It helps to feature the tumour and to determine if it is spreading<sup>70, 77</sup>.

CT scan is used following a lesion identified on liver ultrasound to confirm the diagnosis of HCC by radiological criteria. The sensitivity and specificity of CT are 63-76% and 87-98%, respectively<sup>70, 77, 78</sup>. Due to the increased risks of developing cancer from the ionising radiation from CT scans, this method is generally not used for regular screening for PLC<sup>70, 78</sup>.

#### **1.8.4. MRI scan**

An MRI (magnetic resonance imaging) is a test that uses powerful magnets and radio waves to create detailed cross-sectional pictures of the body<sup>79</sup>. MRIs are increasingly being used to diagnose PLC/HCC due to the high sensitivity (77-90%) and specificity (84-97%), respectively<sup>70, 77, 78</sup>. This investigation is often used as a supplementary to CT scans for diagnosis of PLC<sup>70, 77</sup>. The high costs and time associated with MRIs limit the usefulness of this approach for screening<sup>70, 77</sup>.

#### **1.8.5. Biopsy**

A liver biopsy involves taking a small sample of cells from the affected area for examination of liver histology<sup>80, 81</sup>. It has traditionally been the gold standard approach for assessing liver scarring and for diagnosis of PLC. However, this approach carries a substantial risk of bleeding, iatrogenic injury, and complications<sup>80, 82, 83</sup>. Further, the insertion of a needle into a tumour, the risk of dislodging a cell along the needle path is high<sup>83</sup>. With the advances and availability of imaging techniques, biopsy is now used for less than 5% of HCC diagnoses. Liver biopsy is indicated only when there is uncertainty regarding the results of the AFP test, ultrasound, CT scan/ MRI scans<sup>80, 82, 83</sup>.

#### **1.8.6. Staging of liver cancer**

The stage of PLC involves describing the size of the tumours, the location and spread to other parts of the body<sup>84</sup>. Staging is carried out to inform the treatment plan. Several staging systems have been proposed in HCC management, but there is no globally accepted system. In this thesis, the Barcelona Clinic Liver Cancer (BCLC) staging system is used as it is currently the most widely used internationally as well as in Australia<sup>2</sup>. This system contains four components: the size, number or

spread of the tumour; Child-Pugh score; presence of portal hypertension and the Eastern Cooperative Oncology Group (ECOG) performance scale<sup>84</sup>.

The <b>I</b>	BCLC	staging	system <sup>84</sup>
--------------	------	---------	----------------------

0 (very early stage)	Single tumour less than 2 cm; the liver is working normally without
	portal hypertension (Child-Pugh A)
A (early stage)	Single tumour of any size or up to 3 tumours less than 3 cm; the liver is
	working well (Child-Pugh A or B)
B (intermediate stage)	Many tumours in the liver; single tumour > 5cm, the liver is working
	well (Child-Pugh A or B)
C (advanced stage)	The tumour has grown into one of the main blood vessels of the liver, or
_	spread to the lymph nodes or other body organs; the liver is still
	working well (Child-Pugh A or B)
D (end-stage)	Child-Pugh C with any size tumour

#### Child-Pugh score<sup>84, 85</sup>

The Child-Pugh score is a system for scoring how well the liver is functioning. The system includes the following five factors: bilirubin levels; albumin levels; prothrombin time; ascites; and encephalopathy<sup>84</sup>. Each measure is scored from 1-3, and based on that score, people fall into 1 of 3 categories of the ECOG scale.

А	5-6 points; liver is working well, and cirrhosis is less advanced
В	7-9 points; Liver is working moderately well
С	10-15 points; Liver is not working well, and cirrhosis is advanced

#### The Eastern Cooperative Oncology Group (ECOG) scale<sup>86, 87</sup>

The ECOG scale is a measure used to assess the level of functioning in terms of the daily living abilities of the patients. It is scored from 0 to 5 as follows.

0	Fully active, able to carry on all pre-disease performance without restriction
1	Restricted in physically strenuous activity but ambulatory and able to carry out work of a
	light or sedentary nature, e.g., light housework, office work
2	Ambulatory and capable of all selfcare but unable to carry out any work activities; up and
	about more than 50% of waking hours
3	Capable of only limited selfcare; confined to bed or chair more than 50% of waking hours
4	Completely disabled; cannot carry on any selfcare; totally confined to bed or chair
5	Dead

# 1.9. Surveillance of liver cancer

The WHO's definition of screening is "the presumptive identification of unrecognized disease in an apparently healthy, asymptomatic population by means of tests which can be applied rapidly and easily to the target population"<sup>88</sup>. It is the repeated usage of the screening test at regular intervals to detect disease in latent or asymptomatic patients, for a disease for which an intervention has the potential to alter its course<sup>88</sup>.

The outcomes for PLC patients are highly dependent on the stage at which it is detected. If detected early, HCC is potentially curable<sup>2, 21, 89</sup>. Early identification of small tumours with subsequent treatment has a 5-year survival rate of 50-70%<sup>41, 90, 91</sup>. In contrast, management of patients diagnosed with advanced tumours is palliative<sup>2, 92</sup>. Many studies have identified the effectiveness of HCC surveillance programs<sup>93</sup>.

Many international liver societies recommend surveillance of HCC for high risk populations. For example, the American Association for the Study of Liver Diseases (AASLD) and the European Association for the Study of the Liver (EASL) recommend surveillance every 6 months, consisting of ultrasound  $\pm$  AFP for high risk patients (cirrhosis patients and subgroups of CHB patients)<sup>74, 94</sup>.

In the Australian context, a consensus statement was published in late 2020 by PLC specialists with evidence-based recommendations<sup>95</sup>. This paper states it is now a priority to institute HCC surveillance strategies in Australia for high-risk patients, with the aim of identifying tumours at early stages. The authors recommend surveillance should occur 6-monthly using liver ultrasound and  $\pm$  AFP in<sup>95</sup>:

People with cirrhosis (any aetiology)

Non-cirrhosis patients with CHB infection in:

- Asian men older than 40 years
- Asian women older than 50 years
- People born in sub-Saharan Africa older than 20 years
- Aboriginal and Torres Strait Islander people older than 50 years

## 1.10. The context of primary liver cancer in Australia

#### **1.10.1. Incidence and mortality**

PLC is estimated to be the fourth most common digestive tract cancer and the third most common digestive tract cancer mortality in Australia in 2018<sup>2</sup>. In Australia there are an estimated 2,215 new cases of PLC and 2,088 PLC deaths annually<sup>2</sup>. Incidence in males is 3.4 times higher than for females, and PLC mortality is 2.4 times higher than females in Australia<sup>2</sup>.

Whilst PLC is a relatively rare cancer in Australia, both the incidence and mortality rates have been increasing in recent years. Between 1982 and 2014, ASR incidence increased by over 306% (from 1.8/100,000 persons to 7.4/100,000 persons), and ASR mortality by 184% (from 2.3/100,000 persons to 6.6/100,000 persons)<sup>20</sup>. These trends are in contrast to stable incidence rates for most other cancers, along with their decreasing mortality rates. Whilst this has occurred in the context of improved diagnostic imaging and therapeutic options, the survival time of PLC patients has increased but

remains poor (5 year relative survival rate was  $18\%^2$ ). This is lower compared to 5 year relative survival time for all cancer combined  $(69\%)^2$ .





Figure 4: Incidence trends of liver cancer in Australia from 1982-2015<sup>2</sup>

#### Figure 5: Mortality trends of liver cancer in Australia from 1982-2015<sup>2</sup>

The most common subtype of PLC in Australia is HCC, accounting for 68% of all PLC cases, followed by cholangiocarcinoma type with  $21\%^2$ . The 5-year relative survival rate is highest in HCC patients (22%), whereas lowest in unspecified group (9.5%). These patients belong to this

'unspecified group' when the type of PLC was not specified as the diagnosis were at an advanced stage, and further investigation to identify the type of PLC is often unwarranted as it will have no impact on treatment<sup>2</sup>.

#### 1.10.2. Risk factors for liver cancer in Australia

#### Chronic Hepatitis B

CHB is one of the most common risk factors for PLC. In Australia, there were an estimated 233,947 people living with hepatitis B in 2017<sup>96</sup>. Of those, 90,027 cases (38%) were born in Asia and 26,241 (21%) were Aboriginal and Torres Strait Islanders. The prevalence of Hepatitis B has reduced in younger age groups (under 30), reflecting the impact of the universal infant hepatitis B vaccination program since 2000<sup>96</sup>. Across the 233,947 individuals estimated to be living with CHB, approximately 64% have been diagnosed<sup>96</sup>. Among those, just 18% received regular monitoring, and 8% received antiviral treatment<sup>96</sup>.

#### Chronic hepatitis C

At the end of 2017, there were an estimated 10,537 hepatitis C notifications<sup>96</sup>. Notifications of hepatitis C infections have decreased from 52.6 to 44.2 per 100,000 population between 2008 and 2017<sup>96</sup>. The main route of transmission of CHC in Australia is sharing of injecting equipment in the context of injecting drug use<sup>96</sup>. The highest of hepatitis C notification rates are seen in the Northern Territory (58.0 per 100,000), and lowest in Tasmania (48.6 per 100,000)<sup>96</sup>.

#### Other risk factors

Other risk factors identified in Australia include alcoholic liver disease, obesity, NAFLD and tobacco use<sup>2</sup>. A study in Victoria, Australia has found that alcoholic liver disease was present in 39% of their cohort, and 14% with fatty liver disease<sup>97</sup>.



LIVER CANCER STAGES

Figure 6: Liver cancer stages<sup>98-101</sup>

# **1.11.** What is the need for primary liver cancer research in Tasmania?

In Australia, the incidence and mortality of most cancers have decreased in recent decades. In contrast, both the incidence and mortality of PLC have increased substantially<sup>20, 102</sup>, and this is projected to continue in the coming decades<sup>2, 17</sup>.

This increased burden is associated with increased prevalence of the key risk factors for PLC in Australia: sequelae of excessive alcohol use, CHB, and increasingly NAFLD<sup>2</sup>. Understanding the demographic characteristics of people diagnosed with PLC is required to develop strategies to reduce the burden of PLC. For example, different approaches are required to raise awareness of liver health for patients from different countries and cultures: a one size fits all strategy will not work. To date, the demographic characteristics of PLC cases in Tasmania have been unknown.

In addition, survival of different types of PLC in Tasmania has been unknown. Investigating local data is important in prevention, management and in guiding healthcare policy and research. Thus, understanding the epidemiology of PLC and the factors affecting their survival time is very important as it will have implications for targeting and informing strategies of effective health service to reduce the burden of this cancer in Tasmania.

To provide a comprehensive picture of liver cancer in Tasmania, the quality of cause of death data derived from the TCR and ABS is also investigated. Other studies have shown different cause of death leading to a difference in survival<sup>103, 104</sup>. To-date, no research has been published for PLC examining a) the level of agreement between different agencies coding deaths and medical practitioners, and b) the impact these differences may have on cause-specific survival estimates.

#### **1.12.** Aims of this research

The aims of the studies contained in this thesis are:

- To determine the characteristics and survival time of PLC in Tasmania by subtypes; and
- To estimate the accuracy of cause of death data and the impact on cause-specific survival

The second aim arose from the work undertaken for the first aim. Almost half of all PLC cases were found to have a conflicting cause of death recorded in different datasets. In response, study 2 was undertaken to explore this in the context that previously published studies have reported that cause of death data may be inaccurate and that this can lead to imprecise cause-specific survival estimates<sup>103, 105, 106</sup>.

#### **1.13.** The structure of the thesis

The thesis contains five chapters. The first chapter provides background to understand about PLC, including the descriptions of subtypes of PLC, the incidence and mortality of PLC globally, the demographic profile, risk factors of PLC, the PLC context in Australia, the diagnostic approaches of PLC, staging of PLC and the need for this study.

Chapter 2 describes the methodology used in the two studies.

Chapter 3 presents a draft paper of Study 1 "Characteristics and survival of primary liver cancer patients in Tasmania by subtypes: A data linkage study".

Chapter 4 presents the draft paper of Study 2 "Accuracy of cause of death data: A case study based on primary liver cancer".

Chapter 5 discusses and summarises the whole thesis and offers suggestions for future PLC research.

This thesis is part of a larger study that will develop a patient-centred intervention to support increased participation in liver cancer surveillance. The studies included in this Master's thesis will provide critical information on the epidemiology for diagnosis of PLC in Tasmania and will be used to inform the development of the future surveillance intervention. As such, the primary aim is to determine characteristics and survival time of PLC in Tasmania and the accuracy of PLC data in Tasmania.

# **Chapter 2. Methods**

This chapter provides a brief overview of the study methods, and these are further detailed within Chapters 3 and 4.

# 2.1. Study design

To address the aims of this Master's thesis, a retrospective study design is adopted using linked health administrative datasets. The datasets include:

- Tasmania Cancer Registry (custodian: Tasmanian Department of Health (DOH))
- Tasmanian Emergency Department Presentations (custodian: Tasmanian DOH)
- Tasmania Public Hospital Admissions (custodian: Tasmanian DOH)
- Tasmania Coded Cause of Death (custodian: Australian Bureau of Statistics)

# 2.2. Participants

In Australia, cancer registration is mandatory in all States and Territories. For Tasmania, the TCR is responsible for collecting, coding and reporting incidence, mortality and survival time for all malignant neoplasms within the state<sup>107</sup>.

The participants in this study were defined as any individual aged 18 years or older with a confirmed PLC (ICD-10 code of C22.\*) diagnosed in Tasmania between 1/1/2007 and 31/12/2015.

## 2.3. Data Linkage process

From the year 2013, the TCR used radiological and clinical diagnoses for HCC (coded as 81703-Hepatocellular carcinoma). Then from the year 2017, according to changes in coding rules in other states, cases with 8003 (Malignant neoplasm) were coded as HCC. As a result, the number of tumours coded to 81703 dropped substantially. Changes in TCR staff and lack of documentation of coding rules would have impacted the number of tumours coded to 81703 from 2013 onwards. The flowchart below describes the processes for all stages of the preparation of linked data.

These detailed steps were described in the method of Chapter 3.


Figure 7: Data linkage process

## 2.4. Diagnostic coding of liver cancer

The International Classification of Diseases and Related Health Problems 10th Revision (ICD-10) is a standard that is used to code for all health conditions, including in this context, PLC and its subtypes<sup>108</sup>. This version has been used for coding in the Australian health system since 1997<sup>4</sup>. The codes used to identify cases were:

- C22.0 Liver cell carcinoma
- C22.1 Intrahepatic bile duct carcinoma
- C22.2 Hepatoblastoma
- C22.3 Angiosarcoma
- C22.4 Other sarcomas of liver
- C22.7 Other specified carcinomas of liver
- C22.8 Malignant neoplasm of liver, primary, unspecified as to type
- C22.9 Malignant neoplasm of liver, not specified as primary or secondary

The International Classification of Diseases for Oncology, 3rd Revision (ICD-O3) is an extension of the ICD coding standard for tumour diseases. There is a formal agreement between the International Association for Cancer Registries, the International Agency for Research on Cancer, and the WHO that registered cancer cases should be coded according to the ICD-O3<sup>84</sup>. It is a dual classification system, including topography and morphology of tumours<sup>84</sup>. Thus, each cancer case in the TCR was coded by trained registry staff according to the ICD-O3, based on the information gathered from medical records. For PLC, the relevant ICD-O3 codes are:

Topography codes (indicate the site of origin of a neoplasm)<sup>84</sup>

- C22.0 Liver
- C22.1 Intrahepatic bile duct

Morphology codes record the type of cell that has become neoplastic and its biologic activity. There are three parts to a complete morphology code: the first four digits indicate cell types (histology); the fifth digit indicates the behaviour, and the last digit indicates the grade of the tumour.

Morphology code	ICD-O3 morphology code description <sup>84</sup>
8000	Neoplasm, malignant
8010	Carcinoma, not otherwise specified (NOS)
8033	Pseudosarcomatous carcinoma
8140	Adenocarcinoma, NOS
8160	Cholangiocarcinoma
8162	Klatskin tumor
8170	Hepatocellular carcinoma, NOS
8173	Hepatocellular carcinoma, spindle cell variant

8180	Combined hepatocellular carcinoma and cholangiocarcinoma
8890	Leiomyosarcoma, NOS
8970	Hepatoblastoma
9120	Hemangiosarcoma
9680	Malignant lymphoma, large B-cell, diffuse, NOS
Behaviour code	ICD-O3 behaviour code description <sup>84</sup>
0	Benign
1	Uncertain whether benign or malignant Borderline malignancy Low malignant potential Uncertain malignant potential
2	Carcinoma in situ Intraepithelial Noninfiltrating Non invasive
3	Malignant, primary site
Grade code	ICD-O3 grade code description <sup>84</sup>
1	Grade I Well differentiated, Differentiated, NOS
2	Grade II Moderately differentiated Moderately well differentiated Intermediate differentiation
3	Grade III Poorly differentiated
4	Grade IV Undifferentiated Anaplastic
9	Grade or differentiation not determined, not stated or not applicable

## 2.5. Data collection from each dataset

The variables collected four different sources are detailed in this table:

Variables	Description				
1. Tasmania Public Hospital Admissions (custodian: Tasmanian DOH)					
PPID	A project-specific, unique pseudo identifier that is supplied to researchers that refers to an individual with minimal risk of re-identification.				
Date of Birth	Date				
Sex	<ol> <li>Male</li> <li>Female</li> <li>Indeterminate</li> <li>Not stated/inadequately described</li> </ol>				
Date of Death					
Area of Usual Residence SA2	An SA2 is identifiable by a 9-digit fully hierarchical code. The SA2 identifier is a 4-digit code, assigned in alphabetical order within a Statistical area level 3. An SA2 code is only unique within a state/territory if it is preceded by the state/territory identifier.				
Diagnosis related group version 7	A patient classification scheme which provides a means of relating the number and types of patients treated in a hospital to the resources required by the hospital, as represented by a code.				
Procedure Code 1 to 50	Record and code all procedures undertaken during the episode of care. Procedures are derived from and must be substantiated by clinical documentation.				
Procedure Block 1 to 50	The urgency related group major diagnostic block category into which the patient's emergency department diagnosis is grouped, as represented by a code.				

#### Table 1: Data collection from each dataset

Diagnosis Code 1 to 100	The category into which the patient's diagnosis and the associated					
	The diagnosis codes must be valid codes from the current edition of the					
	ICD-10.					
2. Tasmania Cancer Registry	(custodian: Tasmanian DOH)					
PPID						
Date of Birth						
Sex						
Country of birth	The country in which the person was born, as represented by a code according to the Standard Australian Classification of Countries 2016 (SACC).					
Date of Death	The date of death of the person, expressed as DD MM YYYY					
Age at Death	The age of death of the patient					
Place of Death	The place of death of the patient					
The Statistical area level 2 (SA2)	As described above					
Date of Diagnosis	The date on which the patient was first diagnosed with cancer.					
Age at Diagnosis	The age of patients when they were diagnosed					
Primary Site ICD-O3	C22.0 Liver					
	C22.1 Intrahepatic bile duct					
Primary Site ICD-10	The subtypes of PLC record in TCR data					
Morphology Code	As mentioned above					
Dasis of Diagnosis	<ul> <li>first presentation, as represented by a code.</li> <li>1 Clinical Only</li> <li>2 Clinical Investigation - (including diagnostic techniques i.e. x-ray, endoscopy, imaging, ultrasound and exploratory surgery)</li> <li>4 Specific biochemical and/or immunological tests</li> <li>5 Cytology or haematology - (examination of cells including fluids aspirated by endoscopy or needle, including peripheral blood &amp; bone marrow aspirates)</li> <li>6 Histology</li> <li>7 Death Certificate Only - Information provided is from a death certificate.</li> </ul>					
Spread	<ul> <li>9 Unknown</li> <li>1 Localised to the tissue of origin (includes insitu breast and insitu</li> </ul>					
	<ul> <li>melanoma)</li> <li>Invasion of adjacent tissue or organs (includes subcut fat or muscle and organs adjacent to the primary cancer site)</li> <li>Regional lymph nodes</li> <li>Distant metastases</li> <li>Not applicable. Applies to lymphatic and haemopoetic cancers.</li> <li>Unknown</li> </ul>					
Cause of Death ICD-O3						
Cause of Death ICD-10						
3. Tasmanian Emergency De	partment Presentations (custodian: Tasmanian DOH)					
PPID						
Date of Birth						
Sex						
Country of Birth						

A 611 1D 11				
Area of Usual Residence				
SA2				
Type of Visit	The reason the patient presented to an emergency department, as			
	represented by a code.			
	I Emergency presentation			
	2 Return visit, planned			
	3 Pre-arranged admission			
	5 Dead on arrival			
Diagnosis Classification	The type of classification used for recording emergency department			
Туре	diagnosis, as represented by a code.			
Principal Diagnosis Code	The diagnosis established at the conclusion of the patient's attendance in an			
	emergency department to be mainly responsible for occasioning the			
	attendance following consideration of clinical assessment, as represented by			
	a code.			
Major Diagnosis Block	The urgency related group (URG) major diagnostic block category into			
	which the patient's emergency department diagnosis is grouped, as			
	represented by a code.			
4. Tasmania Coded Cause of Death (custodian: ABS)				
PPID				
Date of Birth				
Sex				
URES_SA2	Australian Statistical Geography Standard code denoting the ceased usually			
	resided. Usual residence refers to that address at which the deceased has			
	lived or intends to live for a total of six months or more in a given reference			
	year.			
Date of Death				
Age at death				
Underlying cause of death	The disease or injury which initiated the train of morbid events leading			
	directly to death. Accidental and violent deaths are classified according to			
	the external cause, that is, to the circumstances of the accident or violence			
	which produced the fatal injury rather than to the nature of the injury.			
	Underlying cause is recorded as four digits.			
RACS 1 to 20	A count of the number of causes recorded in the record axis data field after			
	application of the ICD-10 coding rules and procedures for the selection of			
	underlying and associated causes of death for mortality tabulation.			

## 2.6. Data analysis

<u>Study 1:</u> Characteristics and survival of primary liver cancer patients in an Australian jurisdiction: A data linkage study.

**Aim 1:** Present demographic characteristics by subtypes of PLC patients in Tasmania (2007-2015): ANOVA were used to examine the differences in age at diagnosis. The log-rank test and Kaplan Meier method were used to compare median survival between subtypes, urban and rural residence and country of birth.

**Aim 2:** Estimate relative survival by PLC subtypes and demographic factors. The Ederer II method based on the algorithm developed by Dickman was used to estimate 1-,3- and 5- year relative survival

rates<sup>109</sup>. This method is commonly used for estimating patient survival for population based registries when specific cause of death is unknown. It was calculated as the ratio of the observed survival rate in the cohort relative to the expected survival rate taken from the Tasmania population over the same period.

**Aim 3:** Describe the prevalence of PLC risk factors among patients with PLC in Tasmania: A predefined list of risk factors for PLC was developed according to the ICD-10<sup>108</sup>. These risk factors codes were searched for during each patient's admission, discharge, examination as well as in the TCR and ABS datasets.

Specific details for the analyses used in this study are described in Chapter 3.

Study 2: Accuracy of cause of death data: A case study based on primary liver cancer

**Aim 1:** Investigate the level of agreement on cause of death between the ABS and TCR, along with medical practitioners' opinions: Cohen's kappa statistics were used to assess the concordance by chance alone between the different methods of ascertaining cause of death<sup>110</sup>.

**Aim 2**: Measure the impact of different coding practices and resulting cause of death data on causespecific survival: The cumulative incidence function was used to estimate deaths caused by non-PLC cases based on the final causes of death provided by the medical practitioners in the presence of competing risks<sup>111</sup>. This is preferred to the traditional Kaplan-Meier method or net survival as it provided the appropriate framework to analyse the interplay between deaths from liver cancer and other competing risk factors based on different coding practices.

Specific details for the analyses used in this study are described in Chapter 4. All analyses were conducted with Stata 15 (Ver.15, College Station, Texas, USA).

## 2.7. Ethics

Ethics approval for this study was provided by the Tasmanian Human Research Ethics Committee (Ethics reference number is H0016958).

## Chapter 3. Characteristics and survival of primary liver cancer patients in Tasmania by subtypes: A data linkage study

## **3.1.** Abstract

*Aims:* Describe the demographic characteristics, risk factors and estimate survival time by subtypes for all PLC patients in Tasmania between 2007 and 2015.

*Methods*: Data for all PLC patients in Tasmania, Australia between 2007 and 2015 were obtained from the TCR and linked to public hospital inpatient data. Demographic characteristics and risk factors were examined using descriptive statistics. Median survival was estimated using log-rank test and the Kaplan-Meier method; relative survival time was estimated using the Ederer II method.

*Results*: 293 PLC cases were identified. HCC (51.9%) and cholangiocarcinoma (20.5%) represented the majority of cases, along with 24.9% of 'unspecified' cases. At least one risk factor was reported for 57.0% of cases, most commonly cirrhosis (37.2%), diabetes (26.7%), alcoholic related liver disease (22.5%) and viral hepatitis (16.7%). Mean age at diagnosis for all cases was 69.6 years (95% confidence interval (CI): 68.3 - 70.9), and median survival cases was 6.2 months (95% CI: 4.1-8.3). The 1-, 3- and 5-year relative survival rates were 38.3% (95% CI: 32.0-44.5); 13.8% (95% CI: 9.0-9.8) and 6.7% (95% CI: 2.9-13.0) respectively, with a trend toward increasing survival time over the reporting period.

*Conclusions*: HCC and cholangiocarcinoma were the main types of PLC reported in Tasmania. The identified risk factors can be used to inform interventions to improve early diagnosis. In turn, this would improve the poor survival outcomes for patients diagnosed with PLC.

Keywords: cholangiocarcinoma, hepatocellular carcinoma, liver cancer, risk factor, survival rate.

## **3.2.** Introduction

PLC is the sixth most common cancer globally, accounting for 4.7% of cancer incidence<sup>17</sup>. It is the fourth leading cause of cancer deaths, accounting for 8.2% of all cancer deaths in 2018<sup>17</sup>. The predominant form of PLC is HCC, representing 75%-85% of all cases<sup>2, 17, 21</sup>. This is followed by intrahepatic cholangiocarcinoma (10%-15% of cases), and rare forms of PLC including hepatoblastoma and angiosarcoma<sup>2, 17, 21</sup>.

The most common risk factors for PLC are chronic viral infections caused by the CHB and CHC. These viral infections lead to progressive liver scarring (cirrhosis), which is a common precursor for HCC and is present in 70%-90% of all HCC cases<sup>2, 21</sup>. Worldwide, 80% of PLC cases are attributable to HBV and HCV infections<sup>2</sup>. More recently, NAFLD has emerged as a risk factor for HCC, particularly in high-income countries<sup>64, 112</sup>. NAFLD represents a spectrum of liver diseases, ranging from simple steatosis to decompensated cirrhosis<sup>51</sup>. Prevalence of NAFLD has increased in parallel with rates of obesity<sup>53</sup>. In the US, NAFLD-associated cirrhosis is now the second most common indication for liver transplant<sup>113</sup>. For cholangiocarcinoma, in addition to HBV and HCV, primary sclerosing cholangitis is a risk factor<sup>2, 4</sup>. Liver flukes are an important risk factor in Southeast Asian countries. This parasitic worm is ingested when eating raw or uncooked freshwater fish. They subsequently infect the liver, gallbladder and bile duct. Untreated, the infection can persist for the lifetime of the parasite: 25-30 years<sup>114</sup>.

As HCC is the most common form of PLC, this is the primary focus of this research. HCC incidence and mortality rates vary markedly by region and country, reflecting the heterogeneity in the distribution and natural history of risk factors<sup>2</sup>. Globally, males consistently have incidence rates two-three times higher than females: partly explained by higher levels of androgen in males, which promotes development of HCC<sup>2, 21</sup>. In contrast, higher levels of estrogen in females may inhibit HCC through its anti-inflammatory effects. Further, CHB, CHC, and alcohol consumption are more prevalent in males and contribute to higher rates of HCC<sup>2, 21</sup>.

Whilst approximately 80% of the burden of PLC is observed in low and middle-income countries, Australia has experienced an increasing burden<sup>2, 4, 17</sup>. Between 1982 and 2014, ASR incidence increased by over 300% (from 1.8/100,000 to 7.4/100,000), and ASR mortality by almost 200% (from 2.3/100,000 to 6.6/100,000 )<sup>17</sup>. These trends are in contrast to decreasing rates of most other cancers in Australia, with this trend expected to continue over coming decades<sup>2, 18, 20</sup>. Despite advances in treatment, the 5-year relative survival rate for Australians diagnosed with PLC is 18.1%<sup>2</sup>.

Although recent population-level research on PLC has been conducted elsewhere in Australia<sup>20, 30, 97, 115-117</sup>, in Tasmania this is limited to the TCR's annual report of incidence and mortality<sup>107</sup>. As risk factors for PLC vary by regions (e.g., migration from hepatitis endemic countries), understanding the burden in Tasmania has the potential to contribute to development of targeted local interventions. This study aimed to 1) present demographic characteristics by subtypes of PLC patients in Tasmania (2007-2015); 2) estimate relative survival by PLC subtypes and demographic factors; 3) describe the prevalence of risk factors for PLC in Tasmania.

## **3.3.** Methods

A retrospective data linkage study was conducted using de-identified datasets from the TCR and inpatient admissions from Tasmanian public hospitals from 2007-2015.

#### Data sources

The cohort was provided by the TCR, a population-based registry responsible for collecting and reporting cancer incidence and mortality data The TCR is notified of all cancer diagnoses made in Tasmania by hospitals, pathology, and radiology laboratories.

All incident cases of PLC were defined according to the International Classification of Diseases for Oncology, Third Revision (ICD-O3)<sup>84</sup>. The ICD-O3 coded tumours were converted to ICD-10 (the International Statistical Classification of Diseases and Related Health Problems Tenth Revision). All tumours with a basis of diagnosis other than histology were classified as 'unspecified'. PLC subtypes were categorised into four groups according to ICD-10: HCC (C22.0), intrahepatic cholangiocarcinoma (C22.1) (referred to as cholangiocarcinoma in this study), PLC where the subtype was not specified (C22.9) ('unspecified'). As the number of each remaining type was <5, these cases were combined into 'other' (hepatoblastoma, angiosarcoma of liver, other sarcomas of liver, other specified carcinomas of liver).

Patient data was extracted from the TCR, including country of birth, place of residence, date of diagnosis, primary site ICD-O3, ICD-10 codes, basis of diagnosis, spread of disease, tumour grade, tumour size, cause and date of death.

De-identified administrative data for all admissions to Tasmanian public hospitals were linked to the TCR dataset (see Data Linkage Process below). For each admission, up to 56 ICD-10 codes were reported. As almost all of the TCR cohort had a public hospital admission, these codes were used to identify risk factors for PLC.

Place of residence was categorised as either urban or rural according to the Australian Statistical Geography Standard 2016<sup>118</sup>. Country of birth was coded based on the Standard Australian Classification of Countries 2016 and later categorized into Australian-born or overseas-born<sup>119</sup>. A more detailed breakdown by country or region was not reported to maintain anonymity for the small number of cases for which this was reported.

#### Data Linkage Process

The data linkage process was carried out in three steps. Firstly, the TCR identified the cohort as individuals aged 18 years and above with a confirmed PLC (ICD-10 code of C22.\*) diagnosed between 1/1/2007 and 31/12/2015. Patient identifiers were provided to the Tasmania Data Linkage Unit for probabilistic linkage to identifiers from the Master Linkage Map. The Linkage Unit generated Project Person Identifiers (PPIDs) which were supplied to the data custodians for the extraction of data. Finally, data custodians provided the data to the researchers.

Ethical approval for this study was obtained by the University of Tasmania Human Research Ethics Committee.

29

#### Statistical analysis

Demographic characteristics were summarized using descriptive statistics. Differences in age at diagnosis were examined by ANOVA. Survival time was calculated from date of diagnosis until date of notified death: those cases that were alive on 31 December 2015 were censored in the survival analysis. Those with the date of death equal to the date of diagnosis were excluded from all survival tests.. The log-rank test and Kaplan Meier method were used to compare median survival between histologic subtypes, urban and rural residence and country of birth.

The 1-, 3- and 5-year relative survival rates were calculated using the Ederer II method, based on an algorithm developed by Dickman<sup>109</sup>. Relative survival was calculated as the ratio of the observed survival rate in the cohort relative to the expected survival rate taken from the Tasmania population from 2007 to 2015. Life table data for Tasmanians was taken from the ABS for 2007 to 2015<sup>120</sup>. Relative survival rates were estimated for all cases and between different subtypes of PLC. As recent studies have suggested improved survival time from 1982 to 2014<sup>2, 97, 121</sup>, survival was estimated for 2007-2010 and 2011-2015 to assess for any differences.

A pre-defined list of risk factors for PLC was developed based on published literature. This included HBV<sup>122</sup>, HCV<sup>123</sup>, alcoholic related liver disease<sup>124</sup>, NAFLD<sup>125</sup>, diabetes<sup>126</sup>, and cirrhosis<sup>122</sup>.

Data was analysed using Stata 15 (Ver.15, College Station, Texas, USA). Any cell counts  $\leq$ 5 cases were suppressed to prevent any possibility of identification. A p-value <0.05 was deemed statistically significant.

## **3.4.** Results

#### Demographic characteristics

Between 2007-2015, 293 patients were diagnosed with PLC in Tasmania. Most were diagnosed with HCC (51.9%, n=152) and cholangiocarcinoma (20.5%, n=60) (Table 2). 24.9% (n=73) were reported as 'unspecified' which likely reflects these patients were diagnosed at an advanced stage when further investigation is unwarranted as it will not impact treatment. This assumption is based on discussions with clinicians and data custodians. The remaining cases were classified as 'other' (2.7%, n=8); owing to the small number in this group, we focused on the three more common forms (Table 2). Of the total cohort, 239 patients (81.6%) died during this study period. Mean age at diagnosis for all cases was 69.6 years (95%CI: 68.3-70.9). Breaking this down by subtype, mean age at diagnosis for HCC was 66.5 years (95%CI: 64.7-68.2), and for cholangiocarcinoma was 72.2 years (95%CI: 69.4-75.0) and for the unspecified group 74.2 years (95%CI: 71.6-76.8). Comparisons between ages at diagnosis showed a statistically significant difference between HCC and cholangiocarcinoma (p=0.001), and HCC and the unspecified group (p<0.001) (Table 2).

The majority of all PLC cases were male (74.1%, n=217). Whilst large majorities of HCC and unspecified cases were male, this was not observed for cholangiocarcinoma (53.3% male). Reflecting the population distribution<sup>127</sup>, a greater number of cases across all types of PLC was observed for people living in urban settings (62.1%, n=182) compared to rural areas (37.9%, n=111). Country of birth information was available for 68.6% (n=201): of these over 83.1% (n=167) were born in Australia.

Regarding the basis of diagnosis, from 2007-2015, clinical investigation (e.g., imaging) was the main method (44.7%, n=131). Histology of the primary tumour was the second most common method (36.9%, n=108). Data on spread of tumour was available for 29.4% (n=86) of cases, of these 44.2% (n=38) had distant metastases, 24.4% (n=21) had either localised tumours or PLC with invasion of adjacent tissues/organs respectively. Similarly, for grade and size of tumour, 81.5% (n=238) and 82.9% (n=243) of the data was unavailable (Table 2).

Variables	T 	otal =293)	H (n=	CC =152)	Cholangiocar	rcinoma (n=60)	Uns (	pecified n=73)	Otl (n:	ner =8)
Age at diagnosis (mean, 95% CI) <sup>a</sup>	69.6 (6	8.3-70.9)	66.5 (6	4.7-68.2)	72.2 (6	9.4-75.0)	74.2 (	71.6-76.8)	69.2 (63	.3-75.1)
Sex (n, %)										
Males	217	(74.1)	128	(84.2)	32	(53.3)	53	(72.6)	n/r	n/r
Females	76	(25.9)	24	(15.8)	28	(46.7)	20	(27.4)	n/r	n/r
Place of residence (n, %)										
Rural	111	(37.9)	63	(41.4)	22	(36.7)	20	(27.4)	6	(75.0)
Urban	182	(62.1)	89	(58.6)	38	(63.3)	53	(72.6)	n/r	n/r
Country of birth (n, %)										
Australia born	167	(57.0)	69	(45.4)	44	(73.3)	49	(67.1)	n/r	n/r
Overseas born	34	(11.6)	18	(11.8)	6	(10.0)	9	(12.3)	n/r	n/r
Not stated/missing	92	(31.4)	65	(42.8)	10	(16.7)	15	20.5	n/r	n/r
Basis of diagnosis (n, %)										
Clinical investigation	131	(44.7)	65	(42.8)	19	(31.7)	46	(63.0)	n/r	n/r
Biochemical and/or immunological tests	8	(2.7)	n/r	n/r	n/r	n/r	6	(8.2)	n/r	n/r
Cytology or haematology	18	(6.1)	9	(5.9)	8	(13.3)	n/r	n/r	n/r	n/r
Histology	108	(36.9)	68	(44.7)	33	(55.0)	n/r	n/r	7	(87.5)
Death certificate only	20	(6.8)	n/r	n/r	n/r	n/r	16	(21.9)	n/r	n/r
Unknown	8	(2.7)	n/r	n/r	n/r	n/r	n/r	n/r	n/r	n/r
Spread (n, %)		× ,								
Localised tumour	21	(7.2)	12	(7.9)	9	(15.0)	n/r	n/r	n/r	n/r
Invasion of adjacent tissue/ organs	21	(7.2)	9	(5.9)	6	(10.0)	n/r	n/r	n/r	n/r
Regional lymph nodes	6	(2.0)	n/r	n/r	n/r	n/r	n/r	n/r	n/r	n/r
Distant metastases	38	(13.0)	19	(12.5)	9	(15.0)	9	(12.3)	n/r	n/r
Unknown	207	(70.6)	109	(71.8)	34	(56.7)	58	(79.5)	6	(75.0)
Grade (n, %)										
I (Well differentiated)	21	(7.2)	18	(11.9)	n/r	n/r	n/r	n/r	n/r	n/r
II (Moderately differentiated)	15	(5.1)	6	(4.0)	9	(15.0)	n/r	n/r	n/r	n/r
III (Poorly differentiated)	18	(6.2)	8	(5.3)	8	(13.3)	n/r	n/r	n/r	n/r
Unknown	238	(81.5)	119	(78.8)	40	(66.7)	72	(98.6)	7	(87.5)

 Table 2: Demographic characteristics of liver cancer patients in Tasmania between 2007 and 2015

*n/r:* not reported due to small numbers ( $\leq$ 5), confidentiality or other concerns about the quality of the data

During the study period, 88.1% (n=258) of cases had at  $\geq 1$  public hospital admission. For this group, 57.0% (n=147) had at least one or more risk factors reported in their hospital data. The mean number of risk factors was 1.3 (SD=1.4), with the most frequently reported being cirrhosis (37.2%, n=96), chronic viral hepatitis (35.0%, n=89), diabetes (26.7%, n=69) and alcohol-related liver disease (22.5%, n=58).

	<b>Risk factors</b>	Total	(n=258)	HCC	(n=137)	Cholan	giocarcin	Unsp	ecified	Ot	her
						oma	(n=51)	(n	=64)	( <b>n</b> :	=6)
		n	%	n	%	n	%	n	%	n	%
1	Cirrhosis	96	(37.2)	72	(52.6)	n/r	n/r	20	(31.3)	n/r	n/r
2	Diabetes	69	(26.7)	45	(32.8)	8	(15.7)	14	(21.9)	n/r	n/r
3	Alcoholic related	58	(22.5)	44	(32.1)	n/r	n/r	14	(21.9)	n/r	n/r
	liver disease										
4	Viral hepatitis	43	(16.7)	36	(26.3)	n/r	n/r	6	(9.4)	n/r	n/r
5	Chronic hepatitis C	38	(14.7)	32	(23.4)	n/r	n/r	n/r	n/r	n/r	n/r
6	Obesity	9	(3.5)	8	(5.8)	n/r	n/r	n/r	n/r	n/r	n/r
7	Chronic hepatitis B	8	(3.1)	7	(5.1)	n/r	n/r	n/r	n/r	n/r	n/r
8	Non-alcoholic fatty	6	(2.3)	n/r	n/r	n/r	n/r	n/r	n/r	n/r	n/r
	liver disease										
9	Had at least 1 risk	147	(57.0)	100	(73.0)	12	(23.5)	32	(50.0)	n/r	n/r
	factor										

Table 3: Prevalence of risk factors of liver cancer patients by inpatient admission ICD-10 codes

*n/r:* not reported due to small numbers ( $\leq$ 5), confidentiality or other concerns about the quality of the data

#### Survival time

#### Median survival time

Between 2007 and 2015, 81.6% (n=239) of patients were deceased in Tasmania, in which 46.1% (n=111) were reported with HCC, 24.6% (n=72) in the 'unspecified group', and 20.1% (n=50) with Cholangiocarcinoma. The remaining, 2.5% (n=6) belonged to the 'other' type of liver cancer.

For all PLC cases, median survival (n=260) was 6.2 months (95% CI: 4.1-8.3) (Figure 8). Cases excluded from this analysis included those diagnosed at death (8.4%, n=25) and eight cases that were classified as 'other' types of cancer.

Median survival was similar for HCC (8.8 months, 95%CI: 4.2-13.5) and cholangiocarcinoma (9.2 months, 95%CI: 6.2-12.1). For 'unspecified' cases, median survival was 2.7 months (95%CI: 1.8-3.6) (Figure 9), reflecting the nature of the grouping.

Median survival across all PLC cases was similar for males (5.9 months, 95% CI: 3.8-7.9) and females (8.1 months, 95% CI: 3.0-13.1), and for those residing in rural (8.2 months, 95% CI: 3.3-13.2) and

urban areas (4.7 months, 95% CI: 3.0-6.6). In terms of country of birth, median survival was 3.4 months (95% CI: 2.0-4.7) for those born overseas and 4.8 months (95% CI: 2.9-6.7) for people born in Australia. Between 2007-2010 and 2011-2015, a potential trend toward increasing median survival was observed, increasing from 5.7 months (95% CI: 2.7-8.6) to 6.8 months (95% CI: 2.1-11.5). The overlapping confidence intervals indicate no real difference was observed.



Figure 8: Kaplan-Meier survival curve for all liver cancer cases from the time of diagnosis between 2007 and 2015



Figure 9: Kaplan-Meier survival curves for those diagnosed with liver cancer by subtypes

#### Relative survival rates

The 1-, 3- and 5-year relative survival rates for all cases across the study period were 38.3% (95%CI: 32.0-44.5); 13.8% (95%CI: 9.0-19.8) and 6.7% (95%CI: 2.9-13.0), respectively. A trend towards improved relative survival was observed for the 2011-2015 period compared with 2007-2010 for all cases as well as for each subtype (Table 4). Specifically, 1-year survival increased from 32.9% (95%CI: 23.9-42.3) to 42.4% (95%CI: 33.9-50.6), the 3-year survival from 10.7% (95%CI: 5.5-18.0) to 18.6% (95%CI: 10.5-28.7), and 5-year survival increased from 4.7% (95%CI: 1.5-10.8) to 13.5% (95%CI: 4.1-29.0). Again, the overlapping 95% CIs mean these results should be interpreted with caution.

Relative survival rate	1 year	3 year	5 year
All cases (%, 95% CI)	38.3 (32.0-44.5)	13.8 (9.0-19.8)	6.7 (2.9-13.0)
Time period (%, 95%)			
2007-2010 (n=104 cases)	32.9 (23.9-42.3)	10.7 (5.5-18.0)	4.7 (1.5-10.8)
2011-2015 (n=156 cases)	42.4 (33.9-50.6)	18.6 (10.5-28.7)	13.5 (4.1-29.0)
<b>Subtypes</b> (%, 95% CI)			
HCC	47.4 (38.6-55.8)	16.0 (8.9-25.1)	11.6 (4.6-22.6)
Cholangiocarcinoma	41.6 (28.4-54.5)	13.8 (5.1-27.1)	6.4 (1.0-20.1)
Unspecified	12.0 (4.9-22.5)	6.6 (1.7-16.3)	0.0

Table 4: 1-, 3- and 5-year relative survival for patients diagnosed with liver cancer from 2007-2015

## 3.5. Discussion

This study is the first to examine demographic characteristics, risk factors and survival time for PLC patients in Tasmania. Our results show a potential trend of higher relative survival across the study period from 2007 to 2015. However, PLC remains a low survival cancer, and in addition, almost 10% of cases were diagnosed only at time of death. Diagnosed early, curative treatment for PLC is possible. Our data clearly indicated that this is not occurring in the majority of cases, despite recommendations that people at high-risk of developing PLC should be regularly screened<sup>128, 129</sup>.

This study investigated 293 PLC cases from 2007 to 2015. Our results highlighted that the most frequently occurring subtypes were HCC and cholangiocarcinoma. These results are similar to previous reports. A recent Australian report using cancer registry data reported HCC accounted for 67.6% of all liver cases, and cholangiocarcinoma 25.3%<sup>2</sup>; for HCC this is higher than our estimate for

HCC (51.9%). Recent work has demonstrated studies using Australian cancer registry data are likely to underestimate the number of HCC cases<sup>107, 116</sup>. For example, coding in the TCR, is based on the ICD-O3 topography code (C22.\*) and morphology codes. In the absence of histologic data to accurately describe morphology, patients are coded as C22.\*, and a morphology code of 80003 (neoplasm, malignant), i.e., unspecified. When changing from ICD-O3 to ICD-10, the C22.\*, 80003 code is recoded as unspecified liver cancer (C22.9). This is likely to explain the differences in results. A recent Australian study highlighted this issue<sup>107</sup>, with recoding of data resulting in 75.9% of cases originally coded as C22, 80003 morphology being recorded as HCC (C22.0, 81703 morphology).

As an example of these coding issues, a recent study using data from the Australian Cancer Database to assess incidence of HCC between 1982 and 2014 also reported data quality and registration methodology were heterogeneous<sup>116</sup>. These authors reported data from two different registries included relatively high rates (41% and 29%) of cases coded as C22.\*, 80003 (unspecified type), with the remaining registries reporting this code for 0.5%-7.2% of cases. Based on the results of Hong<sup>107</sup>, the authors adopted the assumption all unspecified cases were HCC in order to estimate survival time<sup>116</sup>. In our study, the rate of 'unspecified' liver cancer was relatively high (24.9%), and the mean age at diagnosis was significantly higher than for HCC. Therefore, we assume that the number of HCC cases has been underestimated due to coding practices.

In this study, cirrhosis was reported in 37.2% (n=96) of cases, viral hepatitis (not further specified) in 16.7% (n=43), HCV in 14.7% (n=38), and HBV in 3.1% (n=8) of cases. These results differ from those of a recently published study in Victoria<sup>97</sup>. This study reported on HCC patients, with 22.1% having HBV and 41.2% with HCV. This difference can be largely explained by the different rates of HBV in these jurisdictions, with 28.2/100,000 notifications in Victoria, compared with 9.5/100,000 in Tasmania in 2017<sup>130</sup>. To some degree, this difference is related to the different rates of migration in these states, with Tasmania having a lower rate of migrats from HBV endemic countries. This is not the case for HCV, with 31.0/100,000 notifications of HCV in Victoria, compared with 45.9/100,000 people in Tasmania in 2017<sup>96</sup>. Another difference observed was our study reported just 2.3% of cases had a recorded diagnosis of NAFLD, compared with 14.3% of cases in this Victorian study. This is in contrast to the prevalence of overweight and obesity, with 70.9% of Tasmanians and 68.3% of Victorians in these categories in the 2017-2018<sup>131</sup>. This difference, therefore, may be related to assessment and diagnosis, documentation and coding of NAFLD.

In our study, the overall median survival time from diagnosis was 6.2 months and the 5-year relative survival rate was 6.7% between 2007 and 2015. A recently published estimate of PLC survival in Australia reported median survival time increased from 2.1 months (95% CI: 1.6-2.6 months) to 12.1 months (95% CI: 11.2-13.0) during 2010-2014<sup>116</sup>. Whilst improved survival over time was observed, the large confidence intervals suggest this result should be interpreted with caution. Similarly, HCC

and cholangiocarcinoma cases had lower 5-year relative survival rates in our study with 11.6% and 6.4% respectively, compared to national data reporting 5-year survival over 2010-2014 for HCC as 22.1% and 9.0% for cholangiocarcinoma. Part of this difference may be related to the time periods: our data reflects 5-year survival over 2007-2015 as we had insufficient numbers of cases to conduct analyses of cancer subtypes for 2007-2010 and 2011-2015<sup>2</sup>. The other possibility is aetiology and risk factors. In Tasmania, due to low immigration rates, HCC related to non-cirrhotic hepatitis B is rare, which is reflected in our data. The survival of patients with HCC is dependent on tumour number, size and liver disease status. In our cohort, 37.2% had cirrhosis, which can limit treatment options and survival.

Previous Australian research has reported superior survival outcomes for patients born in Asia compared to those born in Europe or Oceania<sup>30, 116, 121</sup>. This disparity may be partly related to the majority of Asian-born patients originating from HBV endemic countries: it is possible that patients with HBV-related HCC may not have cirrhotic livers, and therefore more treatment options are available. Survival was not impacted by being born overseas, current policies that are resulting in increased migration to Tasmania, particularly from countries with high prevalence of HBV and HCV, may impact this<sup>116, 130</sup>. A such, it will be important to monitor health outcomes to prevent avoidable differences in the population.

#### **Strengths and limitations**

This study has several strengths. Data from the TCR with full coverage of the Tasmanian population was used and linked to Tasmanian public hospital admitted patient episodes which provided valuable and reliable data. A further strength was using ICD-10 coded data, which served as a useful tool for consistent classification of PLC by subtype. Notwithstanding this, several limitations must be noted. Firstly, a considerable proportion of unspecified cases may have resulted in an underestimation of the true proportion of each PLC subtype. A further limitation related to use of administrative admitted patient data pertaining to risk factors relates to ICD-10-AM coding. Whilst up to 56 ICD-10-AM codes for each patient admission are possible, typically the only codes recorded are those relevant to the admission. For example, a CHB patient admitted with a fractured femur is unlikely to have CHB recorded. Similarly, we expect that NAFLD was under-reported in our data, particularly in the context of the high rate of overweight and obesity in Tasmania. Therefore, we expect that we have underestimated the prevalence of risk factors such as cirrhosis NAFLD due to the completeness of the administrative dataset. Lastly, the datasets did not include any information on tumour size, which is of critical importance when considering survival.

## 3.6. Conclusions

In summary, this descriptive study identified HCC and cholangiocarcinoma as the main types of PLC in Tasmania. Whilst we observed an increasing survival trend, survival time remains poor. This emphasises the importance of prevention and early detection for people at high risk of developing PLC. Understanding of the most frequent risk factors for PLC in Tasmania can be used to support targeted screening to increase the rate of early diagnosis.

# Chapter 4. Accuracy of coded cause of death data: A case study based on primary liver cancer

Chapter 3 provided information on the characteristics and survival time of liver cancer in Tasmania. Whilst undertaking this study, nearly half of all cases were found to have a conflicting cause of death recorded by the TCR compared to the ABS. Cause of death data is critical to estimate cause-specific mortality and survival time. Previously published studies have shown that inaccurate cause of death data can lead to less precise cause-specific survival estimates<sup>103, 105, 106</sup>. In response, Chapter 4 focuses on the level of agreement between the TCR, ABS and independent medical practitioners to determine causes of death and any impact on cause-specific survival.

## 4.1. Abstract

*Background:* Cause of death data from death certificates is important for understanding mortality related to cancers and their treatments. This information is coded and reported by the ABS. In addition to the information from death certificates, the TCR uses additional sources of information including medical records, pathology, and imaging services to report the underlying cause of death.

*Aims:* Using a cohort of PLC cases, the aims were to investigate the level of agreement for cause of death data between the ABS and TCR along with medical practitioners' opinions, secondly, estimate the impact of different coding practices and resulting cause of death data on cause-specific survival.

*Methods:* Causes of death were compared between the ABS and TCR, with discrepancies independently reviewed by specialist medical practitioners. Cohen's Kappa statistics were applied to evaluate the degree of concordance between the ABS, TCR and medical practitioners' opinions regarding cause of death. The cumulative incidence function was used to estimate cause-specific survival time in the presence of a competing risk framework according to sex, place of residence, country of birth, and type of PLC.

*Results:* A minimal level of agreement (Kappa=0.35) was observed when comparing the TCR and ABS cause of death data. Agreement between the TCR and medical practitioners was weak (Kappa=0.51), moderate between the ABS and medical practitioners (Kappa=0.61), strong (Kappa=0.87) between the medical practitioners. These results reflect a greater level of agreement between medical practitioners than between coding agencies. Overall, cause-specific survival time was similar across the TCR, ABS and medical practitioners by sex, place of residence and country of birth, however, regarding type of PLC, a small difference was observed.

*Conclusions:* Agreement between the TCR and ABS cause of death data was minimal. Interestingly, whilst both the ABS and medical practitioners coded cause of death using death certificate data,

agreement was only moderate. Overall, cause-specific survival time was similar across the TCR, ABS and medical practitioners, however, a small difference was observed for the type of PLC. As PLC is a low survival cancer, such results may be different to cancers with better survival such as breast cancer.

Keywords: primary liver cancer, hepatocellular carcinoma, competing risk, noncancer mortality.

## 4.2. Introduction

PLC has become the fourth leading cause of cancer mortality and is the sixth most frequently occurring cancer globally<sup>17</sup>. In Australia, the age-standardised incidence and mortality rates have increased substantially and are projected to continue to increase in the coming decades<sup>2, 20, 89</sup>.

Cancer mortality data play an important role in the estimation of population-based mortality statistics and survival rates<sup>132</sup>. In turn, these data are used in medical research to develop and evaluate health policies and resource allocation decisions. Therefore, this information needs to be as accurate and complete as possible.

In Australia, a death must be registered with the jurisdictional Registry of Births, Deaths and Marriages as soon as possible<sup>132, 133</sup>. The registration is based on the death certificate completed by the attending medical practitioner or a coroner<sup>132, 133</sup>. This certificate includes information on the underlying cause of death and associated cause of death<sup>132, 133</sup>. According to the Australian Institute of Health and Welfare (AIHW), the *underlying cause of death* is defined as "the disease or injury which initiated the train of morbid events leading directly to death"<sup>132</sup>. The *associated causes of death* are all causes that contributed to the death, other than the underlying cause of death <sup>132</sup>.

#### **Coding Cause of Death by the Australian Bureau of Statistics**

The information from death certificates is provided to the ABS on a monthly basis for coding based on the International Statistical Classification of Diseases and Related Health Problems 10<sup>th</sup> Revision (ICD-10)<sup>108</sup>. It enables comparison of mortality statistics over time and between different areas. This process produces the national Cause of Death Unit Record File (COD-URF)<sup>84</sup>, the mortality dataset including all national information on causes of death registered in Australia from the ABS. The information from this dataset is the major source of Australian cancer mortality and survival data published by the AIHW<sup>132</sup>. For this study, cause of death from the COD-URF was referred to as the ABS dataset.

#### Coding Cause of Death by the Tasmanian Cancer Registry

The Registries of Births, Deaths and Marriages also report death certificate data to jurisdictional cancer registries<sup>132</sup>. Each of death is then recorded in the TCR dataset can include: five causes of

death, eight antecedent causes of death, and two other important conditions recorded at the time of death.

TCR staff then access the Digital Medical Record (DMR) of each public hospital patient and review the cause of death data in the death certificate in the context of the patient's DMR. In addition, the TCR receives cancer notifications from pathology, and radiology laboratories. All documentation provided to the TCR is reviewed to further refine the cause of death data in the TCR database.

The TCR uses the International Classification of Diseases for Oncology, Third Edition (ICD-O3)<sup>84</sup> for coding the site (topography) and morphology of all cases to determine the underlying cause of death and associated cause(s) of death. For the purposes of comparing between different geographic areas of Australia, all causes of death data are converted the ICD-10<sup>108</sup>.

A study estimating survival for Tasmanian patients diagnosed with PLC was conducted. Data on cancer notifications, date and cause(s) of death were taken from the TCR and ABS datasets. We compared these data and identified the cause of death differed in 112 cases (48.3%). Whilst this is not an issue when estimating all-cause mortality, it is important when considering cause-specific mortality.

Several studies have identified that inaccuracies of cause of death information obtained from death certificates<sup>103, 134-137</sup>. For example, a study in the US reported that 50.8% of reviewed death certificates had errors regarding the cause of death<sup>134</sup>. To our knowledge, no studies have investigated this issue in Australia.

The primary aim of this study was to investigate the level of agreement on cause of death between the ABS and TCR, along with medical practitioners' opinions. The secondary aim was to understand the impact of different coding practices and resulting cause of death data on cause-specific survival. We used PLC cases as a case study.





## 4.3. Methods

#### **Data sources**

The cohort was defined as all PLC notifications to the TCR between 01/01/2007 and 31/12/2015, aged >=18 years, and deceased. The ABS dataset provides each patient's information on date of birth, sex, date of death, age at death, underlying cause of death and up to 20 associated cause of death. Linkage to the ABS's COD-URF was undertaken by the Tasmanian Data Linkage Unit using probabilistic linkage <sup>107</sup>.

The records of the underlying cause of death in the TCR dataset with the ABS dataset and information from death certificates were compared. For cases in which a discrepancy was identified, an Excel spreadsheet was developed for review independently by medical practitioners. This deidentified spreadsheet included:

1. ABS data (1 underlying cause of death and up to 8 associated causes of death)

2. TCR data (date of birth, sex, age at death, morphology code, topography code, ICD-10 code, underlying cause of death)

3. Information from death certificates (including up to five causes of death, the disease or condition directly leading to death; up to 8 antecedent causes, and other significant conditions, contributing to the death but not related to the disease, injury or condition causing it).

Three specialist medical practitioners were involved in reviewing this information, including two hepatologists/gastroenterologists (referred as medical practitioner 1 and medical practitioner 2) with expertise in PLC. Where discrepancies remained, a third reviewer, a medical oncologist (referred as medical practitioner 3) made the final determination of the cause of death based on the group of information from the ABS, TCR and death certificates. Each practitioner independently reviewed the available data.

#### Statistical analysis

Descriptive statistics were used to describe the characteristics of the cohort. Cohen's Kappa statistics were applied to evaluate the degree of concordance between the ABS, TCR and medical practitioner's judgement regarding causes of death<sup>110</sup>. Cohen's kappa statistics were used to assess the degree of agreement by chance alone between the different methods of ascertaining cause of death. Values range from 0 to 1; 0 represents no agreement and 1 perfect agreement the interpretation of Cohen's Kappa is interpreted as follow <sup>110</sup>.

Value of Kappa	Level of Agreement	% of Data that are reliable
0-0.20	None	0-4%
0.21-0.39	Minimal	4-15%
0.40-0.59	Weak	15-35%
0.60-0.79	Moderate	35-63%
0.80-0.90	Strong	64-81%
Above 0.90	Almost Perfect	82-100%

The cumulative incidence function was used to estimate deaths caused by non-PLC cases based on the final causes of death provided by the medical practitioners in the presence of competing risks <sup>111</sup>. The cumulative incidence function curves and the subdistribution hazard ratios (SHRs) were generated to describe the incidence of death over time from the event of interest (liver cancer) and the competing risk of death (other causes of death)<sup>111, 139</sup>. Estimates for the cumulative incidence of death after notification of PLC were undertaken according to sex; place of residence (urban/rural) according to the Australian Statistical Geography Standard 2016 <sup>118</sup>; type of PLC based on the TCR data according to the ICD-10 codes for HCC with the code C22.0, cholangiocarcinoma with the code C22.1 and unspecified types - C22.9 (these cases belonged to this group as the subtype was not specified). The patients in this group tend to be diagnosed at an advanced stage, when further investigation to identify the type of PLC is often unwarranted as it will have no impact on treatment decisions. Other rare

forms of PLC were excluded due to small numbers ( $\leq$ 5) (included C22.2 hepatoblastoma, C22.3 angiosarcoma of liver, C22.4 other sarcomas of liver, C22.7 other specified carcinomas of liver) <sup>108</sup>. Country of birth was coded according to the Standard Australian Classification of Countries 2016 <sup>119</sup>. Due to the small number of patients being born in overseas countries, this was reported as either Australian or overseas born. Where the underlying cause of death was recorded as PLC, this was considered as the event of interest; other causes of death were considered as competing risks.

Data were analysed using Stata 15 (Ver.15, College Station, Texas, USA). Any cell counts  $\leq$ 5 cases were suppressed to prevent any possibility of identification. A p-value less than 0.05 was deemed statistically significant.

Ethical approval for this study was obtained by the University of Tasmania Human Research Ethics Committee (H0016958).

## 4.4. Results

Between 2007 and 2015, 293 patients were diagnosed with PLC in Tasmania, of which 239 were deceased. Among these patients, seven cases were not matched between the ABS and TCR datasets (i.e., there were six cases marked as deceased in the TCR but not in the ABS and in turn, one case marked deceased in the ABS but not found in the TCR). These patients were excluded in the agreement analysis as the data provided no potential for comparison. Finally, 232 deceased cases with matches in the TCR and ABS datasets were included in the study.

For all included cases, the majority were male (74.6%, n=173), most resided in urban areas (62.9%, n=146) and over two-thirds were born in Australia (68.1%, n=158). Based on the TCR data, the most common cause of death was HCC (46.1%, n=107), followed by cholangiocarcinoma (21.1%, n=49) while the remaining 30.6% (n=71) were reported as having an unspecified type of PLC. The remaining cases were combined into 'other' (C22.2 hepatoblastoma, C22.3 angiosarcoma of liver, C22.4 other sarcomas of liver, C22.7 other specified carcinomas of liver). Table 5 shows the characteristics of all included cases.

С	haracteristics	Number	%
		232	100
Discrepancies in cause o	f death	112	48.3
Sex	Males	173	74.6
	Females	59	25.4
Place of residence	Rural	86	37.1
	Urban	146	62.9
Country of birth	Australian born	158	68.1
	Overseas born	30	12.9
	Not stated/ Missing	44	19.0
Type of PLC	HCC	107	46.1
	Cholangiocarcinoma	49	21.1
	Unspecified type	71	30.6
	Other types	5	2.2

Table 5: Characteristics of deceased cases among liver cancer patients in Tasmania (2007-2015)

#### Agreement between the ABS, TCR and medical practitioners

#### Agreement for all cases

The comparisons regarding the underlying cause of death by the ABS, TCR and medical practitioners were conducted in three stages. Firstly, the 232 ABS and TCR causes of death were reviewed, with discrepancies identified in 48.3% (n=112) of cases: a minimal level of agreement (Kappa=0.35, p<0.001) (Table 6).

The 112 discrepancies were then reviewed independently by medical practitioners 1 and 2. Of these cases, 16 (17.0%) discrepancies remained and were further reviewed by medical practitioner 3 to achieve a consensus. Interrater reliability between the TCR and final consensus from the medical practitioners showed weak agreement (Kappa=0.51, p<0.001) and between the ABS and medical practitioners' moderate agreement (Kappa=0.61, p<0.001).

The highest interrater reliability was observed between medical practitioners 1 and 2 (Kappa=0.87, p<0.001). This reflects a greater consistency between these experts when deciding the underlying cause of death. Although showing similar trends of agreement across the different comparisons, the Kappa statistics revealed more robust outputs than the percent agreement.

 Table 6: The agreement between the ABS, TCR and the medical practitioners regarding underlying cause of death of liver cancer patients in Tasmania (N=232)

Comparison	Agreement frequency (%)	Kappa statistics
ABS-TCR	120 (51.3%)	0.35*
TCR-Medical practitioners	150 (64.7%)	0.51*
ABS-Medical practitioners	168 (72.4%)	0.61*

\* Significant difference with p<0.001

#### Cumulative incidence of cause-specific deaths

The cumulative incidence of death after being diagnosed with PLC is presented according to (1) sex (males/females), (2) place of residence (urban/rural), (3) country of birth (Australia/overseas), and (4) type of PLC (Unspecified type/ HCC/Cholangiocarcinoma) (Table 7). The calculations from the three different sources of information-the TCR, ABS and the medical practitioners-returned many similar SHR results. First, the cumulative incidence of death (with females as the reference group) was trending towards being higher in males compared with females (Figure 11-2A). The SHRs were across the datasets: 0.74 (95%CI: 0.52-1.1) based on the TCR data; 0.77 (95%CI: 0.53-1.12) based on the ABS data; and 0.74 (95%CI: 0.51-1.01) based on the medical practitioners' data.

Second, the cumulative incidence of death with urban as the reference group was largely similar for rural and urban residence, with each of the three datasets providing similar results (Figure 11-2B): the SHRs being 0.90 (95%CI: 0.68-1.20), 0.93 (95%CI: 0.69-1.26), and 0.93 (95%CI: 0.69-1.25), respectively from the TCR, ABS and medical practitioners data.

Interestingly, the different estimations based on the three datasets (i.e., cause of death datasets from the TCR, ABS and consensus from the medical practitioners), consistently showed that the cumulative incidence of death was higher in patients born in Australia than those overseas-born (Figure 11-2C), with the SHRs being 0.68 (95%CI: 0.57-0.82) based on the TCR data, 0.63 (95%CI: 0.52-0.77) based on the ABS data, and 0.69 (95%CI: 0.58-0.83) based on the medical practitioner's data. All these estimates were statistically significant (p<0.001).

Lastly, the type of PLC was the only factor that contributed to inconsistent estimates in the SHRs (Table 7) and survival time across the datasets (Figure 11-2D). The SHRs (reference group: 'Unspecified type') were 0.48 (95% CI: 0.33-0.70) and 0.50 (95% CI: 0.33-0.77) for HCC and Cholangiocarcinoma groups respectively based on the TCR's data. The SHRs based on ABS's data were similar with 0.45 (95% CI: 0.31-0.67) and 0.40 (95% CI: 0.26-0.63) for HCC and Cholangiocarcinoma respectively, and for the medical practitioners' data the SHRs were 0.64 (95% CI: 0.42-0.96) and 0.69 (95% CI: 0.44-1.08) for HCC and Cholangiocarcinoma respectively. The cumulative incidence of death was highest in cases with unspecified PLC in all datasets. This largely reflects the nature of this unspecified group, i.e., diagnosed at an advanced stage with no subsequent clinical investigations conducted to inform treatment. However, while significant differences were found when estimating the SHRs in the ABS and TCR datasets, the estimation based on the medical practitioner's data was not statistically significant (p=0.11) between the patients with Cholangiocarcinoma and HCC.

46

	Variables	Tasmanian Cancer Registry SHR (95%CI)	Australian Bureau of Statistics SHR (95%CI)	Medical Practitioners SHR (95%CI)
Sex	Males (Ref)			
	Females	0.74 (0.52-1.1)	0.77 (0.53-1.12)	0.74 (0.51-1.01)
Place of	Urban (Ref)			
residence	Rural	0.90 (0.68-1.20)	0.93 (0.69-1.26)	0.93 (0.69-1.25)
Country	Australia born (Ref)			
of birth	Overseas born	0.68 (0.57-0.82) ***	0.63 (0.52-0.77) ***	0.69 (0.58-0.83) ***
Type of	Unspecified type (Ref)			
liver	HCC	0.48 (0.33-0.70)***	0.45 (0.31-0.67)***	0.64 (0.42-0.96)*
cancer	Cholangiocarcinoma	0.50 (0.33-0.77)**	0.40 (0.26-0.63)***	0.69 (0.44-1.08)

 Table 7: Hazard Ratios by sex, place of residence, country of birth, and type of liver cancer

 from the ABS, TCR, Medical Practitioners' data

Note: \*\*\* p<0.001; \*\* p<0.01; \* p<0.05

**Australia Bureau of Statistics** 

**Medical Practitioners** 



Figure 11: Cumulative incidence function curves illustrate deaths from liver cancer patients and other reasons by gender (2-A), place of residence (2-B), country of birth (2-C), and type of liver cancer (2-D) from ABS, TCR, Medical Practitioners' data

#### 4.5. Discussion

Cause of death data derived from death certificates is of critical importance to fully understand mortality related to cancers and their treatments. This study describes both the processes of recording the cause of death and the impact on cause-specific survival using a competing risk framework for a cohort of PLC patients.

Our results showed the overall concordance between the ABS and TCR was minimal (Kappa=0.35, with discrepancies present for nearly half of the cases). When cause of death was based on death certificates only, the results between the ABS and specialists showed moderate agreement (Kappa=0.61). The discrepancies identified between the TCR and ABS were largely due to the additional sources of information used by the TCR, that is, the DMR, pathology and imaging services, to conclude the underlying cause of death.

To the best of our knowledge, no previous studies have been published on the accuracy of cause of death data and the impact on survival estimates for PLC. A study using a similar approach to ours but looking at breast cancer cases, compared cause-specific survival time from death certificates with coded cause of death from the Geneva Cancer Registry. This Registry codes cause of death based on death certificate data along with all available clinical information. The authors reported a high level of agreement (Kappa = 0.82), with 8.8% of cases misclassified. Overall cause-specific survival time was not affected, but differences were observed for some groups such as patients aged over 80 years<sup>103</sup>.

In our study, there were 71 cases diagnosed with a morphology code of 8003 (Neoplasm, malignant) based on the ICD-O3<sup>84</sup>. In the absence of histology data to accurately describe the morphology of a tumour, the TCR coded cases as C22.\*, with the 8003 morphology code recorded as unspecified liver cancer (C22.9). For cases coded as unspecified liver cancer (54 cases/71 cases), a high rate of discrepancies between the TCR and ABS regarding the cause of death was observed. The coding methods of these patients partly contributed to the minimal agreement between the ABS and TCR. A Victorian study <sup>117</sup> highlighted the issue of PLC cases being coded as 'unspecified liver cancer' due to missing morphology code. This study reported that following a review of the medical records, 75.9% of these 'unspecified liver cancer' cases were subsequently recoded as HCC. However, we did not have access to the medical records from public hospitals in Tasmania or results from pathology and imaging laboratories in this study. Thus, we could not draw strong conclusions regarding the accuracy of coded causes of death data from TCR.

Our results using the competing risks framework did not identify sex or rurality to have an impact on the cumulative incidence of death. We did, however, observe that the cumulative incidence of death was highest in cases with unspecified liver cancer. This is in-line with clinical practice that patients diagnosed at an advanced stage of PLC often do not undergo further investigations, and therefore histology data are not available.

In contrast to our results, a US study reported cumulative incidence of breast cancer deaths was overestimated due to misclassification of the cause of death<sup>104</sup>. Comparisons of mortality between breast and PLC using a competing risk framework require careful consideration. Breast cancer is associated with substantially longer survival in both the US and Australia (5 year survival 91.3%<sup>140</sup> and 90.8%<sup>4</sup> respectively) compared to PLC (5 year survival 19.2%<sup>140</sup> and 18.5%<sup>2</sup> respectively). Additionally, breast cancer patients are generally younger than PLC patients, with incidence highest for breast cancer patients between the ages 70-74<sup>141</sup> and for PLC patients between the ages 80-84<sup>89</sup>.

In turn, breast cancer cases have a higher risk of death from other causes of death in comparison to PLC cases. In our study, the majority of PLC cases died from the event of interest (liver cancer) in all three datasets (89.2% for the TCR, 78.0% for the ABS, and 74.6% based on the consensus from medical practitioners). Meanwhile, in the breast cancer study, the number of deaths due to the event of interest (breast cancer) represented a small proportion of all deaths (12.2%)<sup>104</sup>. These different results suggest that the cumulative incidence function may not show substantial impacts on cause-specific survival for cancers with short survival time such as PLC.

#### **Strengths and Limitations**

The use of linked health data is useful as a means of studying many health conditions, outcomes and service provision in a cost-effective manner<sup>142</sup>. In our study, the TCR provided access to high quality data, based on using probabilistic linkage methods to reduce potential errors<sup>107</sup>. The cumulative incidence function was used to estimate deaths caused by noncancer events in the presence of competing risks. This method is preferred to the Kaplan-Meier or net survival methods, which usually overestimates the absolute risk of cause-specific survival and ignores competing causes<sup>111, 139, 143</sup>. The competing risk methods provided the appropriate framework to analyse the interplay between deaths from PLC and other competing risk factors based on different coding practices<sup>139, 143, 144</sup>.

A limitation of this study was that we could not draw strong conclusions regarding the accuracy of cause of death data from the TCR or ABS. Access to patients' digital medical records would be required for this. Future research should focus on better understanding of coding practices and the accuracy of the different approaches. Cause-specific survival estimates have been shown to provide lower survival rates compared to all cause estimates for breast cancer<sup>145</sup>. In this context, robust cause of death data will support more accurate cause-specific survival estimates for patients and clinicians. Whilst this may not be as relevant for low survival cancers such as liver PLC, this may be particularly useful for cancers with longer survival and earlier age at diagnosis such as breast cancer.

## 4.6. Conclusions

This is the first study that has evaluated the impact of different coded causes of death on estimates of cause-specific survival for PLC cases. Agreement between the TCR and ABS cause of death data was minimal, weak between the TCR and medical practitioners, and was only moderate between the ABS and medical practitioners. The cumulative incidence of deaths were similar across the TCR, ABS, and medical practitioners. However, it was different according to the type of PLC, reflecting a need for more cohesive reporting and coding practices that will ultimately improve funding allocation. Lastly, whilst there were no differences observed for cause-specific survival across these datasets, a small difference was observed regarding type of PLC. As PLC is a low survival cancer, such results may be different to cancers with better survival such as breast cancer.

## **Chapter 5. Summary and future directions**

## 5.1. Preface

This thesis presents the demographic characteristics, risk factors and survival time for all PLC cases diagnosed in Tasmania between 2007 and 2015. Secondly, the results of a study investigating the level of agreement for causes of death between the ABS, TCR and medical specialists are reported. The impact on cause-specific survival is presented and discussed. This chapter provides a synopsis, conclusions, and suggestions for future directions.

## 5.2. Summary of the thesis

Chapter 1 provides an overview of the epidemiology and clinical perspectives of PLC. More specifically, the global epidemiology of PLC is presented, and the methods of diagnosis, staging, and surveillance are discussed. This chapter then focuses on the epidemiology of PLC in Australia, the rationale for conducting PLC research in Tasmania, and the aims of this thesis are presented.

Chapter 2 describes the methods used in the studies reported in Chapters 3 and 4. Much of this Master's project has used linked administrative data: the linkage methodologies are described in this chapter, along with the study design, selection of 'participants', cancer and death coding practices in the TCR, and the analysis methods used in both studies. This chapter summarises the main methods used in these studies, whilst more specific information about the methods of each study are included in Chapters 3 and 4.

Chapter 3 (Study 1) is a retrospective data linkage study using de-identified datasets from the TCR and inpatient admissions from Tasmanian public hospitals. This study describes the demographic characteristics, risk factors and relative survival time for subtypes of PLC in Tasmania between 2007 and 2015. Over the study period, there were 293 primary PLC cases identified. HCC (51.9%) and cholangiocarcinoma (20.5%) were the main types reported. Three-quarters of all PLC cases were male, and the average age was 70 years. For cases with public hospital admissions, 43% did not have a risk factor for PLC identified. Of those who did, the most common were cirrhosis (37%), chronic viral hepatitis (35%), diabetes (27%), and alcohol-related liver disease (23%). The mean age at diagnosis for all cases was 69.6 years, with a median survival time of 6.2 months. The 1-,3- and 5- year relative survival rates were 38.3%, 12.8%, and 6.7%. Understanding the most frequent risk factors for PLC in Tasmania can be used to support targeted surveillance to increase the rate of early diagnosis.

Whilst undertaking the study reported in Chapter 3, it was identified that nearly half of all cases had different causes of death recorded by the TCR compared with the ABS. Chapter 4 (Study 2) reports

53

on a study developed to investigate this issue. The aim was to identify consensus for causes of death, and to understand how coding impacts causespecific survival estimates. For the 48.3% (112/232) of cases in which a discrepancy was identified, independent reviews were conducted by medical specialists with expertise in PLC. Concordance of cause of death data was estimated using Kappa statistics, and the impact on cause-specific survival time was explored using a competing risk framework. The overall concordance regarding cause of death data was minimal between the ABS and TCR (Kappa=0.35). Moderate agreement (Kappa=0.61) between the ABS and medical specialists was identified.

Whilst the cumulative incidence of cause-specific survival estimates were similar across the TCR, ABS, and medical practitioners, when assessing this in the context of region of birth, the SHRs were higher for cases born in Australia compared to those born overseas. This result is in-keeping with other Australian studies<sup>116, 121</sup>, and is related to differing aetiologies. For example, Australian-born PLC patients are more likely to present with cirrhosis due to alcoholic liver disease; in contrast, Asian born patients are more likely to present with PLC related to chronic hepatitis B with or without cirrhosis. In turn, the latter patient group are more likely to be eligible for treatments such as liver resection<sup>97</sup>. A recent Australian consensus statement has supported surveillance for high risk patients, including those with cirrhosis, to increase the rate of early diagnosis<sup>95</sup>. Whilst this would reduce the difference in survival across these groups, further work is required to understand the acceptability and uptake of surveillance from the perspectives of patients and health care providers.

In addition, differences were observed when estimating SHRs for the type of PLC. Using "Unspecified" cases as the reference group, statistically significant differences were observed for HCC and cholangiocarcinoma based on the TCR and ABS data. No significant difference was observed for the medical practitioners' data. In order to understand how different coding practices may over or underestimate cause-specific survival, the accuracy of cause of death data would need to be assessed. This could be done using patients' digital medical records to better understand the complexities of each individual's health. In addition, these results highlight the need for more cohesive reporting and coding practices that will ultimately improve understanding of survival of cancers.

## **5.3.** Integrated conclusions of the thesis

Survival of PLC remains poor in Tasmania, with a 5 year survival rate of just 6.7% over 2007 to 2015. This is due to the high number of cases that were diagnosed at late stages (almost 10% of cases at time of death, 24.9% at an advanced stage).

In contrast is the Japanese experience, with a 5 year survival rate of 40.4% for HCC patients diagnosed between 1998 and 2009<sup>146</sup>. This improved survival is largely related to the stage of the

tumours at diagnosis, with more than 60% diagnosed at early stages<sup>146, 147</sup>. This has occurred in the context of a nationwide PLC surveillance program that was developed in the 1980s for high risk patients. This program includes ultrasound and measurement of three tumour markers included AFP, the lens culinaris agglutinin-reactive glycoform, and des-gamma-carboxy prothrombin, every 3-4 months for patients with CHB and CHC-associated cirrhosis, and every 6 months for patients with nonviral cirrhosis, and patients with CHB and/or CHC (without cirrhosis)<sup>146-148</sup>.

Many international liver societies recommend surveillance of HCC for high risk populations. For example, the American Association for the Study of Liver Diseases (AASLD) and the European Association for the Study of the Liver (EASL) recommend surveillance every 6 months, consisting of ultrasound  $\pm$  AFP for high risk patients (cirrhosis patients and subgroups of CHB patients)<sup>74, 94</sup>.

In the Australian context, a consensus statement was published in late 2020 by PLC specialists with evidence-based recommendations<sup>95</sup>. This paper states it is now a priority to institute HCC surveillance strategies in Australia for high-risk patients, with the aim of identifying tumours at early stages. The authors recommend surveillance should occur 6-monthly using liver ultrasound and  $\pm$  AFP in<sup>95</sup>:

People with cirrhosis (any aetiology)

Non-cirrhosis patients with CHB infection in:

- Asian men older than 40 years
- Asian women older than 50 years
- People born in sub-Saharan Africa older than 20 years
- Aboriginal and Torres Strait Islander people older than 50 years

A small number of Australian studies have demonstrated the effectiveness of HCC surveillance every 6 months for high risk patients to increase early detection and improve survival<sup>97, 149</sup>. Our results support the need for surveillance to increase the rate of early detection of HCC in high risk patients. In turn, this will improve the very poor survival outcomes for people diagnosed with HCC in Australia.

## 5.4. Future directions

#### Identification of barriers and enablers for liver cancer surveillance

Despite Australian recommendations since 2005<sup>150</sup> to screen high risk patients, uptake and adherence remain poor. A recent population-based study of patients diagnosed with HCC in Victoria, Australia reported that 40% were participating in surveillance at the time of diagnosis<sup>97</sup>. For patients living with CHB, only 27% adhered to surveillance on a six monthly basis<sup>151</sup>. However, the reasons for such low participation are not well understood in Australia. Internationally, several studies have reported on
multiple barriers to surveillance from the perspective of primary care providers, including failure to identify the clinical need for surveillance for high risk patients, insufficient knowledge of risk factors, indications for surveillance and time intervals, along with competing clinical concerns<sup>152, 153</sup>. Other studies have reported on barriers from patients' perspectives, such as knowledge deficits regarding their risk, difficulty with scheduling processes of tests and appointments, costs of liver imaging and transportation barriers<sup>154</sup>. In addition health care system barriers such as a lack of guidelines for surveillance have also been reported to be a barrier<sup>153, 155</sup>. As these studies were conducted in countries with health systems that differ greatly to Australia's, it is important to understand the local barriers and enablers impacting patients, health care providers, and health care system.

The findings of the current study suggest that better understanding of the barriers and enablers of regular participation in surveillance for high-risk patients is required. This will support the development of a more acceptable and effective surveillance strategy.

### Determination of pathways to diagnosis of liver cancer and associations with survival

As the thesis findings provide a large descriptive picture of the PLC in Tasmania, more challenging research directions are also opened. PLC is frequently diagnosed at a late or end stage, resulting in very poor outcomes for patients. Characterisation of the pathways to diagnosis, time between first presentation to a medical professional and clinical diagnoses, reasons for emergency department presentations and inpatient admissions before the diagnosis of PLC should be understood. As no such study has been published in Australia, research on these pathways to diagnosis will be the first step to understanding why most PLC patients are diagnosed at late stages. In turn, this information can be used to inform the development of a surveillance strategy.

# Develop a strategy/program to support increased participation in regular surveillance for those at risk and liver cancer patients

The effectiveness of HCC surveillance programs is largely accepted and has been recommended for high risk patients by multiple international professional bodies and in guidelines<sup>3, 23, 73, 74, 156</sup>. However, their effectiveness in clinical settings is hampered by multiple barriers<sup>152-155</sup>. Additionally, there have been three well established national screening programs for breast, bowel, and cervical cancers in Australia<sup>157-159</sup>. Whilst HCC screening will need to be targeted to high-risk individuals, valuable lessons from these programs should be considered. Based on the results from suggestions above, it is now of critical importance to work with patients and clinicians to develop appropriate surveillance programs to increase the rate of early diagnosis of PLC and eventually improve patient survival.

### Assess the cost effectiveness of HCC surveillance

As noted, effectiveness of HCC surveillance for high risk patients has been demonstrated in a number of studies. More specifically, it has been shown to increase the rate of early diagnosis, and ultimately improve survival in Australia<sup>97, 149</sup>. However, no evidence of the cost effectiveness of HCC surveillance has been published for Australia. As surveillance programs are costly to government(s), it is essential to provide such information that can be used in decision making. Therefore, a thorough cost effectiveness evaluation of clinically effective surveillance programs will be required before implementation can occur, to ensure it is financially justified at the population level.

### References

1. Goodman ZD. Neoplasms of the liver. *Modern Pathology*. 2007; **20**(1):S49-S60.

2. Australian Institute of Health and Welfare. *Colorectal and other digestive tract cancers*. Canberra: AIHW; 2018.

3. European Association For The Study Of The Liver. EASL clinical practice guidelines: management of hepatocellular carcinoma. *Journal of hepatology*. 2018; **69**(1):182-236.

4. Australian Institute of Health and Welfare. *Cancer in Australia 2019*. Canberra: Australian Institute of Health and Welfare; 2019.

5. Bridgewater JA, Goodman KA, Kalyan A, Mulcahy MF. Biliary tract cancer: epidemiology, radiotherapy, and molecular profiling. *American Society of Clinical Oncology Educational Book*. 2016; **36**:e194-e203.

6. Cancer Council Australia. *Understanding Cancer in the Liver*2020.

7. Kirstein MM, Vogel A. Epidemiology and risk factors of cholangiocarcinoma. *Visceral medicine*. 2016; **32**(6):395-400.

8. Banales JM, Cardinale V, Carpino G, Marzioni M, Andersen JB, Invernizzi P, et al. Expert consensus document: Cholangiocarcinoma: current knowledge and future perspectives consensus statement from the European Network for the Study of Cholangiocarcinoma (ENS-CCA). *Nature reviews Gastroenterology & hepatology*. 2016; **13**(5):261-80.

9. Sripa B, Pairojkul C. Cholangiocarcinoma: lessons from Thailand. *Current opinion in gastroenterology*. 2008; **24**(3):349.

10. Shin H-R, Oh J-K, Masuyer E, Curado M-P, Bouvard V, Fang Y, et al. Comparison of incidence of intrahepatic and extrahepatic cholangiocarcinoma–focus on East and South-Eastern Asia. *Asian Pac J Cancer Prev.* 2010; **11**(5):1159-66.

11. Chaudhary P, Bhadana U, Singh R, Ahuja A. Primary hepatic angiosarcoma. *European Journal of Surgical Oncology (EJSO)*. 2015; **41**(9):1137-43.

12. Millan M, Delgado A, Caicedo LA, Arrunategui AM, Meneses CA, Villegas JI, et al. Liver Angiosarcoma: Rare tumour associated with a poor prognosis, literature review and case report. *International journal of surgery case reports*. 2016; **28**:165-8.

13. Sharma D, Subbarao G, Saxena R, editors. Hepatoblastoma. Seminars in diagnostic pathology; 2017: Elsevier.

14. Hiyama E. Pediatric hepatoblastoma: diagnosis and treatment. *Translational pediatrics*. 2014; **3**(4):293.

15. Torbenson M. Fibrolamellar carcinoma: 2012 update. *Scientifica*. 2012; **2012**.

16. Lafaro KJ, Pawlik TM. Fibrolamellar hepatocellular carcinoma: current clinical perspectives. *Journal of hepatocellular carcinoma*. 2015; **2**:151.

17. Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. *CA: A Cancer Journal for Clinicians*. 2018; **68**(6):394-424.

18. Valery PC, Laversanne M, Clark PJ, Petrick JL, McGlynn KA, Bray F. Projections of primary liver cancer to 2030 in 30 countries worldwide. *Hepatology*. 2018; **67**(2):600-11.

19. Altekruse SF, McGlynn KA, Reichman ME. Hepatocellular carcinoma incidence, mortality, and survival trends in the United States from 1975 to 2005. *Journal of clinical oncology*. 2009; **27**(9):1485.

20. Cocker F, Chien Yee K, Palmer AJ, de Graaff B. Increasing incidence and mortality related to liver cancer in Australia: time to turn the tide. *Aust N Z J Public Health*. 2019; **43**(3):267-73.

21. El-Serag HB. Epidemiology of viral hepatitis and hepatocellular carcinoma. *Gastroenterology*. 2012; **142**(6):1264-73.

22. Chen J-G, Egner PA, Ng D, Jacobson LP, Muñoz A, Zhu Y-R, et al. Reduced aflatoxin exposure presages decline in liver cancer mortality in an endemic region of China. *Cancer prevention research*. 2013; **6**(10):1038-45.

23. Singal AG, Pillai A, Tiro J. Early detection, curative treatment, and survival rates for hepatocellular carcinoma surveillance in patients with cirrhosis: a meta-analysis. *PLoS Med*. 2014; **11**(4):e1001624.

24. Bannon F, Di Carlo V, Harewood R, Engholm G, Ferretti S, Johnson CJ, et al. Survival trends for primary liver cancer, 1995–2009: analysis of individual data for 578,740 patients from 187 population-based registries in 36 countries (CONCORD-2). *Annals of Cancer Epidemiology, 2019, vol 3, núm 6*. 2019.

25. Parkin DM. The evolution of the population-based cancer registry. *Nature Reviews Cancer*. 2006; **6**(8):603-12.

26. Zhu RX, Seto WK, Lai CL, Yuen MF. Epidemiology of Hepatocellular Carcinoma in the Asia-Pacific Region. *Gut Liver*. 2016; **10**(3):332-9.

27. Galle PR, Foerster F, Kudo M, Chan SL, Llovet JM, Qin S, et al. Biology and significance of alpha-fetoprotein in hepatocellular carcinoma. *Liver International*. 2019; **39**(12):2214-29.

28. Shimakawa Y, Lemoine M, Bottomley C, Njai HF, Ndow G, Jatta A, et al. Birth order and risk of hepatocellular carcinoma in chronic carriers of hepatitis B virus: a case–control study in The Gambia. *Liver international*. 2015; **35**(10):2318-26.

29. Islami F, Miller KD, Siegel RL, Fedewa SA, Ward EM, Jemal A. Disparities in liver cancer occurrence in the United States by race/ethnicity and state. *CA: a cancer journal for clinicians*. 2017; **67**(4):273-89.

30. Clark PJ, Stuart KA, Leggett BA, Crawford DH, Boyd P, Fawcett J, et al. Remoteness, race and social disadvantage: disparities in hepatocellular carcinoma incidence and survival in Queensland, Australia. *Liver International*. 2015; **35**(12):2584-94.

31. Nordenstedt H, White DL, El-Serag HB. The changing pattern of epidemiology in hepatocellular carcinoma. *Digestive and Liver Disease*. 2010; **42**:S206-S14.

32. Schuppan D, Afdhal NH. Liver cirrhosis. *The Lancet*. 2008; **371**(9615):838-51.

33. Yang JD, Kim WR, Coelho R, Mettler TA, Benson JT, Sanderson SO, et al. Cirrhosis is present in most patients with hepatitis B and hepatocellular carcinoma. *Clin Gastroenterol Hepatol*. 2011; **9**(1):64-70.

34. Lopez PM, Villanueva A, Llovet JM. Systematic review: evidence-based management of hepatocellular carcinoma--an updated analysis of randomized controlled trials. *Aliment Pharmacol Ther*. 2006; **23**(11):1535-47.

35. Llovet JM, Burroughs A, Bruix J. Hepatocellular carcinoma. *Nature Reviews Disease Primers*. 2016; **16018**:1907-17.

36. Llovet JM, Brú C, Bruix J, editors. Prognosis of hepatocellular carcinoma: the BCLC staging classification. Seminars in liver disease; 1999: Thieme Medical Publishers.

37. World Health Organization. Global Hepatitis Report 2017. 2017.

38. Szabó E, Páska C, Novák PK, Schaff Z, Kiss A. Similarities and differences in hepatitis B and C virus induced hepatocarcinogenesis. *Pathology & Oncology Research*. 2004; **10**(1):5.

39. Beasley RP, Lin C-C, Hwang L-Y, Chien C-S. Hepatocellular carcinoma and hepatitis B virus: a prospective study of 22 707 men in Taiwan. *The Lancet*. 1981; **318**(8256):1129-33.

40. Gurtsevitch V. Human oncogenic viruses: hepatitis B and hepatitis C viruses and their role in hepatocarcinogenesis. *Biochemistry (Moscow)*. 2008; **73**(5):504-13.

41. Singal AG, El-Serag HB. Hepatocellular Carcinoma From Epidemiology to Prevention: Translating Knowledge into Practice. *Clin Gastroenterol Hepatol*. 2015; **13**(12):2140-51.

42. Freeman AJ, Dore GJ, Law MG, Thorpe M, Von Overbeck J, Lloyd AR, et al. Estimating progression to cirrhosis in chronic hepatitis C virus infection. *Hepatology*. 2001; **34**(4):809-16.

43. Nelson P, Mathers B, Cowie B, Hagan H, Des Jarlais D, Horyniak D, et al. The epidemiology of viral hepatitis among people who inject drugs: results of global systematic reviews. *Lancet*. 2011; **378**(9791):571.

44. Rinaldi L, Nevola R, Franci G, Perrella A, Corvino G, Marrone A, et al. Risk of Hepatocellular Carcinoma after HCV Clearance by Direct-Acting Antivirals Treatment Predictive Factors and Role of Epigenetics. *Cancers*. 2020; **12**(6):1351.

45. O'Shea R, Dasarathy S, McCullough A. Alcoholic Liver Disease. AASLD Practice Guidelines. Hepatol. 2010; 51 (1): 307-328. 2010.

46. International Agency for Research on Cancer. *World Cancer Report*. 2014.

47. Masarone M, Rosato V, Dallio M, Abenavoli L, Federico A, Loguercio C, et al. Epidemiology and natural history of alcoholic liver disease. *Reviews on recent clinical trials*. 2016; **11**(3):167-74.

48. Basra S, Anand BS. Definition, epidemiology and magnitude of alcoholic hepatitis. *World journal of hepatology*. 2011; **3**(5):108.

49. Matsushita H, Takaki A. Alcohol and hepatocellular carcinoma. *BMJ open gastroenterology*. 2019; **6**(1):e000260.

50. Wong VWS, Chan WK, Chitturi S, Chawla Y, Dan YY, Duseja A, et al. Asia–Pacific Working Party on Non-alcoholic Fatty Liver Disease guidelines 2017—part 1: definition, risk factors and assessment. *Journal of gastroenterology and hepatology*. 2018; **33**(1):70-85.

51. Toplak H, Stauber R, Sourij H. EASL-EASD-EASO Clinical Practice Guidelines for the management of non-alcoholic fatty liver disease: guidelines, clinical reality and health economic aspects. *Diabetologia*. 2016; **59**(6):1148-9.

52. Global Burden of Disease Cancer C, Fitzmaurice C, Dicker D, Pain A, Hamavid H, Moradi-Lakeh M, et al. The Global Burden of Cancer 2013. *JAMA Oncol*. 2015; **1**(4):505-27.

53. Younossi Z, Anstee QM, Marietti M, Hardy T, Henry L, Eslam M, et al. Global burden of NAFLD and NASH: trends, predictions, risk factors and prevention. *Nature reviews Gastroenterology* & hepatology. 2018; **15**(1):11.

54. World Health Organization. *Food Safety Digest*. 2018.

55. McGlynn KA, Petrick JL, London WT. Global epidemiology of hepatocellular carcinoma: an emphasis on demographic and regional variability. *Clinics in liver disease*. 2015; **19**(2):223-38.

56. Hamid AS, Tesfamariam IG, Zhang Y, Zhang ZG. Aflatoxin B1-induced hepatocellular carcinoma in developing countries: Geographical distribution, mechanism of action and prevention. *Oncology letters*. 2013; **5**(4):1087-92.

57. Liu Y, Wu F. Global burden of aflatoxin-induced hepatocellular carcinoma: a risk assessment. *Environmental health perspectives*. 2010; **118**(6):818-24.

58. Department of Agriculture WatE. Food testing - Information for food importers 2020 [cited 2020. Available from: <u>https://www.agriculture.gov.au/import/goods/food/inspection-</u> <u>compliance/information-importers</u>.

59. World Health Organization. Obesity and overweight World Health Organization2020 [Available from: <u>https://www.who.int/news-room/fact-sheets/detail/obesity-and-overweight</u>.

60. Saitta C, Pollicino T, Raimondo G. Obesity and liver cancer. *Annals of Hepatology*. 2019; **18**(6):810-5.

61. El-Serag HB, Kanwal F. Obesity and hepatocellular carcinoma: hype and reality. *Hepatology* (*Baltimore, Md*). 2014; **60**(3):779.

62. Gupta A, Das A, Majumder K, Arora N, Mayo HG, Singh PP, et al. Obesity is independently associated with increased risk of hepatocellular cancer-related mortality: A systematic review and meta-analysis. *American journal of clinical oncology*. 2018; **41**(9):874.

63. Bruix J, Reig M, Sherman M. Evidence-Based Diagnosis, Staging, and Treatment of Patients With Hepatocellular Carcinoma. *Gastroenterology*. 2016; **150**(4):835-53.

64. Estes C, Razavi H, Loomba R, Younossi Z, Sanyal AJ. Modeling the epidemic of nonalcoholic fatty liver disease demonstrates an exponential increase in burden of disease. *Hepatology*. 2018; **67**(1):123-33.

65. Li X, Wang X, Gao P. Diabetes mellitus and risk of hepatocellular carcinoma. *BioMed research international*. 2017; **2017**.

66. Sherman M, Peltekian KM, Lee C. Screening for hepatocellular carcinoma in chronic carriers of hepatitis B virus: incidence and prevalence of hepatocellular carcinoma in a North American urban population. *Hepatology*. 1995; **22**(2):432-8.

67. Görög D, Regöly-Mérei J, Paku S, Kopper L, Nagy P. Alpha-fetoprotein expression is a potential prognostic marker in hepatocellular carcinoma. *World journal of gastroenterology: WJG*. 2005; **11**(32):5015.

68. Chen THH, Chen CJ, Yen MF, Lu SN, Sun CA, Huang GT, et al. Ultrasound screening and risk factors for death from hepatocellular carcinoma in a high risk group in Taiwan. *International journal of cancer*. 2002; **98**(2):257-61.

69. Bottelli R, Tibballs J, Hochhauser D, Watkinson A, Dick R, Burroughs A. Ultrasound screening for hepatocellular carcinoma (HCC) in cirrhosis: the evidence for an established clinical practice. *Clinical radiology*. 1998; **53**(10):713-6.

70. Hanna RF, Miloushev VZ, Tang A, Finklestone LA, Brejt SZ, Sandhu RS, et al. Comparative 13year meta-analysis of the sensitivity and positive predictive value of ultrasound, CT, and MRI for detecting hepatocellular carcinoma. *Abdominal Radiology*. 2016; **41**(1):71-90.

71. Yoon JH, Park J-W, Lee JM. Noninvasive diagnosis of hepatocellular carcinoma: elaboration on Korean Liver Cancer Study Group-National Cancer Center Korea practice guidelines compared with other guidelines and remaining issues. *Korean journal of radiology*. 2016; **17**(1):7-24.

72. Pocha C, Dieperink E, McMaken K, Knott A, Thuras P, Ho S. Surveillance for hepatocellular cancer with ultrasonography vs. computed tomography–a randomised study. *Alimentary pharmacology & therapeutics*. 2013; **38**(3):303-12.

73. Omata M, Cheng A-L, Kokudo N, Kudo M, Lee JM, Jia J, et al. Asia–Pacific clinical practice guidelines on the management of hepatocellular carcinoma: a 2017 update. *Hepatology international*. 2017; **11**(4):317-70.

74. Heimbach JK, Kulik LM, Finn RS, Sirlin CB, Abecassis MM, Roberts LR, et al. AASLD guidelines for the treatment of hepatocellular carcinoma. *Hepatology*. 2018; **67**(1):358-80.

75. Elsayes KM, Hooker JC, Agrons MM, Kielar AZ, Tang A, Fowler KJ, et al. 2017 version of Ll-RADS for CT and MR imaging: an update. *Radiographics*. 2017; **37**(7):1994-2017.

76. Bruix J, Sherman M. Management of hepatocellular carcinoma: an update. *Hepatology* (*Baltimore, Md*). 2011; **53**(3):1020.

77. Marrero JA, Kulik LM, Sirlin CB, Zhu AX, Finn RS, Abecassis MM, et al. Diagnosis, Staging, and Management of Hepatocellular Carcinoma: 2018 Practice Guidance by the American Association for the Study of Liver Diseases. *Hepatology*. 2018; **68**(2):723-50.

78. Di Martino M, De Filippis G, De Santis A, Geiger D, Del Monte M, Lombardo CV, et al. Hepatocellular carcinoma in cirrhotic patients: prospective comparison of US, CT and MR imaging. *European radiology*. 2013; **23**(4):887-96.

79. Van der Pol CB, Lim CS, Sirlin CB, McGrath TA, Salameh J-P, Bashir MR, et al. Accuracy of the liver imaging reporting and data system in computed tomography and magnetic resonance image analysis of hepatocellular carcinoma or overall malignancy—a systematic review. *Gastroenterology*. 2019; **156**(4):976-86.

80. Tapper EB, Lok AS-F. Use of liver imaging and biopsy in clinical practice. *New England Journal of Medicine*. 2017; **377**(8):756-68.

81. Russo FP, Imondi A, Lynch EN, Farinati F. When and how should we perform a biopsy for HCC in patients with liver cirrhosis in 2018? A review. *Digestive and liver disease*. 2018; **50**(7):640-6.

82. Di Tommaso L, Spadaccini M, Donadon M, Personeni N, Elamin A, Aghemo A, et al. Role of liver biopsy in hepatocellular carcinoma. *World journal of gastroenterology*. 2019; **25**(40):6041.

83. Silva MA, Hegab B, Hyde C, Guo B, Buckels JA, Mirza DF. Needle track seeding following biopsy of liver lesions in the diagnosis of hepatocellular cancer: a systematic review and metaanalysis. *Gut.* 2008; **57**(11):1592-6.

84. World Health Organization. *International Classification of Diseases for Oncology* Edition T, editor2013.

85. Tellapuri S, Sutphin PD, Beg MS, Singal AG, Kalva SP. Staging systems of hepatocellular carcinoma: A review. *Indian Journal of Gastroenterology*. 2018; **37**(6):481-91.

86. Falkson G, MacIntyre JM, Moertel CG, Johnson LA, Scherman R. Primary liver cancer. An eastern cooperative oncology group trial. *Cancer*. 1984; **54**(6):970-7.

87. Oken MM, Creech RH, Tormey DC, Horton J, Davis TE, McFadden ET, et al. Toxicity and response criteria of the Eastern Cooperative Oncology Group. *American journal of clinical oncology*. 1982; **5**(6):649-56.

88. Wilson JMG, Jungner G, World Health Organization. Principles and practice of screening for disease. 1968.

89. Australian Institute of Health and Welfare (AIHW) 2018 Cancer Data in Australia. *Australia Cancer Incidence and Mortality (ACIM) books: liver cancer Canberra: AIHW.* 2018.

90. Arii S, Yamaoka Y, Futagawa S, Inoue K, Kobayashi K, Kojiro M, et al. Results of surgical and nonsurgical treatment for small-sized hepatocellular carcinomas: a retrospective and nationwide survey in Japan. The Liver Cancer Study Group of Japan. *Hepatology*. 2000; **32**(6):1224-9.

91. Ghouri YA, Mian I, Rowe JH. Review of hepatocellular carcinoma: Epidemiology, etiology, and carcinogenesis. *J Carcinog*. 2017; **16**:1.

92. Danta M, Barnes E, Dusheiko G. The surveillance and diagnosis of hepatocellular carcinoma. *European journal of gastroenterology & hepatology*. 2005; **17**(5):491-6.

93. Nguyen ALT, Nguyen HTT, Yee KC, Palmer AJ, Blizzard CL, de Graaff B. A Systematic Review and Narrative Synthesis of Health Economic Evaluations of Hepatocellular Carcinoma Screening Strategies. *Value in Health*. 2021.

94. Liver EAFTSOT. EASL clinical practice guidelines: management of hepatocellular carcinoma. *Journal of hepatology*. 2018; **69**(1):182-236.

95. Lubel JS, Roberts SK, Strasser SI, Thompson AJ, Philip J, Goodwin M, et al. Australian recommendations for the management of hepatocellular carcinoma: a consensus statement. *Medical Journal of Australia*. 2020.

96. System ANNDS. *HIV, viral hepatitis and sexually transmissible infections in Australia: Annual surveillance report 2018.* 2018.

97. Hong TP, Gow PJ, Fink M, Dev A, Roberts SK, Nicoll A, et al. Surveillance improves survival of patients with hepatocellular carcinoma: a prospective population-based study. *Medical Journal of Australia*. 2018; **209**(8):348-54.

98. Tang A, Hallouch O, Chernyak V, Kamaya A, Sirlin CB. Epidemiology of hepatocellular carcinoma: target population for surveillance and diagnosis. *Abdominal Radiology*. 2018; **43**(1):13-25.

99. Michael CW, David BP, Mark WS, Leon AA, Gary PJ. Hepatocellular carcinoma in Australia 1982-2014: Increasing incidence and improving survival. *Liver International*. 2018:1-9.

100. Ghouri YA, Mian I, Rowe JH. Review of hepatocellular carcinoma: Epidemiology, etiology, and carcinogenesis. *Journal of carcinogenesis*. 2017; **16**.

101. Younossi ZM, Otgonsuren M, Henry L, Venkatesan C, Mishra A, Erario M, et al. Association of nonalcoholic fatty liver disease (NAFLD) with hepatocellular carcinoma (HCC) in the United States from 2004 to 2009. *Hepatology*. 2015; **62**(6):1723-30.

102. Cocker F, Chien Yee K, Palmer AJ, de Graaff B. Increasing incidence and mortality related to liver cancer in Australia: time to turn the tide. *Australian and New Zealand journal of public health*. 2019; **43**(3):267-73.

103. Schaffar R, Rapiti E, Rachet B, Woods L. Accuracy of cause of death data routinely recorded in a population-based cancer registry: impact on cause-specific survival and validation using the Geneva Cancer Registry. *BMC cancer*. 2013; **13**(1):609.

104. Hinchliffe SR, Abrams KR, Lambert PC. The impact of under and over-recording of cancer on death certificates in a competing risks analysis: a simulation study. *Cancer epidemiology*. 2013; **37**(1):11-9.

105. Tan KS. Misclassification of the actual causes of death and its impact on analysis: a case study in non-small cell lung cancer. *Lung Cancer*. 2019; **134**:16-24.

106. Barlow L, Westergren K, Holmberg L, Talbäck M. The completeness of the Swedish Cancer Register–a sample survey for year 1998. *Acta oncologica*. 2009; **48**(1):27-33.

107. Tasmanian Cancer Registry. *Cancer in Tasmania Incidence and Mortality 2015*. Menzies's Institute for Medical Research 2015.

108. World Health Organization. *International Statistical Classification of Diseases and Related Health Problems* 2010.

109. Dickman PW, Coviello E. Estimating and modeling relative survival. *The Stata Journal*. 2015; **15**(1):186-215.

110. McHugh ML. Interrater reliability: the kappa statistic. *Biochemia medica: Biochemia medica*. 2012; **22**(3):276-82.

111. Dutz A, Löck S. Competing risks in survival data analysis. *Radiotherapy and Oncology*. 2019; **130**:185-9.

112. Sherif ZA, Saeed A, Ghavimi S, Nouraie S-M, Laiyemo AO, Brim H, et al. Global epidemiology of nonalcoholic fatty liver disease and perspectives on US minority populations. *Digestive diseases and sciences*. 2016; **61**(5):1214-25.

113. Wong RJ, Cheung R, Ahmed A. Nonalcoholic steatohepatitis is the most rapidly growing indication for liver transplantation in patients with hepatocellular carcinoma in the U.S. *Hepatology*. 2014; **59**(6):2188-95.

114. Gupta A, Dixon E. Epidemiology and risk factors: intrahepatic cholangiocarcinoma. *Hepatobiliary surgery and nutrition*. 2017; **6**(2):101.

115. Amin J, O'Connell D, Bartlett M, Tracey E, Kaldor J, Law M, et al. Liver cancer and hepatitis B and C in New South Wales, 1990–2002: a linkage study. *Australian and New Zealand journal of public health*. 2007; **31**(5):475-82.

116. Carville KS, MacLachlan JH, Thursfield V, Cowie BC. Hepatocellular carcinoma over three decades in Victoria, Australia: epidemiology, diagnosis and trends, 1984-2013. *Intern Med J*. 2018; **48**(7):835-44.

117. Hong TP, Gow P, Fink M, Dev A, Roberts S, Nicoll A, et al. Novel population-based study finding higher than reported hepatocellular carcinoma incidence suggests an updated approach is needed. *Hepatology*. 2016; **63**(4):1205-12.

118. Australian Bureau of Statistics. Australian Statistical Geography Standard (ASGS): Volume 4 - Significant Urban Areas, Urban Centres and Localities, Section of State. 2016.

119. Australian Bureau of Statistics. *Standard Australian Classification of Countries*. 2016.

120. Australian Bureau of Statistics. *Life Tables, States, Territories and Australia, 2007 - 2015*.2018.

121. Wallace MC, Preen DB, Short MW, Adams LA, Jeffrey GP. Hepatocellular carcinoma in Australia 1982-2014: Increasing incidence and improving survival. *Liver Int*. 2019; **39**(3):522-30.

122. Lapointe-Shaw L, Georgie F, Carlone D, Cerocchi O, Chung H, Dewit Y, et al. Identifying cirrhosis, decompensated cirrhosis and hepatocellular carcinoma in health administrative data: A validation study. *PLoS One*. 2018; **13**(8):e0201120.

123. Park H, Wang W, Henry L, Nelson DR. Impact of All-Oral Direct-Acting Antivirals on Clinical and Economic Outcomes in Patients With Chronic Hepatitis C in the United States. *Hepatology*. 2019; **69**(3):1032-45.

124. Liang W, Chikritzhs T, Pascal R, Binns CW. Mortality rate of alcoholic liver disease and risk of hospitalization for alcoholic liver cirrhosis, alcoholic hepatitis and alcoholic liver failure in Australia between 1993 and 2005. *Intern Med J.* 2011; **41**(1a):34-41.

125. Fialoke S, Malarstig A, Miller MR, Dumitriu A, editors. Application of machine learning methods to predict non-alcoholic steatohepatitis (NASH) in non-alcoholic fatty liver (NAFL) patients. AMIA Annual Symposium Proceedings; 2018: American Medical Informatics Association.

126. Dugan J, Shubrook J. International Classification of Diseases, 10th Revision, Coding for Diabetes. *Clinical Diabetes*. 2017; **35**(4):232-8.

127. Australian Bureau of Statistics. *Net overseas migration, Arrivals and Departures, State/territory, Major groupings and visa*. 2019.

128. Amarapurkar D. Asia-Pacific working party on prevention of hepatocellular carcinoma. Application of surveillance programs for hepatocellular carcinoma in the Asia-Pacific Region. *J Gastroenterol Hepatol*. 2009; **24**:955-61.

129. Gastroenterological Society of Australia. *Australia and New Zeland Chronic Hepatitis B (CHB) Recommendations*. 2010.

130. Sinclair M, Roberts S, Kemp W, Knight V, Dev A, Gow P, et al. Epidemiology of hepatitis Bassociated hepatocellular carcinoma in Victoria. *Internal medicine journal*. 2013; **43**(5):501-6.

131. Australian Bureau of Statistics. *National Health Survey*. 2018.

132. Australian Institute of Health and Welfare. Deaths data Australian Institue of Health and Welfare2018 [Available from: <u>https://www.aihw.gov.au/about-our-data/our-data-</u> <u>collections/national-mortality-database/deaths-data</u>.

133. Tasmania Government. The Tasmanian Registry of Births, Deaths and Marriages 2020 [cited 2020 10 June]. Available from: <u>https://www.justice.tas.gov.au/</u>.

134. McGivern L, Shulman L, Carney JK, Shapiro S, Bundock E. Death certification errors and the effect on mortality statistics. *Public Health Reports*. 2017; **132**(6):669-75.

135. Filippatos G, Andriopoulos P, Panoutsopoulos G, Zyga S, Souliotis K, Gennimata V, et al. The quality of death certification practice in Greece. *Hippokratia*. 2016; **20**(1):19.

136. Middleton D, Anderson R, Billingsly T, Virgil NBM, Wimberly Y, Lee R. Death certification: issues and interventions. *Open J Prev Med*. 2011; **1**(3):167-70.

137. Yin D, Morris CR, Bates JH, German RR. Effect of misclassified underlying cause of death on survival estimates of colon and rectal cancer. *Journal of the National Cancer Institute*. 2011; **103**(14):1130-3.

138. Australian Bureau of Statistics. Causes of Death, Australia: Doctor Certified Deaths, Summary Tables: Australian Bureau of Statistics; 2020 [Available from: <u>https://www.abs.gov.au/statistics/health/causes-death/causes-death-australia-doctor-certified-</u> <u>deaths-summary-tables/latest-release#doctor-certified-mortality-data</u>.

139. Mariotto AB, Noone A-M, Howlader N, Cho H, Keel GE, Garshell J, et al. Cancer survival: an overview of measures, uses, and interpretation. *Journal of the National Cancer Institute Monographs*. 2014; **2014**(49):145-86.

140. American Cancer Society. *Cancer facts and figures*. American Cancer Society; 2020.

141. Australian Institute of Health and Welfare. *Cancer Data in Australia* Canberra; 2020.

142. Jutte DP, Roos LL, Brownell MD. Administrative record linkage as a tool for public health research. *Annual review of public health*. 2011; **32**:91-108.

143. Howlader N, Mariotto AB, Woloshin S, Schwartz LM. Providing clinicians and patients with actual prognosis: cancer in the context of competing causes of death. *Journal of the National Cancer Institute Monographs*. 2014; **2014**(49):255-64.

144. Lambert P, Dickman P, Nelson C, Royston P. Estimating the crude probability of death due to cancer and other causes using relative survival models. *Statistics in medicine*. 2010; **29**(7-8):885-95.

145. Howlader N, Ries LA, Mariotto AB, Reichman ME, Ruhl J, Cronin KA. Improved estimates of cancer-specific survival rates from population-based data. *JNCI: Journal of the National Cancer Institute*. 2010; **102**(20):1584-98.

146. Kudo M, Izumi N, Kubo S, Kokudo N, Sakamoto M, Shiina S, et al. Report of the 20th Nationwide follow-up survey of primary liver cancer in Japan. *Hepatology Research*. 2020; **50**(1):15-46.

147. Kudo M. Management of hepatocellular carcinoma in Japan as a world-leading model. *Liver Cancer*. 2018; **7**(2):134-47.

148. Kudo M. Surveillance, diagnosis, treatment, and outcome of liver cancer in Japan. *Liver cancer*. 2015; **4**(1):39-50.

149. Weber MF, Marshall HM, Rankin N, Duffy S, Fong KM, Dunlop K, et al. Cancer screening in Australia: future directions in melanoma, Lynch syndrome, and liver, lung and prostate cancers. *Public health research & practice*. 2019; **29**(2).

150. Bruix J, Sherman M. Management of hepatocellular carcinoma. *Hepatology*. 2005; **42**(5):1208-36.

151. Allard N, Cabrie T, Wheeler E, Richmond J, MacLachlan J, Emery J, et al. The challenge of liver cancer surveillance in general practice: do recall and reminder systems hold the answer? *Australian Family Physician*. 2017; **46**(11):859-64.

152. Simmons OL, Feng Y, Parikh ND, Singal AG. Primary care provider practice patterns and barriers to hepatocellular carcinoma surveillance. *Clinical Gastroenterology and Hepatology*. 2019; **17**(4):766-73.

153. Dalton-Fitzgerald E, Tiro J, Kandunoori P, Halm EA, Yopp A, Singal AG. Practice patterns and attitudes of primary care providers and barriers to surveillance of hepatocellular carcinoma in patients with cirrhosis. *Clinical Gastroenterology and Hepatology*. 2015; **13**(4):791-8. e1.

154. Farvardin S, Patel J, Khambaty M, Yerokun OA, Mok H, Tiro JA, et al. Patient-reported barriers are associated with lower hepatocellular carcinoma surveillance rates in patients with cirrhosis. *Hepatology*. 2017; **65**(3):875-84.

155. Ispas S, So S, Toy M. Barriers to disease monitoring and liver cancer surveillance among patients with chronic hepatitis B in the United States. *Journal of community health*. 2019; **44**(3):610-25.

156. Singal AG, Lampertico P, Nahon P. Epidemiology and surveillance for hepatocellular carcinoma: new trends. *Journal of hepatology*. 2020; **72**(2):250-61.

157. Worthington J, Feletto E, Lew J, Broun K, Durkin S, Wakefield M, et al. Evaluating health benefits and cost-effectiveness of a mass-media campaign for improving participation in the National bowel cancer screening program in Australia. *Public health*. 2020; **179**:90-9.

158. Lew J-B, Feletto E, Wade S, Caruana M, Kang Y-J, Nickson C, et al. Benefits, harms and costeffectiveness of cancer screening in Australia: an overview of modelling estimates. *Public health research & practice*. 2019; **29**(2).

159. Australian Institute of Health and Welfare. Cancer screening programs: quarterly data: Canberra: AIHW; 2020 [Available from: <u>https://www.aihw.gov.au/reports/cancer-screening-programs-participation</u>.

## Appendix A: A publication co-authored by the candidate

Nguyen ALT, **Nguyen HTT**, Yee KC, Palmer AJ, Blizzard CL, de Graaff B. A Systematic Review and Narrative Synthesis of Health Economic Evaluations of Hepatocellular Carcinoma Screening Strategies. Value in Health. 2021.

I co-authored the paper in which I helped with data acquisition, data analysis and interpretation, provision of study materials or patients.





### ScienceDirect

Contents lists available at sciencedirect.com Journal homepage: www.elsevier.com/locate/jval

### A Systematic Review and Narrative Synthesis of Health Economic Evaluations of Hepatocellular Carcinoma Screening Strategies

Anh Le Tuan Nguyen, MSc, Hoa Thi Thu Nguyen, BSc, Kwang Chien Yee, MBBS, BMedSci, FRACP, PhD, Andrew J. Palmer, MBBS, BMedSci, PhD, Christopher Leigh Blizzard, PhD, Barbara de Graaff, PhD

### ABSTRACT

*Objectives*: Many economic evaluations of hepatocellular carcinoma (HCC) screenings have been conducted; however, these vary substantially with regards to screening strategies, patient group, and setting. This review aims to report the current knowledge of the cost-effectiveness of screening and describe the published data.

*Methods:* We conducted a search of biomedical and health economic databases up to July 2020. We included full and partial health economic studies if they evaluated the costs or outcomes of HCC screening strategies.

*Results:* The review included 43 studies. Due to significant heterogeneity in key aspects across the studies, a narrative synthesis was conducted. Most studies reported using ultrasound or alpha fetoprotein as screening strategies. Screening intervals were mostly annual or biannual. Incidence, diagnostic performance, and health state utility values were the most critical parameters affecting the cost-effectiveness of screening. The majority of studies reported HCC screening to be cost-effective, with the biannual ultrasound + alpha fetoprotein standing out as the most cost-effective strategy. However, few studies considered the utilization rate, and none considered the diagnostic performance of ultrasound in the context of central adiposity. Computed tomography and magnetic resonance imaging were also evaluated, but its cost-effectiveness was still controversial.

*Conclusions:* Although many studies suggested HCC screening was cost-effective, substantial limitations of the quality of these studies means the results should be interpreted with caution. Future modeling studies should consider the impact of central adiposity on the precision of ultrasound, real-world utilization rates and projections of increased HCC incidence.

Keywords: cost-effectiveness, hepatocellular carcinoma, primary liver cancer, screening, surveillance

VALUE HEALTH. 2021; ■(■): ■-■

### Introduction

Primary liver cancer (PLC) is the sixth most commonly occurring cancer in the world<sup>1</sup> and third most common cause of cancer mortality.<sup>2-4</sup> In Australia, from 1982 to 2014, the age standardized incidence and mortality rate of this cancer increased by 306% and 184%, respectively.<sup>5</sup> Meanwhile, the age standardized incidence and mortality rate mortality rates for colorectal, lung, breast, and all cancers combined decreased during the same period.<sup>5</sup> PLC is the fastest increasing cause of cancer mortality in this country.<sup>6</sup>

Furthermore, incidence is expected to increase in the future.<sup>7,8</sup> Two main drivers are the trajectory of chronic hepatitis B virus (HBV) infections and increasing prevalence of obesity and nonalcoholic fatty liver disease (NAFLD).<sup>9</sup> HBV is strongly oncogenic.<sup>10</sup> and is endemic in many East Asian and Sub-Saharan Africa countries.<sup>11</sup> As such, the increasing numbers of people migrating to Australia from these countries is projected to contribute to increasing incidence of PLC.<sup>12</sup> Furthermore, NAFLD is associated with an increased risk of PLC, and is now the second most common indication for liver transplant in the United States.<sup>13</sup>

PLC places a substantial economic burden on the Australian health system, with an estimated annual cost of AUD50.2 million for patients admitted to hospitals in 2012.<sup>14</sup> The costs associated with hepatocellular carcinoma (HCC), the most common PLC form (accounts for 80% of all cases).<sup>15</sup> have also been reported for the United States and Canada. The average annual cost per patient with HCC in the United States was \$147 000 in 2016.<sup>16</sup> and the extrapolated net cost (ie, difference between the mean costs for patients with HCC and matched controls without HCC) to the whole Canadian population was \$25 million annually in 2009, and these costs are likely to rise.<sup>17</sup>

Globally, HCC represents the fifth most common cancer in men and seventh in women.<sup>3,4</sup> Timing of HCC diagnosis is critical to patient outcomes. Patients diagnosed in early stages have more treatment options available and improved outcomes than those diagnosed at later stages, for whom palliative care is often the only option.<sup>18,19</sup>

1098-3015/\$36.00 - see front matter Copyright @ 2021, ISPOR-The Professional Society for Health Economics and Outcomes Research. Published by Elsevier Inc.

#### 2 VALUE IN HEALTH

However, HCC is infrequently detected early.<sup>17,20-22</sup> Late diagnosis is partly related to the lack of HCC symptoms in the early stages.<sup>23-25</sup> Additionally, the cost of diagnostic tests and low awareness of the disease among primary care physicians have been cited as factors contributing to delayed diagnosis.<sup>21,26</sup> Further complicating this is the fact that HCC can develop in the absence of risk factors (eg, cirrhosis) and in patients with normal liver histology.<sup>27-29</sup> Between 30% to 50% of HBV-related HCC cases occur in the absence of cirrhosis.<sup>15,30</sup> Additionally, up to 90% of HBV carriers worldwide are unaware of their HBV status.<sup>31</sup> An Australian study reported only 26% of people newly diagnosed with HCC were at early stages.<sup>32</sup> Similar results have been published in the United States and Austria, with less than 20% of HCC cases diagnosed at early stages.<sup>33,34</sup>

HCC screening (or surveillance) for high-risk populations is recommended as a method for early detection of the disease by many authors<sup>4,8,22,35-37</sup> and professional bodies.<sup>38-41</sup> The recommended screening approach is ultrasound  $\pm$  alpha-fetoprotein (AFP) at regular intervals for high-risk patients (cirrhotic or noncirrhotic chronic HBV/hepatitis C virus [HCV] carriers).

Many economic evaluations of HCC screening have been conducted; however, these vary substantially with regard to intervention, patient group, and setting. To our knowledge, one systematic review on the health economic aspects of HCC screening has been published in 2012.<sup>36</sup> This review only included full economic evaluations and outcomes were limited to qualityadjusted life years (QALYs). Since then, several health economic evaluations of screening have been published. The aim of this article is to present the current knowledge and narratively synthesize the published data.

### Methods

This review was conducted in accordance with the PRISMA Protocol.<sup>42</sup> The study protocol was registered on Prospero (CRD42019130358). The studies selection followed the PICO model (Patients: high HCC risk; Intervention: HCC screening strategies; Comparison: alternative screening strategies or nonscreening; Outcomes: cost/effectiveness of strategies).

#### Search Strategy

The review was performed using published guideline.<sup>43</sup> The initial search was carried out in April 2019 and a further search was performed in July 2020 in PubMed, Embase, Scopus, Cochrane, Econlit, Centre for Review and Dissemination, and cost-effectiveness analysis registry. Searches were also conducted on grey literature (Open Grey and Google Scholar) to capture additional studies. Search strategies were developed in consultation with a research librarian and the economic concepts were based on a published filter.<sup>44,45</sup> Details of the search strategies are in the Supplemental Materials 1 found at https://doi.org/10.1016/j.jval.2 020.11.014.

#### Study Selection

Studies were included if they were full/partial health economic analyses that evaluated the costs or outcomes of HCC screening strategies. The outcomes were health economic measures: QALYs, quality-adjusted life expectancy (QALEs), health-state utility values (HSUVs), life expectancy, and life years (LYs). Studies published in English, German, French, Italian, and Vietnamese were included. Studies were excluded if they were reviews, metaanalyses, letters, editorials, conference abstracts, or studies of other PLC types (eg, angiosarcoma, cholangiocarcinoma) or HCC treatment. Titles and abstracts were independently screened by 2 reviewers (A.N. and H.N.). Full texts of accepted abstracts and those in need of further clarification were reviewed for the second stage. Any discrepancies arising that could not be resolved during the systematic search among 2 reviewers (A.N. and H.N.) were decided through discussion with the third reviewer (B.dG.).

### **Quality Assessment of Studies**

The reporting quality of the accepted articles were assessed against the 24-item Consolidated Health Economic Evaluation Reporting Standards checklist.<sup>46</sup> For items fulfilled, a score of "1" was given, otherwise a "0" was given. A percentage was calculated to compare scores between studies. The denominator was calculated by summing the number of applicable items per study, and the numerator by summing the scores. Studies were deemed to be of high (>75%), moderate (50%-75%), or low (<50%) reporting quality. Although use of the Consolidated Health Economic Evaluation Reporting Standards checklist is most appropriate for studies published subsequent to publication of the checklist, we applied this to all studies as it provides a benchmark for study comparisons. A.N. conducted the quality assessment, and a 10% random selection of articles was reviewed by B.dG. to check for agreement.

### Synthesis of Results

The initial intention, as per the protocol, was to conduct a meta-analysis. As significant heterogeneity in key aspects across the studies was identified (including the economic evaluation methods, patient groups, screening interventions, time horizons, perspectives, outcome measures, and health systems), a narrative synthesis was conducted following published guidance.<sup>47</sup>

### Results

### Search Outcomes

Figure 1 illustrates the study selection process. The searches yielded a total of 6748 studies. After removing 2268 duplicates, another 4342 records were excluded by title and abstract screening; 138 full texts were assessed, and the remaining 43 articles were included.

#### Study Characteristics

Most studies (34.9%, n = 15) were conducted in Asia, 12 (27.9%) in the United States, and 7 (16.3%) in Europe. The majority (65.1%, n = 28) of studies were model-based, of which 22 used Markov model structures and one used both Markov and decision analytical models.<sup>48</sup> For nonmodeling studies describing the populations' demographic characteristics, all studies reported a mean age of >45 years and higher ratios of males to females, with the exception of one study reporting on patients with cirrhosis, which reported 35.4% of the sample were male and 64.6% were female (Table 1).<sup>49</sup>

Twenty-three studies focused on patients with cirrhosis: 6 on HBV carriers, <sup>50-55</sup> and 14 on populations with different liver diseases (cirrhotic or noncirrhotic chronic HBV and HCV, nonalcoholic steatohepatitis, or fibrosis).<sup>48,56-68</sup>

Markov-modeling studies mostly used lifetime horizons,<sup>48,52,53,63,64,69-78</sup> with 4 applying horizons of 10 to 25 years,<sup>62,79-81</sup> 1 of 30 years (for 40-year-old patients),<sup>82</sup> and 1 of between 15 to 50 years for patients of different ages (30, 40, and 50 years old).<sup>66</sup> The horizon was not reported in four modeling studies,<sup>51,54,68,83</sup>



Discount rates were reported in all Markov-modeling studies; a single study applied discounting for both costs and outcomes but did not report the rate used.73 Studies using other modeling approaches did not discount the costs/outcomes, with the exception of a spreadsheet-based study.<sup>50</sup> Discount rates of 3% and 5% were used by  $12^{50,53,62,63,68,70,71,75-77,82,83}$  and  $5^{48,52,64,72,81}$ studies, respectively for both costs and outcomes, 2 studies discounted the costs at 3% but not the outcomes.74,7

All studies considered direct medical costs, with 5 also reporting direct nonmedical costs (food and transportation).<sup>50,53,63,81,84</sup> Three studies adopted a societal perspective, incorporating indirect costs such as productivity losses.<sup>53,63,84</sup> Although 2 other studies also reported adopting a societal perspective, they did not include indirect costs.79,

QALYs were the most frequently used outcome measure (in 15 modeling studies). Other studies used non-preference-based outcomes as effectiveness measures, with LYs gained and HCC cases detected most commonly reported (n = 15).

### **Reporting Quality Assessment**

The overall mean reporting quality score was 67.0% (standard deviation [SD] = 19.0; Table 2). The mean quality score for modeling studies was 73.5% (SD = 14.0) while for nonmodeling studies this was 54.7% (SD = 21.4). For modeling studies, 17 had scores ≥75%, 9 had scores between 50% to 75% and 2 <50%. For nonmodeling studies, 5 were deemed to be of high reporting quality: 3 of moderate and 7 of low quality.

### **Data Synthesis**

All studies used ultrasound ± AFP at annual, biannual, or shorter intervals (3-month in the presence of cirrhosis)48,62,76 as either screening strategies or comparators. Two of 3 studies with shorter screening intervals were conducted in Taiwan,48,62 the other in the United States.<sup>76</sup> Three Italian studies evaluated ultrasound + AFP at 6-month intervals.<sup>79,85,86</sup> Studies set in Japan and the United States evaluated a wide range of screening strategies and intervals. Magnetic resonance imaging (MRI) and computed tomography (CT) screening were mostly included in studies from the United States (7 of 10 studies<sup>67,70,73,74,82,87,88</sup>).

Data were synthesized by 3 groups: cirrhotic, chronic HBV, and high-risk populations with different liver diseases. Patients with chronic HCV were not categorized as a subgroup because all HCVrelated studies targeted either HCV cirrhotic populations or patients with other liver diseases in addition to HCV. Costeffectiveness was based on a willingness-to-pay threshold (WTP), which was within 3 time the given country's gross domestic product per capita.89,9

### **Cirrhosis Only**

Twenty-three studies (53.5%) focused on patients with cirrhosis, with mixed evidence of cost-effectiveness. Studies reporting cost-effectiveness evaluated different strategies and outcomes. An Egyptian study examining biannual ultrasound and ultrasound + AFP compared with no screening reported incremental cost-effectiveness ratios (ICERs) of EGP7907 and

3

--

### 4 VALUE IN HEALTH

### Table 1. General characteristics of included studies (n = 43).

Characteristics	n	%
Publication year 2010-2020 2000-2009 Before 2000	24 13 6	55.8% 30.2% 14.0%
Location of study United States Europe Australia Asia Japan Korea Taiwan Other Other	12 7 3 15 4 4 3 4 6	27.9% 16.3% 7.0% 34.9%
Patient population Cirrhosis only Chronic HBV only High-risk population with different liver disease	23 6 14	53.5% 14.0% 32.5%
Type of economic evaluation Cost-effectiveness analysis Cost-benefit analysis Partial economic evaluation Cost: outcome description Outcome analysis Cost analysis	29 1 13 10 1 2	67.4% 2.3% 30.3%
Perspective Government (healthcare system) Societal Healthcare provider Not reported	27 5 1 10	62.8% 11.6% 2.3% 23.3%
Study design Markov disease-state transition model Other model (decision analytic, nomogram, math, spreadsheet,	22 7	51.2% 16.3%
Stochastic, microsimulation) Nonmodeling studies Retrospective clinical review Prospective clinical trial Prospective population-based study Prospective cohort study RCT	15 7 1 2 3 2	34.9%

HBV indicates hepatitis B; RCT, randomized controlled trial.

8430/QALY respectively, which was within the country's WTP.<sup>49</sup> One Italian study comparing biannual ultrasound with annual ultrasound + AFP showed biannual screening was more costeffective in patients with compensated than decompensated cirrhosis, with an ICER of €1997/quality adjusted life month: less than the country's WTP of €3000/ quality adjusted life month.<sup>79</sup> A US-based study concluded 6-month AFP screening was dominated by 6-month ultrasound + AFP, which was cost-effective with an ICER of \$26 689/QALY compared with no screening.<sup>70</sup> Contrast-enhanced ultrasound in a Japanese study was costeffective compared with no screening (ICER: \$18 384/QALY).<sup>83</sup>

Although the majority of studies reported screening to be costeffective, several studies reported opposing evidence. Those studies compared 6-month ultrasound + AFP with no screening, using cases detected<sup>84–86,91,92</sup> and survival time<sup>72,86,91</sup> as outcome measures. This negative result could imply that, despite the gain in survival time (ranging from 1.5-15 months),<sup>72,86,91</sup> the effect was too small compared with the costs. Similarly, an Indian study used annual CT scans in addition to biannual ultrasound + AFP, adding substantial costs to the screening program.<sup>84</sup>

Seven studies evaluated CT or MRI as screening strategies.<sup>70,73,78,81,82,87,88</sup> Generally, CT and MRI were clinically more effective than ultrasound at the same interval. Two US-based studies using CT and MRI resulted in higher OALYs and OALEs gained than ultrasound.<sup>70,73</sup> However, CT and MRI were much more costly than ultrasound.<sup>70,73,81,82,88</sup> A Korean study estimated an ICER of \$25 202/QALY for the biannual MRI against biannual ultrasound, higher than the WTP of \$20 000.81 However, biannual ultrasound brought about better clinical outcomes than annual CT.73,82,87 A US-based RCT showed biannual ultrasound detected more HCC cases than annual CT, with a lower cost/case detected (\$12 069 vs \$18 768).<sup>87</sup> One US-based modeling study also showed annual CT was dominated by biannual ultrasound.73 Abbreviated MRI (AMRI) was evaluated in 2 studies and reported to be costeffective against ultrasound.<sup>78,88</sup>A Canadian study reported an ICER of CAD 39 681/QALY for AMRI, which was within the WTP of 50 000/QALY.78

### **Chronic HBV**

Six studies focused exclusively on patients with chronic HBV<sup>50-55</sup> and concluded screening this population was costeffective. A Thai study concluded biannual ultrasound  $\pm$  AFP was cost-effective against no screening, with ICERs of 118 796 and 123 451 Baht/QALY, respectively: less than the WTP (160 000 Baht/QALY).<sup>53</sup> CT and MRI were not cost-effective: ICERs of 175 583 and 187 064 Baht/QALY, respectively.

Adding AFP to ultrasound was more cost-effective than ultrasound alone in most studies. A US-based study reported the cost-effectiveness of a staged screening strategy: AFP >10 ng/mL followed by ultrasound compared with biannual ultrasound. Although ultrasound alone detected more early tumors (14 vs 10 cases), its overall cost was more than double that of AFP + ultrasound (\$814 000 vs \$357 000).50 A Korean study reported similar results and suggested the accuracy of detecting HCC would be significantly improved if ultrasound screening incorporated HCC risk predictors (age, sex, cirrhosis status, and AFP).<sup>54</sup> A Singaporean study also showed ultrasound + AFP was more costeffective than ultrasound or AFP separately, and was capable of detecting 90% of cases at early stages.51 However, limitations of this modeling study, including lack of probability distributions and critical parameters (such as transition probabilities from chronic HBV to cirrhosis and HCC) meant these results should be interpreted with caution.

An Australian study assessed biannual ultrasound + AFP screening. Compared with no screening, the ICER was unacceptably high: AUD401 516/QALY. Nevertheless, when pharmacologic treatment of chronic HBV was included, the strategy became costeffective (ICER: AUD12 956/QALY, well below the WTP of AUD50 000<sup>52</sup>).

### High-Risk Populations With Different Liver Diseases

Overall, screening populations with cirrhosis, or no cirrhosis but living with chronic HCV, HBV, nonalcoholic steatohepatitis, or fibrosis appeared to be cost-effective. Two studies concluded ultrasound screening on patients with cirrhosis was more costeffective than for patients with other liver diseases. A Canadian study reported a lower ICER for screening patients with cirrhosis compared to patients with fibrosis using ultrasound at 12- or 6month intervals.<sup>64</sup> Additionally, a Taiwanese study assessed annual ultrasound for patients with cirrhosis with no screening.

5

...

### Table 2. Summary table of included studies.

Author, year	Country	Study design	Screening strategies	Comparators	CE or not	Cheers score
Patients with cirrhosis						
Kim et al, 2019 <sup>81</sup>	Korea	Markov, CEA	MRI 6 mo	US 6 mo	Yes	87.50%
Lima et al, 2019 <sup>78</sup>	Canada	Markov, CEA	US/CT/MRI/AMRI 6 mo	Next least expensive, non- dominated strategy	Yes for some strategies	83.33%
Goossens et al, 2017 <sup>74</sup>	USA	Markov, CEA	US or MRI at different intervals	US 6 mo	Yes for high and intermediate risk	83.33%
Uyei et al, 2019 <sup>76</sup>	USA	Markov, CEA	US 3 or 6 or 12 mo	No screening	Yes	83.33%
Pocha et al, 2013 <sup>87</sup>	USA	RCT, CEA	US+AFP 6 mo	CT 12 mo + AFP 6 mo	No conclusion, CT dominated by US	81.25%
Arguedas, 2003 <sup>70</sup>	USA	Markov, CEA	US/CT/MRI/AFP 6 mo	No screening	Yes, but MRI not CE	79.17%
Bolondi et al, 2001 <sup>86</sup>	Italy	Prospective clinical trial, CEA	US+AFP 6 mo	No screening	No	75.00%
Cadier et al, 2017 <sup>80</sup>	France	Markov, CEA	US 6 mo	No screening	Yes	75.00%
Cucchetti et al, 2012 <sup>79</sup>	Italy	Markov, CEA	US+AFP 6 mo	US+AFP 12 mo	Yes for both strategies	75.00%
Lin et al, 2004 <sup>82</sup>	USA	Markov, CEA	US/CT/AFP 6 or 12 mo	No screening	Yes, CT was CE	75.00%
Patel et al, 2005 <sup>75</sup>	USA	Markov, CEA	US+AFP 6 mo + different treatments	No screening or treatment	Yes	75.00%
Paul et al, 2008 <sup>84</sup>	India	Prospective cohort study, cost analysis	(US+AFP 6 mo) + CT 12 mo	None	Not supported screening	75.00%
Anderssonet al, 2008 <sup>73</sup>	USA	Markov, CEA	US±AFP/CT/MRI 6 or 12 mo	No screening	Yes, but CT and MRI not CE	70.83%
Nouso et al, 2008 <sup>71</sup>	Japan	Markov, CEA	US 6 mo	No screening	No conclusion	70.83%
Parikh et al, 2020 <sup>77</sup>	USA	Markov, CEA	US±AFP 6 mo	No screening	Yes, US+AFP dominated US	70.83%
Thompson Coon et al, 2008 <sup>69</sup>	UK	Markov, CEA	US or AFP 6 or 12 mo	No screening	Yes	70.83%
Eltabbakh et al, 2015 <sup>49</sup>	Egypt	Prospective cohort study, CEA	US±AFP 6 mo	None	Supported screening	64.71%
Sarasin et al, 1996 <sup>72</sup>	Swiss	Markov, CEA	US+AFP 6 mo	No screening	Yes for only few patients with 5-y survival rate of > 80%	60.87%
Violi et al, 2020 <sup>88</sup>	USA	State transition microsimulation model, CEA	Different AMRI 6 mo	US 6 mo	Yes	60.87%
Tanaka et al, 2012 <sup>83</sup>	Japan	Markov, CEA	US or contrast- enhanced US	No screening	Yes	50.00%
Ben Chaabane et al, 2011 <sup>91</sup>	Tunisia	Retrospective clinical review, cost, and outcome description	US+AFP 3 or 6 mo	None	Not supported screening	43.75%
Bolondi et al, 1997 <sup>85</sup>	Italy	Prospective cohort study, cost, and outcome description	US+AFP 6 mo	None	Not supported screening	37.50%
Kchaou-Ouakaa et al, 2007 <sup>92</sup>	Tunisia	Retrospective clinical review, cost, and outcome description	US+AFP 3 or 6 mo	None	Not supported screening	25.00%
						States and the states of the s

continued on next page

### 6 VALUE IN HEALTH

2021

Table 2. Continued
--------------------

Author, year	Country	Study design	Screening strategies	Comparators	CE or not	Cheers score
Patients with chronic HBV						
Robotin et al, 2009 <sup>52</sup>	Australia	Markov, CEA	US+AFP 6 mo ± HBV treatment	No screening	Yes for screening + HBV treatment	95.83%
Sangmala et al, 2014 <sup>53</sup>	Thailand	Markov, CEA	US±AFP/CT/MRI at 6 or 12 mo	No screening	Yes, but CT and MRI not CE	83.33%
Gounder et al, 2016 <sup>50</sup>	USA	Spreadsheet-based model, CEA	AFP -> US 6 mo	US 6 mo	Yes	78.26%
Chung et al, 2017 <sup>54</sup>	South Korea	Nomogram model, efficacy analysis	US + HCC risk predictors (nomogram)	US 6 or 12 mo + blood test	Yes	61.11%
Kang et al, 1992 <sup>51</sup>	Singapore	Mathematical based model, Cost analysis	US or AFP	None	Supported screening	41.18%
Lam et al, 2000 <sup>55</sup>	Hong Kong	Retrospective clinical review, cost, and outcome description	US or AFP 4 or 6 or 12 mo	None	Supported screening	31.25%
Patients with dif	ferent liver o	diseases				
Farhang Zangneh et al, 2018 <sup>64</sup>	Canada	Markov, CEA	US 6 or 12 mo	No screening	Yes for cirrhosis, no for fibrosis	91.67%
Lee et al, 2014 <sup>66</sup>	Korea	Stochastic model, CEA	US+AFP at different intervals, initial and ending ages	Next least expensive, non- dominated strategy	Yes for some strategies	91.30%
Chang et al, 2011 <sup>62</sup>	Taiwan	Markov, CEA	US 3 or 12 mo	No screening	Yes, screening patients with cirrhosis more CE than patients without cirrhosis	82.61%
Kuo et al, 2016 <sup>63</sup>	Taiwan	Markov, CEA	Biomarker -> US or mass US 12 mo	No screening	No	82.61%
Chen et al, 2020 <sup>67</sup>	USA	RCT, CBA	US 6 mo ± CT and/or MRI	Different outreach strategies	Yes, outreach provided good payoff	81.25%
Kwon et al, 2020 <sup>65</sup>	Korea	Retrospective clinical review, cost, and outcome description	US+AFP 6 mo	No screening	No conclusion, but surveillance program improved survival benefits	81.25%
Shih et al, 2010 <sup>48</sup>	Taiwan	Markov + decision analysis model, CEA	Biomarker+US 3 or 6 mo	No screening	Yes	75.00%
Saab et al, 2003 <sup>68</sup>	USA	Markov, CEA	US±AFP/CT 6 mo	AFP 6 mo	No	65.22%
Frey et al, 2015 <sup>57</sup>	Switzerland	Retrospective clinical review, cost, and outcome description	US±AFP 6 mo	None	Supported screening	62.50%
Mary Qian et al, 2010 <sup>59</sup>	Australia	Retrospective clinical review, cost, and outcome description	US+AFP 6 mo	None	Supported screening	62.50%
Ruelas- Villavicencio et al, 2004 <sup>56</sup>	Mexico	Decision analysis model, CEA	US or AFP 6 or 12 mo	Between strategies	No conclusion	39.13%
Mima et al, 1994 <sup>60</sup>	Japan	Prospective population- based study, cost, and outcome description	US+AFP 12 mo	None	Supported screening	37.50%
Larcos et al, 1998 <sup>58</sup>	Australia	Retrospective clinical review; cost and outcome description	US+AFP 6 or 12 mo	None	No conclusion, but US was superior to AFP for HCC detection	31.25%
Rimal et al, 1997 <sup>61</sup>	Japan	Prospective population- based study, cost, and outcome description	Biomarkers	Mass US screening	Yes	31.25%

AFP indicates alpha fetoprotein; AMRI, abbreviated magnetic resonance imaging; CBA, cost-benefit analysis; CE, cost-effective; CEA, cost-effective; SEA, cost-effective; CEA, cos

### Table 3. Evolution of models used in studies.

	1990- 2000	2001- 2010	2011- 2019
Model type			
Decision tree	0	2	0
Markov model	1	9	12
Nomogram	0	0	1
Mathematical	1	0	0
Spreadsheet	0	0	1
Stochastic	0	0	1
Microsimulation	0	0	1
Deterministic sensitivity analysis	1	10	14
Probabilistic sensitivity analysis	; 0	3	7

A significantly lower ICER for patients with cirrhosis (\$16 719/LY) compared with the HBV cohort (\$20 856/LY)<sup>62</sup> was reported; with both ICERs lower than the country's WTP (\$33 000/LY).<sup>63</sup>

A Japanese study compared AFP screening for high-risk individuals (elevated AST and zinc sulphate turbidity) and mass, population-wide ultrasound screening. The authors reported AFP screening was more cost-effective than mass ultrasound screening.<sup>61</sup> These results, however, differ to 2 Australian studies evaluating biannual ultrasound + AFP. These studies reported that more liver abnormalities, which were later diagnosed as HCC, were detected by ultrasound than AFP.<sup>58,59</sup>

### **Evolution of Modeling Studies**

The first modeling HCC screening study was published in 1992, using a mathematical model to estimate the cost of screening/HCC case detected.<sup>51</sup> This model included a limited number of parameters without probability distributions. Markov modeling techniques were used by a single study before 2000<sup>72</sup>; subsequently this approach became more frequently used. Most studies published before 2010 performed univariate and multivariate sensitivity analyses<sup>52,56,68-73,75,82</sup>; from 2011 probabilistic sensitivity analysis was increasingly used to address model parameter uncertainties (Table 3).<sup>53,63,64,77,78,80,81</sup>

### **Key Model Parameters**

#### Incidence

Annual HCC incidence rates were reported in most studies for patients with cirrhosis as one of the most critical parameters affecting cost-effectiveness. Incidence rates ranged between 1.4% to 5%. One US-based study set annual HCC incidence for patients with cirrhosis between 2% to 10%, based on the duration of cirrhosis from 1 to >8 years,<sup>82</sup> and a Taiwanese study used HCC incidence rates of 0.125% and 0.75% for and patients with chronic HBV and cirrhosis, respectively.<sup>62</sup>

### Diagnostic performance

The diagnostic performance of screening was also widely considered as an influential parameter. The sensitivity and specificity of ultrasound were reported to range from 45% to 78% and 80-97%, respectively, and for AFP 41% to 60% and 70% to 93%, respectively. The sensitivity and specificity of ultrasound + AFP ranged between 63% to 85% and 80% to 87%, respectively.

Increased HCC incidence or enhanced screening performance improved the cost-effectiveness of screening compared with no screening. The literature sources, from which those key model parameters' values were drawn, also substantially varied among included studies. Some studies extracted these parameters from a single study, while some estimates were based on a wide range of published articles. However, the method of synthesizing results from several published articles into single parameters was only reported in one study.<sup>53</sup>

### **HSUVs**

Most studies using preference-based outcomes reported HSUVs for cirrhosis and stages of HCC. The majority of US-based studies obtained HSUVs from literature published in the United States by Bennett et al.<sup>93</sup> Kim-WR et al.<sup>94</sup> or Younossi et al.<sup>95</sup> Although Bennett et al used the time trade-off method to elicit utilities for different HCV-related states,<sup>93</sup> Younossi et al used the Health Utility Index Mark 2 to measure the utilities of patients with different liver diseases.<sup>95</sup> Kim-WR et al assigned HSUVs for different liver states but did not state how these were collected.<sup>94</sup> Meanwhile, several studies from Asia obtained utilities from studies by Levy et al<sup>96</sup> and Chong et al.<sup>97</sup> Levy et al elicited utilities for different HBV-related states by using the standard gamble technique.<sup>96</sup> Chong et al used the visual analog scale, standard gamble technique, Health Utility Index Mark 3, and EuroQol Index survey to obtain HCV utilities.<sup>97</sup>

Overall, compensated cirrhosis had higher HSUVs (0.66-0.80) than decompensated cirrhosis (0.39-0.67). The lowest HSUVs for both compensated (0.66) and decompensated cirrhosis (0.39) were from the article by Kim-HL et al.<sup>81</sup> who weighted the values from the study by Levy et al.<sup>96</sup> A HSUV for end-stage HCC of 0.07 was used in studies from Taiwan and Egypt<sup>48,49</sup>; however, both studies derived this from the Dutch disability weight of 0.93.<sup>98</sup> Liver transplantation was the most frequently reported curative treatment, with HSUVs ranging between 0.57 to 0.69  $\leq$ 1 year after the procedure and 0.67 to 0.85  $\geq$ 1 year subsequent. These HSUVs were derived from studies using several multiattribute utility instruments. HSUVs subsequent to surgical resection ranged between 0.70 to 0.80.<sup>64,69,70,75,78</sup> These values were predominantly derived from authors' assumptions or adopted from HSUVs for compensated cirrhosis.

In general, most studies included sufficient HSUVs for relevant health states. However, several studies obtained HSUVs from assumptions, clinical experts, or unclear sources, which made it difficult to assess the HSUVs' quality. Furthermore, because utilities are best measured with the population of interest, HSUVs obtained from other countries might not accurately reflect the targeted population's utilities.

#### Screening utilization

The utilization (or adherence or uptake) rate, measuring the percentage of people who follow the recommended screening interval, was considered in only 4 studies in our review (United States<sup>74,76,77</sup>; Canada<sup>78</sup>). One US-based study incorporated a utilization rate for biannual ultrasound of 15%, based on a metaanalysis.99 This approach was dominated by the no-screening comparator, as it produced lower QALE gains (6.39 vs 6.40) and higher costs (\$44 078 vs \$42 961).<sup>74</sup> When the utilization rate was set at 100%, biannual ultrasound resulted in higher QALE gains (6.51 vs 6.40) together with higher costs (\$51 761 vs \$42 961) in comparison with no screening.<sup>74</sup> Another study from the United States showed biannual ultrasound was cost-effective when 20% of the patients adhered to screening and HCC treatment was suboptimal. If the adherence rate was 100% and HCC treatment became optimal, annual ultrasound would be more costeffective.7

76

7

...

### Discussion

Our systematic review evaluated the cost-effectiveness evidence of HCC screening by describing the costs and outcomes of screening on different patient groups. Overall, the vast majority of studies reported screening to be cost-effective. However, due to the limitations of these studies, care must be taken when interpreting the results.

In general, most of the included studies reported improvements in health outcomes from screening. Evidence of the clinical effectiveness of screening in improving health outcomes and survival has become clearer in recent years. Findings from a 2014 meta-analysis showed HCC screening markedly improved survival as it increased the probability of early tumor detection and patients receiving treatment.<sup>100</sup> A systematic review also showed screened patients were detected with HCC at earlier stages and had improved survival compared with nonscreened patients.<sup>101</sup> However, these conclusions were based upon low-strength evidence due to limitations of the included studies.<sup>101</sup> Another recent systematic review focusing on screening for patients with cirrhosis concluded that despite the low proportion of HCC cases detected, screening facilitated earlier detection of HCC. This resulted in patients being more likely to receive curative treatment options which, in turn, increased life expectancy.<sup>102</sup> An Australian study also reported improved survival among patients who participated in regular surveillance.32

Internationally, Japan implemented effective nationwide screening strategies in the 1980s<sup>103</sup> and Korea in 2003,<sup>65</sup> consisting of ultrasound + AFP for individuals with HBV or cirrhosis, that showed evidence of clinical effectiveness. The 5-year survival rate of patients with HCC in Japan was 43% in 2005, much higher than for the United States (11%-15%) in 2010<sup>104</sup> or Australia (18%) in 2015.<sup>105</sup> The proportion of HCC being diagnosed at early stages in Japan (62%) was twice as high as in Western countries (30%).<sup>104</sup> Korea also experienced a substantial increase in 5-year survival rates for HCC, from 10.7% in 1993 to 26.7% in 2010.<sup>106</sup> It is important to acknowledge that some of these gains may also be partly explained by the population's longer life expectancy and healthcare technology advances.

In our review, biannual ultrasound + AFP stood out as the most cost-effective strategy, which was in line with recommendations from Asia-Pacific and Australia professional bodies<sup>40,41</sup> for patients with cirrhosis and chronic HBV. Our findings support the approach of screening with ultrasound + AFP. Not only did the combination improve the screening sensitivity (despite the trade-off regarding decreased specificity),<sup>107</sup> it was also more cost-effective than ultrasound alone in most included studies.

CT and MRI have been used for screening, especially in the United States, due to their superior sensitivity to ultrasound and AFP.<sup>107,108</sup> CT was deemed cost-effective in 2 studies<sup>70,82</sup> and in a third study when screening utilization was set at 100%.<sup>78</sup> MRI was cost-effective compared with ultrasound when being conducted on high-risk patients with cirrhosis<sup>74</sup> or carried out with fewer sequences (AMRI).<sup>78,88</sup> Nevertheless, the risks associated with CT-based screening, including radiation exposure and contrast-induced renal toxicity,<sup>107</sup> and the time required for MRI investigations, means the risk-benefit needs to be carefully considered.

The most critical parameters affecting the cost-effectiveness of screening were annual HCC incidence rates and diagnostic performance of screening. As the incidence rate has been increasing in many countries, <sup>5,109,110</sup> this could have a substantial impact on the cost-effectiveness of screening.

Of note, no study considered patients' characteristics in the context of the screening strategies' performance. Morbid obesity 2021

can significantly limit the detection of HCC using ultrasound,<sup>111,112</sup> while the accuracy of AFP to detect HCC may be limited in the presence of HCV.<sup>113</sup> These factors are of critical importance when evaluating the cost-effectiveness of screening; as such, future studies should incorporate them.

Screening utilization rate was another important factor affecting screening effectiveness that was largely ignored in earlier studies but has become more common in studies published in recent years. As most studies modeled patients over a lifetime horizon, the cost-effectiveness may not be accurately measured if this was ignored.

The heterogeneity was substantial among influential parameters. HCC incidence rates varied amongst different target group and ranged between 0.125% to 10%, while diagnostic performance of investigations had a broad range of 45% to 97%. Furthermore, HSUVs for a variety of disease states were based on published literature, assumptions, expert opinion, and other sources.

It is essential for future modeling studies to apply these parameters from robust sources to strengthen the cost-effectiveness evidence of screening. Studies should also include the impact of central adiposity on the precision of ultrasound as it could substantially limit detection of HCC. Without including this, particularly in light of increasing global prevalence of overweight and obesity, the cost-effectiveness of ultrasound is not known. Realworld utilization rates should also be incorporated instead of assuming it to be 100%, as well as projections of increased HCC incidence. Additionally, there should be more primary studies evaluating the economic evidence of contrast-enhanced ultrasound on HCC detection due to its high sensitivity and specificity (85% and 91%, respectively).<sup>114</sup>

#### Strengths and limitations

Due to heterogeneity of included studies, a meta-analysis and publication bias assessment were not conducted. We overcame this by conducting a detailed narrative synthesis for reporting the cost-effectiveness evidence of screening programs. Furthermore, the study was not limited to English language as it also included articles in French and Italian.

#### Conclusions

Considering many economic evaluations on HCC screening have been published, this article contributes to the existing literature by describing the published data. Although many studies suggest HCC screening is cost-effective, substantial limitations of these studies mean the results should be interpreted with caution. Future robust studies need to consider all key parameters, including central adiposity, real-world utilization rates, and projections of increasing incidence over time.

### **Supplemental Materials**

Supplementary data associated with this article can be found in the online version at https://doi.org/10.1016/j.jval.2020.11.014.

### **Article and Author Information**

Accepted for Publication: November 17, 2020

#### Published Online: Month xx, xxxx

doi: https://doi.org/10.1016/j.jval.2020.11.014

Author Affiliations: Menzies Institute for Medical Research, University of Tasmania, Hobart, Tasmania, Australia (A. Nguyen, H. Nguyen, Palmer,

q

\_ \_

Blizzard); School of Medicine, University of Tasmania, Hobart, Tasmania, Australia (Yee); Melbourne School of Population and Global Health, University of Melbourne, Melbourne, Victoria, Australia (Palmer).

Correspondence: Barbara de Graaff, PhD, Menzies Institute for Medical Research, University of Tasmania, 17 Liverpool Street, Hobart, Tasmania 7000, Australia. Email: barbara.degraaff@utas.edu.au

Author Contributions: Concept and design: Nguyen A, Yee, Palmer, Blizzard, de Graaff

Acquisition of data: Nguyen A, Nguyen H

Analysis and interpretation of data: Nguyen A, Nguyen H, Yee, Palmer, Blizzard, de Graaff

Drafting of the manuscript: Nguyen A, Palmer, Blizzard, de Graaff

Critical revision of the paper for important intellectual content: Nguyen A, Yee, Palmer, Blizzard, de Graaff

Provision of study materials or patients: Nguyen A, Nguyen H

Obtaining funding: Yee, de Graaff

Supervision: Yee, Palmer, Blizzard, de Graaff

Conflict of Interest Disclosures: The authors reported no conflicts of interest

Funding/Support: The study received no specific funding. Anh Le Tuan Nguyen was supported by a Tasmania Graduate Research Scholarship provided by the University of Tasmania, Australia. Barbara de Graaff is supported by the Sefton Bottomley Liver Cancer Bequest.

Role of the Funder/Sponsor: The funders had no role in the design and conduct of the study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; and decision to submit the manuscript for publication.

Acknowledgment: We would like to thank Michaela Venn, a University of Tasmania librarian, for assistance with developing the comprehensive search strategy for this study.

### REFERENCES

- 1. McGlynn KA, London WT. The global epidemiology of hepatocellular carcioma: present and future. Clin Liver Dis. 2011;15:223-243, vii-x.
- 2. Marcellin P, Kutala BK. Liver diseases: a major, neglected global public health problem requiring urgent actions and large-scale screening. Liver Int. 2018;38(Suppl 1):2-6.
- 3. Jinjuvadia R, Salami A, Lenhart A, et al. Hepatocellular Carcinoma: a decade of hospitalizations and financial burden in the United States. Am J Med Sci. 2017:354:362-369.
- 4. Knudsen ES, Gopal P, Singal AG. The changing landscape of hepatocellular carcinoma: etiology, genetics, and therapy. Am J Pathol. 2014;184:574-583.
- 5. Cocker F, Chien Yee K, Palmer AJ, de Graaff B. Increasing incidence and mortality related to liver cancer in Australia: time to turn the tide. Aust N Z J Public Health. 2019;43:267-273.
- 6. MacLachlan JH, Cowie BC. Liver cancer is the fastest increasing cause of
- cancer death in Australians. Med J Australia. 2012;197:492–493.
  7. Nguyen VTT, Razali K, Amin J, Law MG, Dore GJ. Estimates and projections of hepatitis B-related hepatocellular carcinoma in Australia among people born in Asia-Pacific countries. J Gastroen Hepatol. 2008;23:922-929.
- 8. Asia Pacific Working Party on Prevention of Hepatocellular Carcinoma. Prevention of hepatocellular carcinoma in the Asia-Pacific region: consensus statements. J Gastroenterol Hepatol. 2010;25:657-663.
- Younossi ZM, Koenig AB, Abdelatif D, et al. Global epidemiology of nonal-9. coholic fatty liver disease-meta-analytic assessment of prevalence, incidence, and outcomes. Hepatology. 2016;64:73-84.
- Ringehan M, McKeating JA, Protzer U. Viral hepatitis and liver cancer. Philos Trans R Soc Lond B Biol Sci. 2017;372:20160274.
- 11. World Health Organization. Global Hepatitis Report 2017. Geneva: World Health Organization; 2017.
- Robotin MC, George J, Supramaniam R, Sitas F, Penman AG. Preventing pri-mary liver cancer: how well are we faring towards a national hepatitis B strategy? Med J Aust. 2008;188:363-365.

- 13. Anstee QM, Targher G, Day CP. Progression of NAFLD to diabetes mellitus, cardiovascular disease or cirrhosis. Nat Rev Gastroenterol Hepatol. 2013:10:330-344.
- 14. The Gastroenterological Society of Australia/Australian Liver Association. The economic cost and health burden of liver diseases in Australia. https:// rt.gesa.org.au/membes/files/Resources/Deloitte\_Report\_FINAL\_06032013.pdf. Accessed September 2, 2019.
- Yang JD, Hainaut P, Gores GJ, et al. A global view of hepatocellular carcinoma: trends, risk, prevention and management. Nat Rev Gastroenterol Hepatol. 2019:16:589-604.
- 16. Kaplan DE, Chapko MK, Mehta R, et al. Healthcare costs related to treatment of hepatocellular carcinoma among veterans with cirrhosis in the United States. Clin Gastroenterol H. 2018;16:106-114.
- 17. Thein HH, Isaranuwatchai W, Campitelli MA, et al. Health care costs associated with hepatocellular carcinoma: a population-based study. Hepatology. 2013;58:1375-1384.
- 18. Cabibbo G, Craxi A. Hepatocellular cancer: optimal strategies for screening and surveillance. Digestive diseases (Basel, Switzerland). 2009;27:142-147.
- Balogh J, Victor 3rd D, Asham EH, et al. Hepatocellular carcinoma: a review. | Hepatocell Carcinoma. 2016;3:41-53.
- 20. Tayob N, Lok ASF, Do KA, Feng ZD. improved detection of hepatocellular carcinoma by using a longitudinal alpha-fetoprotein screening algorithm. Clin Gastroenterol H. 2016;14:469-475.
- Pazgan-Simon M, Serafinska S, Janocha-Litwin J, Simon K, Zuwala-Jagiello J. 21. Diagnostic Challenges in primary hepatocellular carcinoma: case reports and review of the literature. Case Rep Oncol Med. 2015;2015:878763.
- 22. The American Cancer Society. Liver Cancer Early Detection, Diagnosis, and Staging. https://www.cancer.org/content/dam/CRC/PDF/Public/8700.00.pdf. Accessed December 10, 2020.
- Tsai WC, Kung PT, Wang YH, Kuo WY, Li YH. Influence of the time interval 23. from diagnosis to treatment on survival for early-stage liver cancer. Plos One. 2018:13(6).
- 24. Dimitroulis D, Damaskos C, Valsami S, et al. From diagnosis to treatment of hepatocellular carcinoma: an epidemic problem for both developed and developing world. World J Gastroenterol. 2017;23:5282-5294.
- Ye SL, Takayama T, Geschwind J, Marrero JA, Bronowicki JP. Current approaches to the treatment of early hepatocellular carcinoma. Oncologist. 2010;15:34-41.
- 26. Patel N, Yopp AC, Singal AG. Diagnostic delays are common among patients with hepatocellular carcinoma. J Natl Compr Canc Netw. 2015;13:543-549.
- Cartier V, Aube C. Diagnosis of hepatocellular carcinoma. Diagn Interv Imag. 27. 2014;95:709-719.
- Forner A, Reig M, Bruix J. Hepatocellular carcinoma. The Lancet. 28. 2018;391:1301-1314.
- Trevisani F, Cantarini MC, Wands JR, Bernardi M. Recent advances in the nat-29 ural history of hepatocellular carcinoma. Carcinogenesis. 2008;29:1299–1305.
- 30. Davis GL, Dempster J, Meler JD, et al. Hepatocellular carcinoma: management of an increasingly common problem. Proc (Bayl Univ Med Cent). 2008:21:266-280.
- 31. Polaris Observatory C. Global prevalence, treatment, and prevention of hepatitis B virus infection in 2016: a modelling study. Lancet Gastroenterol Hepatol. 2018;3:383-403.
- 32. Hong TP, Gow PI, Fink M, et al. Surveillance improves survival of patients with hepatocellular carcinoma: a prospective population-based study. Med J Australia. 2018;209:348-354.
- 33. Khalaf N, Ying J, Mittal S, et al. Natural history of untreated hepatocellular carcinoma in a US cohort and the role of cancer surveillance. Clin Gastroenterol H. 2017;15:273-281.
- 34. Hucke F, Sieghart W, Schoniger-Hekele M, Peck-Radosavljevic M, Muller C. Clinical characteristics of patients with hepatocellular carcinoma in Austria: is there a need for a structured screening program? Wien Klin Wochenschr. 2011;123:542-551.
- 35. Kennedy NA, Rodgers A, Altus R, et al. Optimisation of hepatocellular carcinoma surveillance in patients with viral hepatitis: a quality improvement study. Intern Med J. 2013;43:772-777.
- 36. Ruggeri M. Hepatocellular carcinoma: cost-effectiveness of screening. A systematic review. Risk Manag Healthc Policy. 2012;5:49-54.
- Thompson Coon J, Rogers G, Hewson P, et al. Surveillance of cirrhosis for hepatocellular carcinoma: systematic review and economic analysis. Health Technol Assess. 2007;11:1-206.
- Heimbach JK, Kulik LM, Finn RS, et al. AASLD guidelines for the treatment of hepatocellular carcinoma, *Hepatology*, 2018;67:358–380. European Association For The Study Of The Liver, EASL-EORTC clinical 39.
- practice guidelines: management of hepatocellular carcinoma. J Hepatol. 2012;56:908-943.
- Omata M, Cheng A-L, Kokudo N, et al. Asia-Pacific clinical practice guidelines 40. on the management of hepatocellular carcinoma: a 2017 update. Hepatol Int. 2017:11:317-370.

#### 10 VALUE IN HEALTH

- Moher D, Shamseer L, Clarke M, et al. Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015 statement. Syst Rev. 2015;4.
   Higgins JP, Green S. Chapter 15: Incorporating economics evidence. In:
- Higgins JP, Green S. Chapter 15: Incorporating economics evidence. In: Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0. The Cochrane Collaboration; 2011.
- Centre for Reviews and Dissemination. NHS EED MEDLINE using OvidSP. https://www.crd.york.ac.uk/crdweb/searchstrategies.asp#nhseedmedline. Accessed August 11, 2020.
- Neyt M, Chalon PX. Search MEDLINE for Economic Evaluations: tips to translate an OVID strategy into a PubMed One. *Pharmacoeconomics*. 2013;31:1087–1090.
- Husereau D, Drummond M, Petrou S, et al. Consolidated Health Economic Evaluation Reporting Standards (CHEERS) statement. BMJ. 2013;346:f1049.
- Popay J, Roberts H, Sowden A, et al. Guidance on the conduct of narrative synthesis in systematic reviews. 1. A product from the ESRC methods programme Version; 2006:b92.
- Shih ST, Crowley S, Sheu JC. Cost-effectiveness analysis of a two-stage screening intervention for hepatocellular carcinoma in Taiwan. J Formos Med Assoc. 2010;109:39–55.
- Eltabbakh M, Zaghla H, Abdel-Razek W, et al. Utility and cost-effectiveness of screening for hepatocellular carcinoma in a resource-limited setting. *Med Oncol.* 2015;32:432.
- Gounder PP, Bulkow LR, Meltzer MI, et al. Cost-effectiveness analysis of hepatocellular carcinoma screening by combinations of ultrasound and alphafetoprotein among Alaska Native people, 1983-2012. Int J Circumpolar Health. 2016;75:31115.
- Kang JY, Lee TP, Yap I, Lun KC. Analysis of cost-effectiveness of different strategies for hepatocellular carcinoma screening in hepatitis B virus carriers. J Gastroenterol Hepatol. 1992;7:463–468.
- Robotin MC, Kansil M, Howard K, et al. Antiviral therapy for hepatitis B-related liver cancer prevention is more cost-effective than cancer screening. J Hepatol. 2009;50:990–998.
- 53. Sangmala P, Chaikledkaew U, Tanwandee T, Pongchareonsuk P. Economic evaluation and budget impact analysis of the surveillance program for hepatocellular carcinoma in thai chronic hepatitis B patients. Asian Pac J Cancer Prev. 2014;15:8993–9004.
- Chung JW, Jang ES, Kim J, et al. Development of a nomogram for screening of hepatitis B virus-associated hepatocellular carcinoma. *Oncotarget*. 2017;8:106499–106510.
- Lam CLK. Screening for hepatocellular carcinoma (HCC): is it cost-effective? Hong Kong Practitioner. 2000;22:546–551.
- Ruelas-Villavicencio AL, Vargas-Vorackova F. In whom, how and how often is surveillance for hepatocellular carcinoma cost-effective? Ann Hepatol. 2004;3:152–159.
- Frey RS, Boldanova T, Heim M. Ultrasound surveillance for hepatocellular carcinoma: real-life performance in a hepatology outpatient clinic. Swiss Med Wkly. 2015;145:w14200.
- Larcos G, Sorokopud H, Berry G, Farrell GC. Sonographic screening for hepatocellular carcinoma in patients with chronic hepatitis or cirrhosis: an evaluation. AJR Am J Roentgenol. 1998;171:433–435.
- Mary Qian YY, Rong Yuwei J, Angus P, et al. Efficacy and cost of a hepatocellular carcinoma screening program at an Australian teaching hospital. J Gastroenterol Hepatol. 2010;25:951–956.
- Mima S, Sekiya C, Kanagawa H, et al. Mass screening for hepatocellular carcinoma: experience in Hokkaido, Japan. J Gastroenterol Hepatol. 1994;9:361– 365.
- Rimal N, Ikeda S, Taketa K, Kariya T. Mass screening for early detection of hepatocellular carcinoma by setting a high-risk population with alphafetoprotein and its glycoforms. *Hepatol Res.* 1997;9:9–19.
- Chang Y, Lairson DR, Chan W, Lu SN, Aoki N. Cost-effectiveness of screening for hepatocellular carcinoma among subjects at different levels of risk. J Eval Clin Pract. 2011;17:261–267.
- Kuo MJ, Chen HH, Chen CL, et al. Cost-effectiveness analysis of populationbased screening of hepatocellular carcinoma: comparing ultrasonography with two-stage screening. World J Gastroenterol. 2016;22:3460–3470.
- 64. Farhang Zangneh H, Wong WWL, Sander B, et al. Cost effectiveness of hepatocellular carcinoma surveillance after a sustained virologic response to therapy in patients with hepatitis c virus infection and advanced fibrosis. *Clin Gastroenterol Hepatol*, 2019;17:1840–1849.e16.
- Kwon JW, Tchoe HJ, Lee J, et al. The impact of national surveillance for liver cancer: results from real-world setting in Korea. *Gut Liver*. 2020;14:108–116.
- 66. Lee YH, Choi KS, Jun JK, et al. Cost-effectiveness of liver cancer screening in adults at high risk for liver cancer in the republic of Korea. *Cancer Res Treat*. 2014;46:223.
- 67. Chen Y, Lee J-Y, Sridhar S, et al. Improving cancer outreach effectiveness through targeting and economic assessments: insights from a randomized field experiment. *Journal of Marketing*. 2020;84:1–27.

- Saab S, Ly D, Nieto J, et al. Hepatocellular carcinoma screening in patients waiting for liver transplantation: a decision analytic model. *Liver Transpl.* 2003;9:672–681.
- Thompson Coon J, Rogers G, Hewson P, et al. Surveillance of cirrhosis for hepatocellular carcinoma: a cost-utility analysis. Br J Cancer. 2008;98:1166–1175. 70. Arguedas MR, Chen VK, Eloubeidi MA, Fallon MB. Screening for hepatocel-
- Ingaces and creative product in heating the second product in the second pr
- Pouso K, Tanaka H, Genasu S, et al. Cost-enertweness of the surventance program of hepatocellular carcinoma depends on the medical circumstances. J Gastroenterol Hepatol. 2008;23:437–444.
- Sarasin FP, Giostra E, Hadengue A. Cost-effectiveness of screening for detection of small hepatocellular carcinoma in western patients with Child-Pugh class a cirrhosis. Am J Med. 1996;101:422–434.
- Andersson KL, Salomon JÅ, Goldie SJ, Chung RT. Cost effectiveness of alternative surveillance strategies for hepatocellular carcinoma in patients with cirrhosis. *Clin Gastroenterol Hepatol.* 2008;6:1418–1424.
   Goossens N, Singal AG, King LY, et al. Cost-effectiveness of risk score-
- Goossens N, Singal AG, King LY, et al. Cost-effectiveness of risk scorestratified hepatocellular carcinoma screening in patients with cirrhosis. *Clin Transl Gastroenterol.* 2017;8:e101.
- Patel D, Terrault NA, Yao FY, Bass NM, Ladabaum U. Cost-effectiveness of hepatocellular carcinoma surveillance in patients with hepatitis C virusrelated cirrhosis. *Clin Gastroenterol H.* 2005;3:75–84.
- 76. Uyei J, Taddei TH, Kaplan DE, et al. Setting ambitious targets for surveillance and treatment rates among patients with hepatitis C related cirrhosis impacts the cost-effectiveness of hepatocellular cancer surveillance and substantially increases life expectancy: a modeling study. *PLoS One*. 2019;14(8): e0221614.
- Parikh ND, Singal AG, Hutton DW, Tapper EB. Cost-effectiveness of hepatocellular carcinoma surveillance: an assessment of benefits and harms. *Am J Gastroenterol.* 2020;115(10):1642–1649.
- Lima PH, Fan B, Berube J, et al. Cost-utility analysis of imaging for surveillance and diagnosis of hepatocellular carcinoma. AJR Am J Roentgenol. 2019:1–9.
- Cucchetti A, Trevisani F, Cescon M, et al. Cost-effectiveness of semi-annual surveillance for hepatocellular carcinoma in cirrhotic patients of the Italian Liver Cancer population. J Hepatol. 2012;56(5):1089–1096.
- Cadier B, Bulsei J, Nahon P, et al. Early detection and curative treatment of hepatocellular carcinoma: a cost-effectiveness analysis in France and in the United States. *Hepatology*. 2017;65(4):1237–1248.
- Kim HL, An J, Park JA, et al. Magnetic resonance imaging is cost-effective for hepatocellular carcinoma surveillance in high-risk patients with cirrhosis. *Hepatology*. 2019;69(4):1599–1613.
- Lin OS, Keeffe EB, Sanders GD, Owens DK. Cost-effectiveness of screening for hepatocellular carcinoma in patients with cirrhosis due to chronic hepatitis C. Aliment Pharmacol Ther. 2004;19(11):1159–1172.
- Tanaka H, Iijima H, Nouso K, et al. Cost-effectiveness analysis on the surveillance for hepatocellular carcinoma in liver cirrhosis patients using contrast-enhanced ultrasonography. *Hepatol Res.* 2012;42(4):376-3 84
- Paul SB, Sreenivas V, Gulati MS, et al. Economic evaluation of a surveillance program of hepatocellular carcinoma (HCC) in India. *Hepatol Int.* 2008;2(2):231–236.
- Bolondi L, Gaiani S, Casali A, Serra C, Piscaglia F. Screening for the early diagnosis of hepatocellular carcinoma: cost-effectiveness analysis. [Italian]. *Radiol Med.* 1997;94(1-2):4–7.
- Bolondi L, Sofia S, Siringo S, et al. Surveillance programme of cirrhotic patients for early diagnosis and treatment of hepatocellular carcinoma: a cost effectiveness analysis. *Cut.* 2001;48(2):251–259.
   Pocha C, Dieperink E, McMaken KA, et al. Surveillance for hepatocellular
- Pocha C, Dieperink E, McMaken KA, et al. Surveillance for hepatocellular cancer with ultrasonography vs. computed tomography: a randomised study. *Aliment Pharmacol Ther.* 2013;38(3):303–312.
- Violi NV, Lewis S, Liao J, et al. Gadoxetate-enhanced abbreviated MRI is highly accurate for hepatocellular carcinoma screening. *Eur Radiol*. 2020;1–11.
- Cameron D, Ubels J, Norström F. On what basis are medical cost-effectiveness thresholds set? Clashing opinions and an absence of data: a systematic review. Glob Health Action. 2018;11(1):1447828.
- Nimdet K, Chaiyakunapruk N, Vichansavakul K, Ngorsuraches S. A systematic review of studies eliciting willingness-to-pay per quality-adjusted life year: does it justify CE threshold? PLoS One. 2015;10(4):e0122760.
- Ben Chaabane N, Jeridi N, Loghmari H, et al. Surveillance programme of cirrhotic patients for early diagnosis and treatment of hepatocellular carcinoma in Tunisia: A cost-effectiveness analysis. [French]. Journal Africain d'Hepato-Gastroenterologie. 2011;5(3):159–162.
- Kchaou-Ouakaa A, Belhadjbrik N, Elloumi H, et al. Surveillance for hepatocellular carcinoma: Does it work? [French]. *Tunisie Medicale*. 2007;85(10):866–870.
- Bennett WG, Inoue Y, Beck JR, et al. Estimates of the cost-effectiveness of a single course of interferon-alpha 2b in patients with histologically mild chronic hepatitis C. Ann Intern Med. 1997;127(10):855–865.

11

...

- Kim WR, Poterucha JJ, Hermans JE, et al. Cost-effectiveness of 6 and 12 months of interferon-alpha therapy for chronic hepatitis C. Ann Intern Med. 1997;127(10):866–874.
- Younossi ZM, Boparai N, McCormick M, Price LL, Guyatt G. Assessment of utilities and health-related quality of life in patients with chronic liver disease. Am J Gastroenterol. 2001;96(2):579–583.
- Levy AR, Kowdley KV, Iloeje U, et al. The impact of chronic hepatitis B on quality of life: a multinational study of utilities from infected and uninfected persons. *Value Health*. 2008;11(3):527–538.
- Chong CA, Gulamhussein A, Heathcote EJ, et al. Health-state utilities and quality of life in hepatitis C patients. Am J Gastroenterol. 2003;98(3):630–638.
- Mathers C, Vos T, Stevenson C. The Burden of Disease and Injury in Australia. Canberra: Australian Institute of Health and Welfare; 1999.
- Singal AG, Yopp A, C SS, et al. Utilization of hepatocellular carcinoma surveillance among American patients: a systematic review. J Gen Intern Med. 2012;27:861–867.
- 100. Singal AG, Pillai A, Tiro J. Early detection, curative treatment, and survival rates for hepatocellular carcinoma surveillance in patients with cirrhosis: a meta-analysis. *PLoS Med.* 2014;11(4):e1001624.
- 101. Kansagara D, Papak J, Pasha AS, et al. Screening for hepatocellular carcinoma in chronic liver disease: a systematic review. Ann Intern Med. 2014;161(4):261–269.
- Pascual S, Miralles C, Bernabe JM, Irurzun J, Planells M. Surveillance and diagnosis of hepatocellular carcinoma: a systematic review. World J Clin Cases. 2019;7(16):2269–2286.
- 103. Kudo M. Management of hepatocellular carcinoma in Japan as a worldleading model. *Liver Cancer*. 2018;7(2):134–147.
- Kudo M. Japan's Successful model of nationwide hepatocellular carcinoma surveillance highlighting the urgent need for global surveillance. *Liver Cancer*. 2012;1(3-4):141–143.

- Australian Institute of Health and Welfare. Cancer data in Australia. https:// www.aihw.gov.au/reports/cancer/cancer-data-in-australia/contents/survival. Accessed November 15, 2019.
- Yoon SK, Chun HG. Status of hepatocellular carcinoma in South Korea. Chin Clin Oncol. 2013;2(4):39.
- 107. Tzartzeva K, Obi J, Rich NE, et al. Surveillance imaging and alpha fetoprotein for early detection of hepatocellular carcinoma in patients with cirrhosis: a meta-analysis. *Gastroenterology*. 2018;154(6):1706–1718.
- 108. Yu NC, Chaudhari V, Raman SS, et al. CT and MRI improve detection of hepatocellular carcinoma, compared with ultrasound alone, in patients with cirrhosis. *Clin Gastroenterol Hepatol.* 2011;9(2):161–167.
- 109. Salvatore M, Jeon J, Meza R. Changing trends in liver cancer incidence by race/ethnicity and sex in the US: 1992-2016. Cancer Causes Control. 2019;30(12):1377–1388.
- Cancer Research UK. Liver cancer incidence statistics. https://www. cancerresearchuk.org/health-professional/cancer-statistics/statistics-by-canc er-type/liver-cancer/incidence. Accessed July 26, 2019.
- Wong LL, Reyes RJ, Kwee SA, et al. Pitfalls in surveillance for hepatocellular carcinoma: how successful is it in the real world? *Clin Mol Hepatol.* 2017;23(3):239–248.
- 112. Simmons O, Fetzer DT, Yokoo T, et al. Predictors of adequate ultrasound quality for hepatocellular carcinoma surveillance in patients with cirrhosis. *Aliment Pharmacol Ther*. 2017;45(1):169–177.
- 113. Gopal P, Yopp AC, Waljee AK, et al. Factors that affect accuracy of alphafetoprotein test in detection of hepatocellular carcinoma in patients with cirrhosis. *Clin Gastroenterol Hepatol.* 2014;12(5):870–877.
- 114. Zhang J, Yu Y, Li Y, Wei L. Diagnostic value of contrast-enhanced ultrasound in hepatocellular carcinoma: a meta-analysis with evidence from 1998 to 2016. *Oncotarget*. 2017;8(43):75418–75426.